SOCIETY OF INDUSTRIAL AND OFFICE REALTORS® (SIOR) ETHICS, MEDIATION & ARBITRATION PROCEDURES MANUAL

TABLE OF CONTENTS

PART I: OVERVIEW OF SIOR'S PROFESSIONAL STANDARDS PROCESS	3
PART II: ROCEDURES FOR THE ENFORCEMENT OF ETHICS OBLIGATIONS	10
PART III: MEDIATION AND ARBITRATION OF FINANCIAL DISPUTES	33
PART IV: FORMS USED IN PROFESSIONAL STANDARDS PROCEEDINGS	54
APPENDIX A: CODE OF ETHICAL PRINCIPLES AND STANDARDS OF PROFESSIONAL PRACTICE OF THE SOCIETY OF INDUSTRIAL AND OFFICE REALTORS®	
APPENDIX B: SIOR CODE OF CONDUCT	. 112
APPENDIX C: APPLICATION ETHICS COMPLAINT PROCEDURES	. 113

SOCIETY OF INDUSTRIAL AND OFFICE REALTORS® (SIOR) Ethics, Mediation & Arbitration Procedures Manual

PART I: OVERVIEW OF SIOR'S PROFESSIONAL STANDARDS PROCESS

A.	Governing Documents	. 3
В.	Adherents and Duties of Adherents	. 4
C.	Waiver of Rights of Action	. 4
D.	SIOR Authority to Discipline Adherents	. 4
E.	Nature of Discipline that the Board May Impose	. 5
F.	Professional Standards Committee	. 6
G	Professional Standards Staff	8

PART I: OVERVIEW OF SIOR'S PROFESSIONAL STANDARDS PROCESS

A. Governing Documents

SIOR's professional standards process is designed to provide for the fair and efficient evaluation of claimed violations of SIOR's *Code of Ethical Principles and Standards of Professional Practice* and the mediation and arbitration of certain financial disputes arising from the real estate-related professional or business activities of Active members or Member Associates. The process is grounded in three documents:

- 1. SIOR's Bylaws and Policy Manual establish the basic obligations of Active members and Member Associates, authorize sanctions for non-compliance, and establish organizational elements, including the Professional Standards Committee, that have roles in the process.
- 2. SIOR's Code of Ethical Principles and Standards of Professional Practice specifies the substantive rules governing professional conduct and specifies duties to resolve certain financial disputes through mediation or arbitration.
- 3. This *Ethics, Mediation & Arbitration Procedures Manual* defines specific procedures, policies, and forms used in ethics enforcement and financial dispute resolution.

This manual is organized as follows:

- Part I discusses in general terms, SIOR organizational elements that participate in ethics and arbitration proceedings.
- Part II describes the procedures and related policies governing proceedings related to claimed violations of ethical obligations.
- Part III describes the procedures and related policies governing the mediation or arbitration of financial disputes that are subject to resolution by SIOR.
- Part IV provides a set of the forms to be used as a guide in ethics enforcement and financial dispute resolution proceedings.
- Finally, for reference, the Code of Ethical Principles and Standards of Professional Practice is included as Appendix A.

Questions concerning this manual or related matters may be addressed to:

Professional Standards Staff Officer SIOR Headquarters 1307 New York Ave., NW, Ste 703, Washington, DC, 20005 Phone 202.449.8200 Fax 202.216.9325

SIOR Ethics, Mediation & Arbitration Procedures Manual – *revised 2021* Part I: Overview – page 3

B. Adherents and Duties of Adherents

As used in this manual, the term "Adherent" means any person who is an Active SIOR Designee (i.e., an Active member) or holds Member Associate status in SIOR. They are the persons who are subject to SIOR's Code of Ethical Principles and Standards of Professional Practice.

The duties of Adherents include, but are not limited to, the following:

- 1. To abide by the Code of Ethical Principles and Standards of Professional Practice;
- 2. To submit to mandatory mediation and arbitration all financial disputes specified in *the Code of Ethical Principles and Standards of Professional Practice*, and to abide by arbitration rulings; and
- 3. To abide by the Bylaws of SIOR and the rules and regulations adopted by SIOR's Board of Directors, including SIOR's Code of Conduct (See Appendix B).

The Code of Ethical Principles and Standards of Professional Practice applies to the professional practice of Adherents in commercial real estate transactions, and is enforced through the professional standards grievance review and hearing process, while the Code of Conduct relates to personal conduct, and applies to all participants in SIOR events as well as member communications related to SIOR business. The Code of Conduct is enforced through disciplinary action by SIOR executive leadership.

C. Waiver of Rights of Action

All SIOR Adherents, for and in consideration of their right and responsibility to invoke mediation and arbitration proceedings and to report potential violations or initiate complaints under the *Code of Ethical Principles and Standards of Professional Practice*, waive any rights of action against SIOR, any SIOR member, Member Associate, agent and their counsel, or any other participant arising out of any decisions, determinations, or other actions taken or rendered under these procedures in the absence of willful or wanton misconduct. Further, as a condition of continued SIOR affiliation, every Adherent expressly waives any cause of action for libel, slander, or defamation that might arise from the filing or consideration of any ethics complaint or mediation/arbitration request.

D. SIOR Authority to Discipline Adherents

Pursuant to SIOR's Articles of Incorporation and Bylaws, the power to impose discipline resides in SIOR's Board of Directors, whose exercise of that authority has been delegated in several ways, including to the committee and staff functions described in this manual.

Discipline may be imposed on an Adherent:

- 1. For violation by the Adherent of any duty described above.
- 2. For unethical and/or illegal acts of other Adherents or persons who are not themselves Adherents, in circumstances where the Adherent is engaged in a real estate-related professional activity that is conducted by, through, or in conjunction with one or more such persons as employees, agents, subordinates, colleagues, or associates.
- 3. On the Adherent's conviction, judgment plea, or other determination of guilt by a final judgment of any court of competent jurisdiction of a felony or a crime involving moral turpitude.

E. Nature of Discipline that the Board May Impose

Disciplinary action that SIOR's Board of Directors may impose on Adherents or Applicants for membership may consist of one or more of the following:

- 1. A letter of warning with a copy placed in the individual's file;
- 2. A letter of reprimand with a copy placed in the individual's file;
- 3. A requirement that the individual attend a SIOR course or seminar or a program sponsored by another REALTOR group that addresses ethics and can be reasonably attended taking into consideration cost, location, and duration;
- 4. An appropriate and reasonable fine not to exceed a maximum amount equal to ten times SIOR's annual dues in effect at the time the fine is imposed;
- 5. Placing the individual on probation for a stated period of time, not less than thirty (30) days nor more than one (1) year in duration;
 - PROBATION means the discipline recommended will be held in abeyance for a stipulated period of time, which may not exceed one (1) year. Any subsequent finding of a violation of SIOR's Bylaws or Code of Ethical Principles during the probationary period may, at the discretion of SIOR's Board of Directors, result in the imposition of the suspended discipline. Absent any subsequent findings of a violation during the probationary period, both the probationary status and the suspended discipline shall be considered fulfilled, and the record shall reflect the fulfillment:
- 6. Suspension of membership or Member Associate status for a period not less than thirty (30) days nor more than one (1) year, with automatic reinstatement to good standing at the end of the suspension period;
 - SUSPENSION of membership means temporary termination of membership rights and privileges and denial of SIOR services and benefits for a period not less than thirty (30) days nor more than one (1) year, on terms and conditions expressly stated for an established period of time, including, but not limited to, use of the SIOR designation. At the expiration of the suspension period, all withdrawn membership rights and privileges will be automatically restored. During the suspension period, membership per se continues and the suspended members remain obligated for payment of membership dues and to abide by SIOR's Bylaws and Code of Ethical Principles. Any failure to abide by the terms and conditions of the suspension, or any subsequent finding of a violation of SIOR's Bylaws or Code of Ethical Principles, shall be grounds for consideration as to possible extension of the suspension or expulsion from membership in SIOR by SIOR's Board of Directors. In lieu of suspension, at the option of the Board, an assessment not to exceed an amount equal to ten times the SIOR's annual dues in effect at the time the assessment is imposed, which assessment may be imposed on a given Adherent only once in any three (3) year period;
- 7. Expulsion from membership or Member Associate status with no reinstatement opportunity for a specified period of one (1) to three (3) years, with any subsequent readmission to be by application only after the specified period of expulsion, the decision on readmission to be made based on the merits of the application, and to be considered under the admissions standards in effect at the time.
- 8. For Applicants to Membership: Denial of membership, or prohibition of application for Membership for up to five (5) years from the date of findings of an SIOR Hearing Panel.

SIOR has adopted criteria limiting imposition of the more serious sanctions, suspension and expulsion. An Adherent may be suspended or expelled where the Hearing Panel finds:

- a. A violation that caused death or significant personal injury; or
- b. A violation that has caused substantial economic injury to the party represented or to any other party to whom a duty of honesty, faith, trust, or fiduciary is owed; or
- c. A violation that has produced governmental criticism or substantial media criticism of the adherent and/or has cast aspersions on the real estate profession or SIOR; or
- d. That the respondent's misconduct involved a violation as equally serious as those addressed by the above three criteria but not literally covered by any one of them; or
- e. That within the 10 years preceding the complaint, the respondent has been found to have violated the Code on two (2) or more separate occasions.

Criteria a. and b. are intended to identify violations that have caused serious harm to individuals or businesses. Criterion c. encompasses certain violations that have adversely affected the reputation of the profession or SIOR. Criterion d. covers violations of equal or greater severity that do not fall literally within the language of any of the first three criteria.

Criterion e. covers the situation where an adherent has been found to have engaged in repeated violations over a 10-year period, indicating a pattern of non-compliance. It is not required that each violation is serious; the emphasis is on repetition. For purposes of this guideline, the violation found by the panel counts as one of the two. Many violative-conduct scenarios involve violations of more than one *Code* Principle or Standard of Practice. This guideline is not triggered by the violation of two Principles or Standards in a single incident; rather, it applies where there are found (partly by earlier Hearing Panels in most instances) to have been two or more separate incidents involving violations by the respondent.

The Board of Directors may, at its discretion, take all actions at law and equity to enforce discipline imposed on SIOR Adherents.

F. Professional Standards Committee

SIOR's Bylaws and Policy Manual provide for the establishment of a Professional Standards Committee, which consists of a Chair, a Vice Chair, and not more than eighteen members representing geographically diverse areas (including a representative from each of SIOR's regions when possible). Recent experience on a local Grievance or Professional Standards Committee is preferred for appointment to this committee, and other dispute resolution experience (mediation, arbitration, ombudsman) is considered helpful. Familiarity with SIOR's Code of Ethical Principles and Standards of Professional Practice and its enforcement and administrative procedures is required of all members of this committee.

The purposes of the Professional Standards Committee are to investigate, review, and evaluate ethics complaints and arbitration requests, to determine their efficacy and the course required for the appropriate disposition of such matters through Grievance Review Panels which are formed as needed; to provide mediation and arbitration facilities for the resolution of financial and contractual disputes involving members (through mediators and Hearing Panels formed as needed); to conduct hearings for the adjudication of complaints involving allegations of professional misconduct by SIOR members, Member Associates, and Applicants for either Active Designation or Member Associate, through Hearing Panels formed as needed; to evaluate and recommend changes to the Code and its related policies and procedures, to develop case studies and other interpretational aids to illustrate the application of the Code in business and professional circumstances; to educate the membership about the practical applications of SIOR's Code of Ethical Principles and Standards of Professional Practice, and to provide

a pool of members who are experienced in professional standards to fulfill SIOR's Grievance Review and Hearing Panel needs.

A Grievance Review Panel appointed by the Professional Standards Committee Chair will be responsible for reviewing and evaluating written complaints and other reported potential violations of the Code of Ethical Principles and Standards of Professional Practice to determine whether they warrant a professional standards hearing. The Grievance Review Panel refers those cases which it determines warrant hearings to a Hearing Panel to be appointed by the Professional Standards Committee Chair. The Professional Standards Committee itself may draft ethics complaints and refer them for hearing. A Hearing Panel is responsible for adjudicating complaints related to ethics that are referred by a Grievance Review Panel. The Professional Standards Committee Chair may appoint Hearing Panels to hold hearings on claimed violations of under the Code of Ethical Principles and Standards of Professional Practice and render decisions on whether violations occurred. Where one or more violations are found, the Hearing Panel will select appropriate disciplinary action that will be forwarded as a recommendation to the Board of Directors.

The Professional Standards Committee also supervises the mediation of financial disputes referred by a Grievance Review Panel for mediation/arbitration review, and where mediation does not lead to an agreed resolution, provides arbitration Hearing Panels. The Professional Standards Committee Chair may designate qualified members of the Committee as mediation officers and may appoint Grievance Review Panels for review of a mediation/arbitration request and arbitration Hearing Panels. A Grievance Review Panel itself does not mediate or arbitrate business disputes. Arbitration Hearing Panels conduct hearings on financial disputes, and render decisions that may include arbitration awards, subject to procedural review by the Board of Directors.

Proceedings before a Grievance Review Panel or an ethics or arbitration Hearing Panel will be conducted in accordance with applicable provisions of SIOR's Ethics, Mediation and Arbitration Procedures Manual.

No Committee member who has participated in the consideration of a matter by a Grievance Review Panel will participate in the consideration of the same matter as a Hearing Panel member.

The Committee develops professional standards policies and procedures applicable to members, individuals holding Member Associate status in SIOR, or Applicants for either Active Designee or Member Associate status. This function includes developing interpretations and recommendations for changes to the Code of Ethical Principles and Standards of Professional Practice. It also includes developing interpretations and recommendations for changes to the *Ethics, Mediation and Arbitration Procedures Manual.*

The Committee will work with other SIOR relevant committees to develop and conduct educational activities related to ethics and arbitration obligations of members and Member Associates, as well as related procedures. This function includes development and implementation of educational courses, conference programs, and roundtables.

In accordance with procedures outlined in this manual, a Grievance Review Panel reviews ethics complaints to determine if they are in the proper form and whether, assuming the facts alleged are true, a hearing is warranted. When a complaint is received by the Committee, the Professional Standards Committee Chair will designate specific members of the Professional Standards Committee, totaling not less than three (3) in number, who shall function as the Grievance Review Panel for assessing that complaint. While the Grievance Review Panel has meetings, it does not hold hearings; and it does not decide whether Adherents have violated the Code of Ethics.

Standard of Practice 15.4 obligates all Adherents to report to SIOR potential ethics violations by other Adherents as to which the reporting Adherent has firsthand knowledge. When such a report is received, the Professional Standards Committee Chair will designate specific members of the Professional Standards Committee, totaling not less than three (3) in number, who shall function as the Grievance Review Panel for assessment of that report. The Grievance Review Panel reviews and investigates such

reports and, in appropriate circumstances, may initiate an ethics enforcement proceeding. Where an ethics enforcement proceeding is commenced based on one or more reports of potential violation, or in response to an investigation requested by the Board of Directors, the Grievance Review Panel will draft a complaint, and a Grievance Review Panel member will act as complainant in subsequent proceedings.

The Professional Standards Committee Chair forms Hearing Panels that hold hearings on claimed violations of the *Code of Ethical Principles and Standards of Professional Practice* and render decisions on whether violations occurred. Where one or more violations are found, the panel selects appropriate disciplinary action, which is then forwarded as a recommendation to the Board of Directors.

Similarly, the Professional Standards Chair supervises the mediation of financial disputes, and where mediation does not lead to an agreed resolution, forms arbitration Hearing Panels. An arbitration Hearing Panel will consist of not fewer than three (3) members of the Professional Standards Committee. Arbitration Hearing Panels conduct hearings on financial disputes and render recommended decisions that may include arbitration awards.

Subject to review by the Board of Directors, panel decisions represent the conclusion of an ethics violation hearing or an arbitration hearing.

G. Professional Standards Staff

SIOR maintains a professional staff that is based at SIOR's Washington Headquarters. SIOR's chief administrative officer is its Chief Executive Officer, whose responsibilities are defined in the SIOR Bylaws. The staff professionals under the supervision and control of the Chief Executive Officer include the Professional Standards Staff Officer who provides the principal day-to-day staff support for the Board of Directors and the Professional Standards Committee, with respect to ethics enforcement proceedings and financial dispute resolution proceedings. The Professional Standards Staff Officer also provides impartial assistance to Adherents and other persons involved in proceedings or otherwise have an interest in SIOR's professional standards program and processes.

SOCIETY OF INDUSTRIAL AND OFFICE REALTORS® (SIOR) Ethics, Mediation & Arbitration Procedures Manual

PART II: PROCEDURES FOR THE ENFORCEMENT OF ETHICS OBLIGATIONS

A.	Initiating an Ethics Proceeding by Complaint	10
B.	Reports of Potential Ethics Violations	10
C.	Timely Filing of Ethics Complaints	10
D.	Action By or Involving the Professional Standards Staff Officer	10
E.	Grievance Review Panel's Review of an Ethics Complaint	11
F.	Ethics Hearing Preparations	14
G.	Postponement of a Hearing and Failure to Appear	16
H.	Conduct of an Ethics Hearing	16
l.	Hearing Format	17
J.	Roles of Complainant, Respondent, and Counsel	17
K.	Decision of Hearing Panel	18
L.	Parties' Rehearing Rights	18
M.	Action by the Board of Directors If There Is No Appeal	19
N.	Appeals by Parties to the Board of Directors	19
Ο.	Confidentiality of Ethics Proceedings and Results	21
Ρ.	Withdrawal of an Ethics Complaint	21
Q.	Resignation of a Respondent before Final Disposition of an Ethics Complaint	22
R.	Initial Action by the Board of Directors for Failure to Arbitrate or Abide by the Award	22
S.	Ethics Case Statistics May Be Published	22
HEA	ARING PANEL CHAIR PROCEDURAL GUIDE FOR ETHICS HEARINGS	23
۸۵۲	DEAL DANIEL CHAID DECCEDIDAL CHIDE FOR ETHICS ADDEAL HEADINGS	20

PART II: ROCEDURES FOR THE ENFORCEMENT OF ETHICS OBLIGATIONS

A. Initiating an Ethics Proceeding by Complaint

Any person having reason to believe that any conduct may be subject to disciplinary action as a violation of SIOR's *Code of Ethical Principles and Standards of Professional Practice* may file a complaint in writing stating the facts on which it is based, using SIOR-PS Form 2 ("Ethics Complaint"). Copies of key documents relied upon should accompany the complaint. The complaint should be dated, signed, and verified by the complaining person. Complaints should be sent to SIOR's Washington Headquarters to the attention of the Professional Standards Staff Officer.

B. Reports of Potential Ethics Violations

Standard of Practice 15.4 obligates all Adherents to report to SIOR all potential violations of SIOR's *Code of Ethical Principles and Standards of Professional Practice* by other Adherents of which the reporting Adherent has first-hand knowledge. Such reports should be made using SIOR-PS Form 2A ("Report of Potential Ethics Violation"). They should be sent to SIOR's Washington Headquarters to the attention of the Professional Standards Staff Officer. SIOR-PS Form 2A should not be submitted where the same Adherent has filed a SIOR-PS Form 2 complaint as to the same conduct by the same Adherent. SIOR-PS Form 2A reports will be handled by the Professional Standards Staff Officer and assessed by a Grievance Review Panel in accordance with the procedures stated below for review of ethics complaints.

C. Timely Filing of Ethics Complaints

Complaints alleging ethical misconduct on the part of an Adherent must be mailed or otherwise transmitted to SIOR within one (1) year of the later of: (a) the date of the closing of a sale from which the matter arises or to which it is related; (b) the commencement date of a lease (or its renewal or extension) from which the matter arises or to which it is related; (c) the rental commencement date of a lease (or its renewal or extension) from which the matter arises or to which it is related; (d) the date on which any litigation arising from or related to the matter is concluded; (e) the date on which any proceeding (arising from or related to the matter) before a licensing or regulatory authority is concluded; or (f) the date on which the matter is known, or could be known, in the exercise of reasonable diligence.

D. Action By or Involving the Professional Standards Staff Officer

After the earlier to occur of receipt of a "Response to Ethics Complaint" (SIOR-PS Form 6) or the expiration of the thirty (30) day period following the staff officer's issuance of a "Request for Information" (SIOR-PS Form 4), the Professional Standards Staff Officer will promptly refer any complaint to the Professional Standards Committee Chair, who will supervise assembly of a file to facilitate a Grievance Review Panel determination as to whether to:

- 1. Dismiss the complaint as unworthy of further consideration:
- 2. Refer it back to the complainant as appropriate for mediation/arbitration rather than disciplinary action: or
- 3. Refer it back to the Professional Standards Staff Officer with instructions to initiate steps leading to a hearing by Committee members designated by the Professional Standards Committee Chair.

The Professional Standards Staff Officer shall, following receipt of an Ethics Complaint (see SIOR-PS Form 2), promptly request a written response from the party complained of (called the "respondent") by sending a copy of the complaint to the respondent and requesting the respondent to submit a response (see SIOR-PS Forms 4, "Request for Information," and 6, "Response to Ethics Complaint"). The respondent will be advised that failure to respond to the complaint within thirty (30) days may result in the complaint's being forwarded for a hearing and may subject the respondent to a charge of violating the

Code of Ethical Principles and Standards of Professional Practice for failing to submit pertinent facts to an appropriate body. Only one response to a complaint by a complainant may be submitted.

Within fifteen (15) days of receipt of a response from the respondent, the Professional Standards Staff Officer will provide a copy of the response to the complainant. The staff officer will also provide copies of the complaint and response to the Grievance Review Panel. The Panel may, at its discretion, consider the response in evaluating the ethics complaint.

All parties will be required to sign a Confidentiality Statement (see SIOR-PS Form 23).

E. Grievance Review Panel's Review of an Ethics Complaint

The function of a Grievance Review Panel is to make only such preliminary review and evaluation of a complaint as is required to determine whether the complaint warrants further consideration by a Hearing Panel appointed by the Professional Standards Committee Chair. A Grievance Review Panel does not conduct hearings and does not determine if a violation of the *Code of Ethical Principles and Standards of Professional Practice* has occurred.

1. Chair's Action upon Receipt of an Ethics Complaint

Upon receipt of an ethics complaint from the Professional Standards Staff Officer, the Professional Standards Committee Chair will review the complaint and any accompanying evidence and/or documentation. The Chair will appoint a Grievance Review Panel consisting of not fewer than three (3) and always an odd number of members of the Professional Standards Committee, and will assign a member of the Panel to review the complaint and to make specified inquiries to both parties. A copy of the complaint will be provided to the assigned member by the Professional Standards Staff Officer upon instruction from the Chair.

The reviewer will complete the assignment promptly and prepare a report, which may include a recommendation, for the Grievance Review Panel. After reviewing the report, the Chair will schedule a meeting of the Grievance Review Panel, which may meet by audio or video conference call. The staff officer will provide members of the Grievance Review Panel with copies of the case file, including the reviewer's report, if any, prior to such meeting.

If civil litigation, regulatory proceedings, or administrative proceedings including, without limitation, an ethics complaint filed in another forum related to the same transaction are pending, the Professional Standards Committee Chair may instruct the Professional Standards Staff Officer to have SIOR's legal counsel review the complaint and advise if any hearing should proceed (presuming the matter would otherwise warrant a hearing), with counsel to consider:

- i. the similarity of factors giving rise to the pending litigation or regulatory or administrative proceedings and the ethics complaint;
- ii. the degree to which resolution of the pending civil litigation or regulatory or administrative proceeding could make consideration of the ethics complaint unnecessary;
- iii. the degree to which the pending litigation or regulatory or administrative proceeding could delay prompt disposition of the ethics complaint;
- iv. any relationship between a pending litigation and an Adherent's duty under the Code of Ethical Principles and Standards of Professional Practice to arbitrate financial disputes;
- v. the nature of the alleged violation and the extent to which it could impact on cooperation with either SIOR members or Member Associates; and

vi. whether consideration of an ethics complaint during the pendency of other proceedings appears likely to be procedurally unfair to the respondent.

Counsel's report will be made part of the case file submitted to the Grievance Review Panel.

2. Consideration of an Ethics Complaint by the Grievance Review Panel

In reviewing an ethics complaint, the Grievance Review Panel will consider the following:

- a. Is the ethics complaint acceptable in form as received by the Panel? If not in proper form, the Panel may request that the staff officer contact the complainant to advise that the complaint must be re-submitted in proper form.
 - NOTE: If deemed appropriate by the Grievance Review Panel, the Professional Standards Staff Officer may be assigned to contact the complainant and provide procedural assistance to amend the complaint or resubmit a new complaint in proper form and with proper content. The person providing such assistance will ensure that only procedural assistance is provided to the complainant and that the complainant understands and agrees that the person is not representing the complainant.
- b. Was the complaint initially filed within one (1) year* of the later of: (i) the date of the closing of a sale from which the matter arises or to which it is related; (ii) the commencement date of a lease (or its renewal or extension) from which the matter arises or to which it is related; (iii) the rental commencement date of a lease (or its renewal or extension) from which the matter arises or to which it is related; (iv) the date on which any litigation arising from or related to the matter is concluded; (v) the date on which any proceeding (arising from or related to the matter) before a licensing or regulatory authority is concluded; or (vi) the date on which the matter is known, or could be known, in the exercise of reasonable diligence?

*Ten-year period for an Ethics Complaint against an Applicant for membership or Member Associate.

- c. Are all necessary parties named in the complaint?
- d. Is each respondent named in the complaint an Adherent, and was each respondent an Adherent at the time of the alleged offense? SIOR is not authorized to discipline persons who are neither members nor Member Associates or to require that non-Adherents participate in its proceedings.
- e. Is litigation or any government agency investigation or other governmental action pending related to the same transaction?
 - i. If criminal litigation is pending related to the same transaction, the Grievance Review Panel will suspend its consideration and instruct the staff officer to hold the file open until such time as the criminal litigation is concluded and all appellate remedies exhausted.
 - ii. If civil litigation or regulatory or administrative proceedings related to the same transaction are pending, the Grievance Review Panel will consider whether a grievance proceeding should go forward, taking into account the report submitted by SIOR's legal counsel.
- f. Is there any reason to conclude that SIOR would be unable to provide an impartial Hearing Panel?

- g. Are the specific provisions of the Code of Ethical Principles and Standards of Professional Practice cited in the complaint appropriate in light of the facts asserted? Should additional provisions be cited? Are any inappropriate provisions cited?
- h. If the facts alleged in the complaint were taken as true on their face, is it reasonably possible that a violation of the Code of Ethical Principles and Standards of Professional Practice occurred?

The Panel may, at its discretion, instruct the Professional Standards Staff Officer to obtain additional information from either the Complainant(s) or the Respondent(s), or to obtain clarification about information already provided.

If all relevant questions have been answered to the satisfaction of a majority of the Grievance Review Panel, and if the facts alleged appear to a majority to indicate the reasonable possibility of violation, the Panel will direct the Professional Standards Staff Officer to initiate steps leading to a hearing by Professional Standards Committee members designated by the Professional Standards Committee Chair; Otherwise, the Grievance Review Panel will order dismissal of the complaint and state the basis of that decision. In event of a tie among Grievance Review Panel members, the Panel Chair will have the deciding vote.

If a complaint states multiple instances of alleged unethical conduct and the Grievance Review Panel determines that one or more of those instances would not, under any reasonably foreseeable circumstances, constitute a violation, such portion of the complaint may be dismissed while the balance of the complaint is forwarded for a hearing before a Hearing Panel appointed by the Professional Standards Committee Chair. However, the complainant has the right to appeal the partial dismissal to the Board of Directors.

If the Grievance Review Panel concludes that a respondent's alleged conduct may be the basis for a violation of the *Code of Ethical Principles and Standards of Professional Practice* but that an inappropriate provision has been cited, the Panel may amend the complaint by deleting any inappropriate provision(s) and by adding any appropriate provision(s). It may also add or delete individuals as respondents to the complaint. If the complainant disagrees with any amendment to the *Code of Ethical Principles and Standards of Professional Practice* provisions alleged to have been violated or to the deletion of any respondent, the complainant may appeal to the Board of Directors. If the Grievance Review Panel determines that additional provision(s) or additional respondent(s) should be added to the complaint, but the complainant will not agree to pursue the addition, the Grievance Review Panel may file and prosecute its own complaint covering the addition, and both complaints will be heard simultaneously by the same Hearing Panel.

3. Any action by the Grievance Review Panel dismissing all or part of a complaint as unworthy of further consideration may be appealed by Complainant to the Board of Directors via the Professional Standards Staff Officer within twenty (20) days from receipt of the dismissal notice. The SIOR President and Chairman of the Board may appoint a panel of directors to act on behalf of the Board of Directors to consider and decide such an appeal. Any appeal panel so appointed must be composed of at least five (5) directors. Alternatively, the president may direct that the appeal may be heard by the Board's Governance Committee.

Only those materials and information that were available to the Grievance Review Panel when that Panel made its decision will be presented to the Board of Directors and considered with the appeal; and neither the complainant nor the respondent shall have the right to appear before the Board of Directors. If the Board determines that all or part of the complaint was improperly dismissed by the Grievance Review Panel Chair, it will refer that portion of the complaint to the Professional Standards Committee for a hearing. The decision of a Board appeal panel or the Governance Committee is final and binding, and is not subject to appeal or further review by the Board of Directors.

F. Ethics Hearing Preparations

- If not timely received, a Hearing Panel may accept late filing of a response to an ethics complaint at its discretion. Within fifteen (15) days of receipt of a response to the complaint from the respondent, the Professional Standards Staff Officer will provide a copy to the complainant. The Professional Standards Staff Officer will also provide copies of the complaint and response to the Professional Standards Committee Chair.
- 2. The Professional Standards Staff Officer will promptly transmit to each of the parties a list of the names of those members of the Professional Standards Committee selected by the Professional Standards Committee Chair as potential Hearing Panel members. No person who has served on the Grievance Review Panel for that complaint may be included. The parties have ten (10) days to challenge any names on the list (using SIOR-PS Form 1A, "Notice of Right to Challenge Hearing Panel Members"). A potential member of a Hearing Panel will be disgualified if she/he is:
 - a. related by blood or other familial relationship to either the complainant or the respondent;
 - b. an employer, partner, or employee directly associated in business with, or affiliated with any person or entity directly associated in business with, either the complainant or the respondent;
 - c. a party to the hearing or a party or a witness in any other pending case involving the complainant or the respondent; or
 - d. in any other relationship or has knowledge which may prevent him/her from rendering an impartial decision.

The Professional Standards Committee Chair will have the final authority to determine the validity and legitimacy of all challenges which are timely raised. The Professional Standards Committee Chair will appoint, from among those names not successfully challenged by either party, three (3) or more members to serve on a panel to hear the complaint. If challenges to Committee members result in there being an insufficient number of Committee members to constitute a panel, the SIOR President may appoint qualified Board of Directors members to serve as panel members. The same challenge procedures and disqualification grounds apply to such directors. All Hearing Panels must be composed of an odd number of members. Each person chosen as a Hearing Panel member will promptly complete and sign a Certificate of Hearing Panel Qualification (SIOR-PS Form 1).

as panel Chair. The Chair of the Hearing Panel will designate the time and place of the hearing and the Professional Standards Staff Officer will notify the complainant and respondent (on SIOR-PS Form 8, "Official Notice of Ethics Hearing") a minimum of thirty (30) days prior to the scheduled hearing. Because SIOR's membership is spread over wide geographical areas, whenever practical, hearings will be scheduled during or adjacent to regularly scheduled semi-annual SIOR World Conferences to minimize travel and time inconvenience for all parties.

Upon recommendation of the SIOR Grievance Review Panel in a case, and written agreement of both Complainant(s) and Respondent(s), an ethics hearing involving a dispute between SIOR Members may be conducted by teleconference or videoconference. If both of these conditions are not met, an in-person hearing will be held during or adjacent to an SIOR World Conference as soon as feasible. The hearing format and all other ethics procedures outlined in this Manual, including the right of appeal, shall be the same for a hearing by teleconference or videoconference as for in in-person hearing.

3. The Professional Standards Committee Chair will select one of the Hearing Panel members to serve

In-person hearings are required for all Admissions cases.

Participation in a hearing without objection by a party will constitute a waiver of any defect in a notice of hearing.

The notice of hearing will contain names of members of the Hearing Panel and should be accompanied by an "Outline of Procedure for Ethics or Arbitration Hearings" (SIOR-PS Form 10).

The parties will not discuss the case with any member of the Hearing Panel or the Board of Directors at any time prior to announcement of a decision in the case.

Each party may be accompanied at an ethics hearing by a single non-legal counsel who is an Adherent to the Code or by legal counsel, provided that notice of intention to do so (which notice includes the name of the non-legal or legal counsel) is transmitted to all other parties and to the Hearing Panel at least fifteen (15) days prior to the hearing. Failure to provide timely notice will not invalidate a party's right to counsel but may result in a continuance of the hearing if the Hearing Panel determines that the other party has been prejudiced by late or lack of notice.

4. The Hearing Panel Chair will prescribe any specific procedures for the hearing required by the circumstances of a particular case, in a manner not inconsistent with these provisions.

Absent specific prior arrangements, in-person hearings typically are held in rooms such as hotel meeting rooms and are scheduled for less than half of a day. If any party requires unusual physical arrangements or period of time in excess of three hours in order that the party's case may be fully and fairly presented, the party should report such need or request to the Professional Standards Staff Officer in writing as soon as possible, so that there will be a reasonable opportunity to assess and, if appropriate, accommodate the request. Decisions on requests for unusual hearing physical arrangements or times will be ruled upon by the panel Chair, in consultation with the Professional Standards Staff Officer.

If any party, witness, or counsel involved in the proceeding requires accommodation because of a disability, SIOR will make commercially reasonable efforts to provide such accommodation. Individuals requiring such accommodation must contact the Professional Standards Staff Officer at least fifteen (15) days before the hearing.

- 5. Once the Grievance Review Panel concludes that a complaint requires a hearing before a Hearing Panel, the complaint may be withdrawn by the complainant only with the Hearing Panel's approval. The approved withdrawal of a complaint is not a determination on its merits.
- 6. Amendment of complaints:
 - (i) At any time prior to the hearing of a complaint, the complainant may file an amended complaint with the Professional Standards Staff Officer to clarify or supplement the allegations that are the subject of the hearing (excluding: (a) amendments pertaining to provisions or allegations previously dismissed by the Grievance Review Panel; and (b) any information regarding any circumstance which occurred, as opposed to which was first discovered, subsequent to the circumstance which is the subject of the complaint). If an amended complaint is filed at least thirty (30) days prior to the scheduled hearing, the respondent will be notified, provided a copy, and afforded an opportunity to file an amended response. An amendment filed less than 30 days prior to a hearing may cause the hearing to be postponed so the other party can respond.
 - (ii) Less than thirty (30) days prior to a hearing, or during a hearing, a complaint may be amended either by the complainant or upon motion of the Hearing Panel. Amendments to include Code of Ethical Principles and Standards of Professional Practice provisions or allegations previously dismissed by the Grievance Review Panel may be made only on the motion of the Hearing Panel.

In such event, the hearing, with the concurrence of the respondent, may proceed uninterrupted or may be reconvened on a date within two scheduled semi-annual meetings of

SIOR from the date of adjournment. If the respondent knowingly waives the right to an adjournment, the record should reflect the fact that the respondent was aware of the right to an adjournment but chose to proceed with the hearing without interruption on the basis of the amended complaint. If the hearing is adjourned to be reconvened at a later date, the amended complaint will be filed in writing, signed by the complainant and by the Hearing Panel Chair, and will be promptly served on the respondent.

7. After the Official Notice of Ethics Hearing (SIOR-PS Form 8) has been issued, the parties may provide the Professional Standards Staff Officer with additional written materials and/or statements to support their positions. Written affidavits should include notarized proof of their authorship and execution. At the hearing, panel members may also receive other written statements, documents, or other papers and will hear testimony from the parties and their witnesses. It is the responsibility of all parties to arrange for their witnesses to be present at hearing. In appropriate circumstances and subject to prior notice to the other parties and the Professional Standards Staff Officer, witnesses may participate by telephone or other electronic means. The Hearing Panel may receive and consider any evidence it deems material and proper, including evidence of experts. Non-legal counsel may communicate only with his/her client(s) during a hearing. Each party is responsible for the expenses of their witnesses and counsel.

G. Postponement of a Hearing and Failure to Appear

- 1. Postponement may be granted upon request if justified by circumstances. Requests for postponement must be made in writing. Permission can be granted by the Hearing Panel Chair. All parties will be advised of the date of the rescheduled hearing at least sixty (60) days in advance of the rescheduled hearing.
- 2. No hearing will be held in the absence of a complainant. In the event that a complainant refuses or is unable to attend a hearing, the complainant will be advised that refusal to participate in a hearing, absent a satisfactory reason, may result in a charge that the complainant has violated an Adherent's obligation to place all pertinent facts before an appropriate Hearing Panel. If a complainant continues to refuse the Hearing Panel's written request to appear, or if the complainant is excused from appearing for reasons deemed valid by the Hearing Panel Chair, the hearing will not take place but rather the complaint will be referred back to the Grievance Review Panel. If the Grievance Review Panel determines that there is sufficient information for a Hearing Panel to consider, the complaint may be amended to name the Grievance Review Panel as complainant, and the hearing shall be continued to a new date. In such cases, the respondent will be provided with a copy of the amended complaint at least thirty (30) days prior to the date of a rescheduled hearing.
- 3. If multiple complainants are involved in an ethics hearing, a complainant may testify remotely by telephone or video call for a compelling reason and at the discretion of the Professional Standards Committee Chair as long as at least one complainant is physically present at the hearing.

In the event that a respondent fails to appear at a duly noticed hearing without first obtaining a continuance or adjournment thereof, the Hearing Panel may proceed with the hearing in the respondent's absence and will reach its decision based on the evidence made available at the hearing.

H. Conduct of an Ethics Hearing

An ethics hearing involves a series of procedural steps or events that are coordinated by the Hearing Panel Chair. While each proceeding will present somewhat different circumstances, and the panel Chair will apply the standard procedures so as to accommodate the fair and efficient presentation of the parties' respective cases, the hearing will generally follow the procedural steps outlined in the "Hearing Panel Chair Procedural Guide for Ethics Hearings" which appears as Annex A at the end of this part of the *Manual*. It is essentially a form that the panel Chair uses as a starting point in preparing to direct the hearing. The parties are urged to consider these procedures and present any resulting issues or requests

in writing to deviate from the Procedural Guide to the Professional Standards Staff Officer well prior to the start of the hearing, so that their resolution will not cause unnecessary delay or inconvenience to other participants.

SIOR may, and any party may at its own expense, have a court reporter or mechanical audio recorder present at a hearing to record the proceedings. If a party records the proceedings and prepares a transcript, that party shall, at its sole expense, provide a copy to the staff officer, who may provide copies thereof to the Hearing Panel and to the other party. A hearing may not be videotaped without the prior written consent of all parties and SIOR.

I. Hearing Format

A Hearing Panel will not be bound by the rules of evidence applicable in courts of law, but will afford all parties a full opportunity to be heard, present witnesses, and offer evidence, subject to the Hearing Panel Chair's judgment as to relevance of the evidence offered.

The general hearing format will be as follows:

- 1. Opening statement by the Hearing Panel Chair, citing the panel's authority to hear the case and explaining the reasons for the hearing.
- 2. The ethics complaint will be read into the record.
- 3. All parties and witnesses will affirm that they will give truthful testimony.
- 4. The parties will be given an opportunity to present documentary evidence and testimony on their behalf, and they may call witnesses. All witnesses will be excused from the hearing except while testifying.
- 5. The parties will be afforded an opportunity to examine and cross-examine all witnesses and *parties*.
- 6. Panel members may ask questions at any time during the hearing.
- 7. The panel Chair may exclude any questions which the Chair deems irrelevant or argumentative.
- 8. Each party may make a closing statement. The complainant will make the first closing statement and the respondent will make the final closing statement.
- 9. The hearing will be adjourned.
- 10. The Hearing Panel will go into executive session to decide the case.

J. Roles of Complainant, Respondent, and Counsel

A party may be represented in an ethics hearing by a non-legal counsel (an "Advisor"), who must be an Adherent to the Code, or by legal counsel (both are referred to as "counsel"). Only one Advisor per complainant or respondent will be permitted. Parties will be required to respond directly to requests for information or questions addressed to them by members of the Hearing Panel except on the grounds of self-incrimination, or on other grounds deemed by the panel Chair to be appropriate. Parties will be held responsible for the conduct of their counsel. Counsel will be permitted to communicate only through their client during the hearing. Any effort by counsel to harass, intimidate, coerce, or confuse the panel members or any party to the proceedings, or any action by counsel which is viewed by the Hearing Panel as disruptive of the proceedings, will be grounds for exclusion of counsel from the hearing. The decision to exclude counsel for any of the foregoing reasons will be the result of a majority vote of the members of the Hearing Panel and will be non-appealable. In the event counsel is excluded, the hearing will be

postponed to a future date to enable the party to obtain alternate counsel, provided, however, such postponement will not be authorized if it appears to members of the panel that the action of counsel has been undertaken to obtain a postponement or delay of the hearing. If a party agrees to proceed in the absence of counsel, the hearing will not be postponed.

K. Decision of Hearing Panel

The decision of a Hearing Panel will be determined by a simple majority vote. The panel decision will be stated in writing (on SIOR-PS Form 11, "Ethics Hearing Panel Decision") and will contain findings of fact, findings as to violation of the *Code of Ethical Principles and Standards of Professional Practice* provisions involved, and a description of the recommended disciplinary action(s), if any. If any provision of SIOR's Bylaws, rules, or regulations relative to the Hearing Panel's handling of a matter is involved, the interpretation of the Bylaws, rule, or regulation will be set forth as a separate finding, and the Board of Directors, in any appeal from a decision of a Hearing Panel, will not be bound by the panel's interpretation. Panel decisions will not be disclosed during any appeal, or rehearing, to any persons except the complainant and his/her counsel, if any, the respondent and his/her counsel, if any, SIOR legal counsel, and the Professional Standards Staff Officer. However, failure of confidentiality will not invalidate the decision.

The decision will be filed with the Professional Standards Staff Officer within ten (10) days after the Hearing Panel's decision is final. A final Hearing Panel decision results when it is in writing and signed by members of the Hearing Panel following their personal review and any review by SIOR's legal counsel which may be requested by the Panel. Any member of a Hearing Panel not voting with the majority may dissent from all or any portion of the findings or decision and may file a dissent in writing with the staff officer for consideration by the Board of Directors at the same time the decision is considered. In the event the respondent is found in violation, the Hearing Panel may, at its discretion, consider all records of previous violations and sanctions imposed, if any, in the Adherent's file in determining discipline recommendations.

The Professional Standards Staff Officer will transmit a copy of the decision to the complainant and respondent within five (5) business days after the staff officer receives the written Hearing Panel decision, except that reasonable additional delay will not invalidate the decision.

L. Parties' Rehearing Rights

Within twenty (20) days after the decision has been transmitted to the parties by the Professional Standards Staff Officer, the complainant or the respondent may petition the Hearing Panel for a rehearing as to an adverse decision, solely on the grounds of newly discovered material evidence which the petitioner could not, with reasonable diligence, have discovered and produced at the original hearing. The petition must be in writing and include:

- 1. A summary of the new evidence;
- 2. A statement of what the new evidence is intended to show and how it might affect the Hearing Panel's decision; and
- 3. An explanation of why the petitioner could not have discovered and/or produced the evidence at the time of the original hearing.

Whenever possible, rehearing petitions will be considered by the original Hearing Panel. Rehearing will be granted on majority vote of the panel. When such petition is granted or denied, the Professional Standards Staff Officer will immediately inform the respondent and the complainant. **No more than one** (1) petition for rehearing may be filed by each party. Notice of a rehearing will be given not less than sixty (60) days before date of the rehearing.

M. Action by the Board of Directors If There Is No Appeal

- 1. If no appeal is filed, the Board of Directors must, subject to the process described below, adopt the Hearing Panel's recommendation and issue its order accordingly, except that the Board, if concerned with a possible procedural deficiency, may on its own motion refer the decision back to the Professional Standards Committee Chair for a new hearing and recommendation by a different Hearing Panel. If, however, the Board does not agree with the recommendation of sanction, the Board may impose alternative discipline that does not exceed that recommended by the Hearing Panel, or may refer the decision back to the original Hearing Panel for further consideration and recommendation accompanied by a description of the Board's concerns regarding the panel's proposed discipline. In such matters, advice of SIOR's legal counsel will be requested and considered. The Governance Committee may act for the Board unless the sanction involves suspension or expulsion from SIOR, which requires full Board action. The action of the Board will be recorded on SIOR-PS Form 12 ("Ethics Hearing Action by the Board of Directors").
- 2. If the respondent is currently on probation as the result of an earlier proceeding, the Board, upon reviewing the findings and recommendation for discipline resulting from the subsequent hearing, will also determine whether to impose any discipline that was held in abeyance during the probationary period. The Board will determine whether the subsequent ethics violation occurring during the probationary period was minor, inadvertent, or unrelated to the original violation in making their determination.
- 3. Any discipline imposed that requires an action on the part of the disciplined Adherent should also indicate any additional disciplines or penalties that may be automatically invoked for failure to comply by a specified compliance date. In the absence of such a provision, failure to comply with the discipline imposed should not be considered grounds for an additional ethics hearing, but rather should constitute the basis for consideration by the Board of Directors with regard to any additional action required to ensure compliance with the original discipline imposed. In the event that additional penalties are contemplated, the party failing to comply with the discipline originally imposed will have an opportunity to appear before the Board of Directors to explain the failure to comply. Absent an explanation acceptable to the Board, additional discipline, including possible suspension or expulsion, may be imposed in a manner consistent with this manual.
- 4. Upon final action by the Board of Directors, the Professional Standards Staff Officer will disseminate to the complainant, the respondent, the Board of Directors, the Chair and members of the Hearing Panel, SIOR legal counsel, the president of any Realtor® board or association to which the respondent belongs, and to any governmental agency as directed by the Board of Directors, such notice of the action as the staff officer deems appropriate under the circumstances to effectuate the discipline prescribed, provided that the nature, form, content, and extent of the notice will be specifically approved by SIOR's legal counsel prior to dissemination. SIOR Active members and Member Associates, other than those specified in the preceding sentence, will be notified only in respect to a respondent's suspension or expulsion from membership or Member Associate status.

N. Appeals by Parties to the Board of Directors

- 1. If no petition for rehearing is filed, either party may appeal to the Board of Directors within thirty (30) days after a Hearing Panel's decision has been transmitted to the parties. If rehearing has been sought, appeals must be requested within ten (10) days after denial of a petition for rehearing. An appeal is requested by timely filing with the Professional Standards Staff Officer an SIOR-PS Form 18 ("Request for Appeal of an Ethics Hearing Decision"). An appeal request must be accompanied by a deposit in the amount of \$1,000 (payable to SOCIETY OF INDUSTRIAL AND OFFICE REALTORS®). The appeal request must clearly indicate the bases on which the Hearing Panel's decision and/or recommendation for discipline is being challenged. Appeals may be granted only on the basis of:
 - a. procedural deficiency or error resulting in the appellant being deprived of due process of law;

- b. misapplication or misinterpretation of a provision of Code of Ethical Principles and Standards of Professional Practice found to have been violated; or
- c. inappropriateness or excessiveness of the discipline recommended by the Hearing Panel.

Only the stated bases and issues raised by the appellant may be considered in any appeal hearing before the Board of Directors. The written request for appeal will be reviewed by the Professional Standards Staff Officer only for the purpose of determining whether the appeal states an appropriate basis for consideration by the Board of Directors. If facially insufficient, it will be returned to the appellant, accompanied by an explanation and a request for additional detail to be received within ten (10) days. This initial administrative review is not a decision on the merits of the appeal request but is only intended to ensure compliance with the requirement that an appeal clearly set forth all bases that will be presented to the Board of Directors for its consideration.

2. When a request for appeal in proper form is received, the Professional Standards Staff Officer will immediately send a copy to the other party, and shall notify all parties at least thirty (30) days in advance of the time and place of the appeal hearing. As directed by the SIOR president, the hearing may be before the Governance Committee or a panel composed of not less than five (5) members of the Board of Directors (an "Appeal Panel"). The time set for the Appeal Panel hearing will be a time adjacent to the Board's next regular meeting or at a special meeting called for that purpose. Notice of the appeal hearing will be given using SIOR-PS Form 19 ("Official Notice of Ethics Appeal Hearing"). That form will provide the names of the members of the appeal Hearing Panel and will state the grounds upon which an Appeal Panel member may be disqualified.

That notice will be accompanied by SIOR-PS Form 20 ("Notice of Right to Challenge Hearing Panel Members in an Ethics or Arbitration Appeal Hearing"). A challenge may be submitted, using that form, within ten (10) days after the notice is sent. Anyone who is a party to or testified during the original ethics hearing, was a member of the Grievance Review Panel present during the meeting when this complaint was reviewed, was a member of the original Hearing Panel, or who was otherwise directly involved in the matter prior to the appeal, is disqualified from serving on the Appeal Panel. Furthermore, anyone who is related by blood or other family relationship to any party to the hearing, or who is an employer, partner, employee, or directly associated in business with any of the parties, or affiliated with any person or entity who is directly associated in business with any of the parties, is also disqualified. A party to the appeal may also cite any other ground that may prevent an appeal member from rendering an impartial decision. Challenges will be ruled upon by the SIOR President.

- 3. Any party may be represented by legal counsel and each party is requested to provide written notice of an intention to have counsel present at least ten (10) days before the scheduled hearing date. Failure to provide this notice will not invalidate a party's right to legal representation but may result in a continuance of the hearing if the Appeal Panel determines that the rights of any other party require representation by legal counsel.
- 4. At the appeal hearing, the Chair of the original Hearing Panel will present a transcript of the case or, if there is no transcript, will summarize the complaint, response, evidence presented, findings made, and discipline recommended. Each party will be entitled to offer corrections to the summary. Each party may present to the Appeal Panel reasons why the Hearing Panel's recommendation should or should not be followed, but no new evidence will be received (except as such new evidence may bear upon a claim of deprivation of procedural fairness). The hearing will be generally organized by and will follow the procedural steps outlined in the "Appeal Panel Chair Procedural Guide for Ethics Appeal Hearings," which appears as Annex B at the end of this part of the manual. It is essentially a form that the panel Chair uses as a starting point in preparing to direct the hearing. The parties are urged to consider these procedures and present any resulting issues or requests in writing to deviate from the procedures to the Professional Standards Staff Officer well prior to the start of the hearing, so that their resolution will not cause unnecessary delay or inconvenience to other participants. Issues raised in this manner will be ruled upon by the Appeal Panel Chair.

5. The Appeal Panel will render its decision promptly, by majority vote. The decision will be documented using SIOR-PS Form 12, "Ethics Hearing Action by the Board of Directors." The Appeal Panel may adopt or modify the recommendation of the Hearing Panel, including the discipline proposed. The Appeal Panel may dismiss the complaint if it concludes the findings of fact do not support the Hearing Panel's conclusion as to unethical conduct. The Appeal Panel may refer the matter back to the Professional Standards Committee Chair for a new hearing and recommendation by a different Hearing Panel if it determines that a procedural deficiency or error did or was likely to have materially influenced the outcome of the Hearing.

If the Appeal Panel is concerned with the appropriateness of the recommendation of sanction, the Appeal Panel may impose alternative discipline that does not exceed that recommended by the Hearing Panel, or may refer the decision back to the original Hearing Panel for further consideration and recommendation accompanied by a description of the Appeal Panel's concerns regarding the proposed discipline. In such matters, the advice of SIOR legal counsel will be requested and considered.

- 6. If the recommendation of the Hearing Panel is confirmed, the money deposited by the appellant will pass into SIOR's treasury. If the Hearing Panel's decision or discipline recommendation is rejected, the deposit will be returned to the party who made the deposit. If the recommendation is modified, the appeal panel will determine the disposition of the deposit.
- 7. If the recommendation of the Hearing Panel is modified, or if the complaint is dismissed, the Appeal Panel may state its reasons in writing, but failure to do so will not invalidate the decision of the Appeal Panel. Any member of the Appeal Panel not concurring with the decision of the Appeal Panel will be entitled to file a dissenting statement. Under no circumstances may the discipline imposed exceed that recommended by the Hearing Panel.
- 8. The decision of the Appeal Panel is final and will not be subject to further review by or appeal to the Board of Directors. Each Active member, and Member Associate, by applying for or becoming and remaining affiliated with SIOR, agrees not to seek review in any court of law in the absence of willful or wanton misconduct.

O. Confidentiality of Ethics Proceedings and Results

All members of any Hearing Panel will have an obligation to maintain and protect the confidentiality of the proceedings and deliberations of the Hearing Panel before, during, and after its determinations and recommendations. Hearing Panel members will not discuss Hearing Panel proceedings and deliberations with any person(s) except as required by these procedures or by law.

All other participants in an ethics proceeding also have an obligation to maintain and protect the confidentiality of the proceedings and their results.

Unauthorized disclosure includes any report or publication under any circumstances not established in this manual. The following are circumstances where disclosure by a party to an ethics and/or mediation/arbitration proceeding is authorized:

- 1. Where the dissemination of a decision to individuals who have some knowledge of the proceeding might vindicate an Adherent's professional reputation; or
- 2. Where there is a civil, criminal or administrative proceeding involving the same facts and circumstances which gave rise to the proceedings.

P. Withdrawal of an Ethics Complaint

An ethics complaint may be withdrawn only with the concurrence of the original grievance review panel. Both parties need to submit a request to do so in writing to the Chair of the Professional Standards Committee, who will forward the request to the original grievance review panel in the case for their decision.

Q. Resignation of a Respondent before Final Disposition of an Ethics Complaint

If an ethics respondent resigns or otherwise causes membership or Member Associate status in SIOR to terminate after an ethics complaint is filed but before final action is taken by the Board of Directors, the hearing process will suspend and the Professional Standards Staff Officer will forward the complaint to any other appropriate Realtor® board or association in which the respondent continues to maintain membership. If the respondent does not hold membership in a Realtor® board or association, or if the Professional Standards Staff Officer is unable to determine if the respondent holds membership in a Realtor® board or association, the complaint will be held in abeyance until such time as the respondent rejoins SIOR or it can be determined that the respondent holds membership in another Realtor® board or association. In any case where a complaint is transferred to another Realtor® board or association, the complainant and respondent will be so advised in writing.

R. Initial Action by the Board of Directors for Failure to Arbitrate or Abide by the Award

1. If the complaint against an Adherent is that the Adherent has improperly refused to submit a controversy to mediation/arbitration, the complaint will not be referred to a Grievance Review Panel or a Hearing Panel, but will be brought before the Board of Directors at its next regular meeting or at a special meeting called by the president for that purpose. The procedures for notices, time of notice and hearing prescribed for matters before a Hearing Panel will apply. The sole question for the Board to decide will be whether the respondent has failed to submit an arbitrable matter to mediation/arbitration.

Upon determination that a member has refused to mediate/arbitrate a properly arbitrable matter, the Board of Directors may direct the implementation of an appropriate sanction and may, if it has reason to believe that the imposition of a sanction will become the basis of litigation and a claim for damages consequent to such sanction, delay the effective date of implementing the sanction until final action has been taken on a petition for declaratory relief filed by SIOR to confirm the propriety of its determination and sanction.

2. On the other hand, if the complaint against the member is that, having properly submitted a controversy to arbitration, the member has refused to abide by the award, such refusal should not be referred to a Grievance Review Panel as a violation of the Code of Ethical Principles and Standards of Professional Practice unless it reflects an established pattern or practice of noncompliance with the commitment to arbitrate. A refusal to abide by an award in arbitration should be enforced in the manner set forth in the section of this manual governing arbitration disputes.

S. Ethics Case Statistics May Be Published

SIOR may publish, from time to time, a statistical summary reporting the number of ethics complaints filed, the provisions of *Code of Ethical Principles and Standards of Professional Practice* cited, the number of complaints dismissed by Grievance Review Panels, the number of violations of particular articles, the number of cases resulting in violations, or conversely, no violation, the number of cases in which sanctions were imposed, the range of sanctions imposed, and other appropriate information.

HEARING PANEL CHAIR PROCEDURAL GUIDE FOR ETHICS HEARINGS

The hearing will start promptly at the appointed time. All parties to the hearing are asked to enter the hearing room.

CHAIR'S OPENING STATEMENT

Ladies and gentlemen, I now call this hearing to order. The Hearing Panel of the Professional Standards Committee is charged with holding hearings for SOCIETY OF INDUSTRIAL AND OFFICE REALTORS® in accordance with the established procedures in matters concerning alleged unethical conduct of SIOR Applicants, Active members, or Member Associates, or in the arbitration of financial disputes arising out of the real estate business. The body meeting here is an impartial panel of the Professional Standards Committee that has been selected and called here today to ascertain the truth in the particular matter at hand, which is an ethics proceeding, and to render a decision on the testimony and evidence presented. An ethics proceeding should be recognized as different from an arbitration proceeding. The particular matter to be considered by this panel at this time is an ethics proceeding.

The Professional Standards Committee is a body duly constituted under the authority of the Bylaws of the Society of Industrial and Office Realtors®, and its members have been duly appointed by the SIOR president and approved by the Board of Directors. At this time, I would like to introduce the members of this panel, all of whom are members of the Professional Standards Committee.

Panel.		, a	and I WIII Se	erve as Cha	air of this	Hearing
The other members of this panel	are:					,
					,	and
Present at this hearing isbusiness associate(s),			, tl	ne complair	nant, and	his/her
<u> </u>	and	wit	nesses	who		include
	nd					
(If the complainant is accompanied by counsel not, ask if there is an objection. If none, have the Also present at this hearing isbusiness associate(s),	ne respon	dent sigr	n a stateme	nt to that eff	ect.)	
The respondent is accompanied by a non-leg counsel,	and	ł .	witnesses	wh	10	include
	nd					,
(If the respondent is represented by counsel, coask if there is an objection. If none, have the co						e. If not,
SIOR's legal counsel,Officer,				rofessional	Standard	ls Staff

BASIS OF THE HEARING

This hearing has been established to consider the c	complaint of	_
against	, alleging violation of	of
SIOR's Code of Ethical Principles and Standards of complainant is as follows:	of Professional Practice. The complaint, as f	iled by the

(Read the Ethics Complaint, SIOR-PS Form 2, into the record.)

This panel is not governed by the technical rules of evidence which may apply in a court of law. This panel will seek to determine all ascertainable and relevant facts pertaining to the matter under consideration to arrive at a peer judgment and decision by the panel that is fair to all of the parties. The panel is governed by the Bylaws of the Society of Industrial AND Office Realtors® and the Code of Ethical Principles and Standards of Professional Practice. It functions in accordance with SIOR's Ethics, Mediation and Arbitration Procedures Manual. The panel determines its own rules of evidence and its own specific procedures with objectives of equity and due process.

The following has been generally accepted and ruled on by this panel as to procedures to be followed during this hearing:

- All parties and their representatives to these proceedings will be allowed full opportunities to be heard on matters relevant to the issue. The panel may rule at any time during the hearing on the relevance of testimony being given or on questions being directed to any party or his/her representative or to witnesses providing testimony. All parties and witnesses will be asked to affirm that testimony given is the truth to the best of their knowledge.
- 2. (Read only if one or both parties have counsel present; otherwise, proceed to 3.)

A party may be accompanied by a single non-legal counsel who is an Adherent to the Code or by legal counsel. However, no party may refuse to respond directly to requests for information or questions addressed to him/her by members of the panel except on grounds of self-incrimination or other grounds which the panel deems appropriate. Counsel is present to advise and consult with his/her client, and will be permitted to communicate only through his/her client during the hearing. This panel will countenance no effort by any party or by any counsel to any party to harass, intimidate, coerce, or confuse the panel members or any party to the proceedings.

- 3. The panel may rule at any time on the admissibility of evidence. As Chair, I will act as keeper of the evidence introduced at this hearing and mark each with an exhibit identification number and date.
- 4. The members of this panel are authorized, individually, to ask questions as they deem pertinent and significant at any time during this hearing. To preserve order, I will rule on questions or testimony by the parties or by witnesses in these proceedings. If deemed necessary, I will consult with the members of this panel and with SIOR legal counsel concerning such rulings.
- 5. At this time, I request that all persons present in the room who expect to testify at this hearing stand and make appropriate affirmation.

(Ask each party and witness to stand and affirm.)

Raise your right hand, and following the question I now pose, answer in the affirmative if you do so affirm:

Do you affirm that the testimony you are about to give in this proceeding will be the truth, the whole truth, and nothing but the truth?

(If all parties affirm) Let the record show that all parties have answered in the affirmative. Please be seated. Witnesses are excused from the hearing at this time until called to provide testimony, and shall be excused after completion of their testimony.

OUTLINE OF PROCEDURE FOR HEARING

Both the complainant and the respondent were sent a copy of the "Outline of Procedure for Ethics or Arbitration Hearings" (SIOR-PS Form 10). Did each of you receive the Outline? (If yes) Let the record show that both the complainant and respondent have stated that they did receive the Outline. (If no, the party should be given a copy of the Outline and the Chair should determine whether that party has any objections to proceeding.) Do you have any questions concerning the Outline of Procedure? (If none) Let the record show that neither the complainant nor the respondent has any questions concerning the "Outline of Procedure for Ethics or Arbitration Hearings."

We will now proceed with the hearing.

PRESENTATION BY THE COMPLAINANT

The complainant will now state his/her case and present any evidence or witnesses that he/she may desire.

(Before each witness is excused)

CROSS EXAMINATION BY RESPONDENT

QUESTIONS FROM PANEL MEMBERS

PRESENTATION BY THE RESPONDENT

The respondent will now state his/her case and present any evidence or witnesses that he/she may desire.

(Before each witness is excused)

CROSS EXAMINATION BY COMPLAINANT

QUESTIONS FROM PANEL MEMBERS

CLOSING STATEMENTS

At this time, both the complainant and the respondent will be given opportunities to make a summary or closing statement if they so desire. The complainant's closing statement will be heard first.

CLOSING STATEMENT BY THE COMPLAINANT

CLOSING STATEMENT BY THE RESPONDENT

Does each of you feel that this hearing has been conducted fairly? (If yes) Let the record show that both the complainant and the respondent have indicated that they feel this hearing has been conducted fairly. Have each of you had adequate opportunity to testify, present evidence and witnesses, and conduct cross-examination? (If yes) Let the record show that both the complainant and the respondent have indicated that they have had the opportunity to testify, present evidence and witnesses, and conduct cross-examination. (If any party answers "no," ask him/her to explain his/her concern for the record.)

Before we adjourn the hearing, all persons are advised that the report and findings of the panel are considered confidential. They will be available only to members of this panel, the parties, counsel, and

staff as required, and to SIOR's Board of Directors as directed by provisions of the *Ethics, Mediation and Arbitration Procedures Manual.* The parties will be notified of the decision within the required time after this hearing is adjourned.

There being no further business to be considered in this hearing, this portion of the hearing stands adjourned.

EXECUTIVE SESSION

After adjournment, the panel will remain in executive session to deliberate and frame a decision stating its findings of fact, findings as to violation, and recommendation as to discipline, if applicable. The panel will follow the procedure set forth in SIOR's Ethics, Mediation and Arbitration Procedures Manual. The Hearing Panel's decision will be reported using SIOR-PS Form 11, "Ethics Hearing Panel Decision."

Disciplinary action that SIOR's Board of Directors may impose on Adherents or Applicants for membership may consist of one or more of the following:

- 1. a letter of warning with a copy to be placed in the Adherent's file;
- 2. a letter of reprimand with a copy to be placed in the Adherent's file;
- 3. a requirement that the Adherent attend a SIOR course or seminar or a program sponsored by another REALTOR[®] group that addresses ethics and can be reasonably attended taking into consideration cost, location, and duration;
- 4. an appropriate and reasonable fine not to exceed a maximum amount equal to ten times SIOR's annual national dues in effect at the time the fine is imposed:
- 5. placing the Adherent on probation for a stated period of time, not less than thirty (30) days nor more than one (1) year in duration;
- 6. suspension of Active membership or Member Associate status for a period not less than thirty (30) days nor more than one (1) year, with automatic reinstatement to good standing at the end of the suspension period; or, in lieu of suspension, at the option of the Board, an assessment not to exceed an amount equal to ten times SIOR's annual dues in effect at the time the assessment is imposed, which assessment may be imposed on a given Adherent only once in any three (3) year period;
- 7. expulsion from Active membership or Member Associate status with no reinstatement opportunity for a specified period of one (1) to three (3) years, with any subsequent readmission to be by application only after the specified period of expulsion, the decision on readmission to be made based on the merits of the application, considered under the admissions standards in effect at the time.
- 8. For Applicants to Membership: Denial of membership, or prohibition of application for Membership for up to five (5) years from the date of findings of an SIOR Hearing Panel.

SIOR has adopted criteria limiting when the more serious sanctions, suspension and expulsion may be imposed. An Adherent may be suspended or expelled where the Hearing Panel finds:

- a. A violation that caused death or significant personal injury; or
- b. A violation that has caused substantial economic injury to the party represented or to any other party to whom a duty of honesty, faith, trust, or fiduciary is owed; or

- c. A violation that has produced governmental criticism or substantial media criticism of the adherent and/or has cast aspersions on the real estate profession or SIOR; or
- d. That the respondent's misconduct involved a violation as egregious as those addressed by the above three criteria but not specifically covered by any one of them; or
- e. That within the 10 years preceding the complaint, the respondent has been found to have violated the Code on two (2) or more separate occasions.

Criteria a. and b. are intended to identify violations that have caused serious harm to individuals or businesses. Criterion c. encompasses certain violations that have adversely affected the reputation of the profession or SIOR. Criterion d. covers violations of equal or greater severity that do not fall specifically within the language of any of the first three criteria.

Criterion e. covers the situation where an adherent has been found to have engaged in repeated violations over a 10-year period, indicating a pattern of non-compliance. It is not required that each violation be serious; the emphasis is on repetition. For purposes of this guideline, the violation found by the panel counts as one of the two. Many violative-conduct scenarios involve violations of more than one Code Principle or Standard of Practice. This guideline is not triggered by the violation of two Principles or Standards in a single incident; rather, it applies where there are found (partly by earlier Hearing Panels in most instances) to have been two or more separate incidents involving violations by the respondent.

APPEAL PANEL CHAIR PROCEDURAL GUIDE FOR ETHICS APPEAL HEARINGS

The hearing will start promptly at the appointed time. All parties to the hearing are asked to enter the hearing room.

CHAIR'S OPENING STATEMENT

Ladies and gentlemen, I now call this hearing to order. SIOR's *Ethics, Mediation and Arbitration Procedures Manual* provides for the right to appeal the decisions rendered by ethics Hearing Panels. An appeal may be granted, and a panel decision set aside, on the basis of (1) misapplication or misinterpretation of a provision of the *Code of Ethical Principles and Standards of Professional Practice*; (2) procedural deficiency or errors resulting in the Respondent being deprived of due process of law; or (3) inappropriateness or excessiveness of the discipline recommended by the Hearing Panel. This particular appeal will be heard by [the Board of Directors, a panel of directors appointed by the President, or SIOR's Governance Committee], hereinafter referred to as the Appeal Hearing Panel.

The particular matter to be considered by the Appeal Hearing Panel is an appeal of the decision of an

ethics	Hearing	Panel	composed		•		ppcai oi iii		. an
	•		•						and
			,	_, and chaire	d by			, at an et	hics
hearing	conducted or	າ			The ba	isis of the c	riginal ethic	cs hearing wa	as a
compla	int brought					, the	complain	ant(s), aga	iinst
								g violation	
			Code of Ethica	al Principles	and Sta	andards of	Profession	nal Practice.	The
decisio	n was as follo	WS:							
(Read t	he Ethics Hea	ırinα Panel Γ	ecision, SIOR	P-PS Form 11	1 into th	ne record)			
(11000		ing ranor	00000011, 01011		.,	10 100014.)			
Prior to	the original e	thics hearin	g, the compla	int was revie	ewed by	SIOR's G	rievance Re	eview Panel	and
referre	d to SIOR's Pr	ofessional S	tandards Com	mittee for a	hearing.				
- .									
			nt here today i						
		NDUSTRIAL A	AND OFFICE R	EALTORS® ar	na SIOF	R's Etnics,	Mediation	and Arbitra	ition
Proced	lures Manual.								
At this t	ime. I would li	ke to introdu	ce the membe	ers of this An	neal He	aring Pane	I		
,	o, i wodia			7.0 0. tillo 7 tp	pour i io	anng r ano	••		
My nar	ne is				, and	d I will ser	ve as Cha	ir of this App	peal
-	g Panel.								,
The	other meml	pers of	this panel	are:					,
			.,	-				,	and

Anyone who was a party to the original ethics hearing, a member of the Grievance Review Panel that reviewed this complaint, a member of the original Hearing Panel, or who was otherwise directly involved in this matter prior to the appeal, is disqualified from serving on the Appeal Hearing Panel. Furthermore, anyone who is related by blood or other family relationship to any party to the hearing, or who is an employer, partner, employee, directly associated in business with any of the parties, or affiliated with any person or entity who is directly associated in business with any of the parties, is also disqualified. A potential member of an Appeal Hearing Panel also will be disqualified if she/he is a party to the hearing or a party or a witness in any other pending case involving the complainant or the respondent, or in any other similar relationship or has knowledge which may prevent him/her from rendering an impartial decision. I will now ask the members of the Appeal Hearing Panel if they know of any reason why they should be disqualified. (*If none*): Let the record show that the members of the Appeal Hearing Panel have

not indicated any reason why they are not qualified to serve. Although the parties have been previously notified of their right to challenge members of this Appeal Hearing Panel, I will now ask the parties if they are aware of any reason why any member of this Appeal Hearing Panel is not qualified to serve. (*If none*) Let the record show that the parties have not indicated any reason why any member of this Appeal Hearing Panel is not qualified to serve.

Present at this appeal hearing are, the complainar				
the original ethics hearing and				
	, the respondent in the original	al ethics hearing and his/her		
counsel (if any)		original ethics hearing Chair,		
	; SIOR's	legal counsel,		
	; and SIOR's Profess	ional Standards Staff Officer,		
APPEAL LIMITATIONS In this appeal bearing we will consider	dor the arguments of	horoinaftar		
	der the arguments ofed a Request for Appeal. The appeal Appeal, which is as follows:			
(Read the Request for Appeal of an	Ethics Hearing Decision, SIOR-PS Fo	orm 18, into the record.)		

GUIDELINES

The Appeal Hearing Panel is not governed by the technical rules of evidence which may apply in courts of law. This Appeal Hearing Panel will seek to determine all ascertainable and relevant facts pertaining to the matter under consideration to arrive at a peer judgment and decision by the Appeal Hearing Panel that is fair to all of the parties. All parties or their representatives to these proceedings will be allowed a full opportunity to be heard on matters relevant to the issue. The Appeal Hearing panel may rule at any time during the appeal hearing on the relevance of testimony being given. All parties giving testimony will be asked to affirm that their testimony is the truth to the best of their knowledge. A party may be represented by legal counsel. The Appeal Hearing Panel need not accept the statements of counsel as being the statements of their clients if this Appeal Hearing Panel desires direct testimony. Counsel is present to advise and consult with their clients, and to speak for them, subject to appropriate rulings or determinations by the Appeal Hearing Panel. This Appeal Hearing Panel will countenance no effort by any party or by counsel to any party to harass, intimidate, coerce, or confuse the Appeal Hearing Panel or any party to the proceedings. After the Appeal Hearing Panel has heard all of the evidence and testimony, we will go into executive session to render our decision, by majority vote. The decision will be based solely upon the arguments, evidence, and testimony offered during this appeal hearing.

OUTLINE OF PROCEDURE

At this time, I would like to explain the order of proceedings for this appeal hearing. First, the original ethics Hearing Panel Chair (or another representative from the original panel) will summarize the case (or provide copies of transcripts). Then, the appellant will have an opportunity to offer any corrections or modifications to the summary (or transcripts), to introduce supporting evidence and/or the statements of any other parties to the original ethics hearing, and to explain the basis for the appeal. Next, the original Hearing Panel Chair and/or other parties to the original ethics hearing will have an opportunity to explain why the original ethics Hearing Panel's decision should be upheld. At any time during this proceeding, members of the Appeal Hearing Panel may ask questions. Finally, following the appeal hearing, the Appeal Hearing Panel will go into executive session to render a decision. Do any of the parties have any questions regarding the outline of the procedure? (If none) Let the record show that none of the parties has any questions regarding the outline of the procedure.

(Ask each party and witness to affirm.)

Raise your right hand, and following the question I now pose, answer in the affirmative if you do so affirm: Do you affirm that the testimony you are about to give in this proceeding will be the truth, the whole truth, and nothing but the truth?

(If all parties affirm) Let the record show that all parties have answered in the affirmative. Please be seated.

APPEAL HEARING

We shall now proceed with the appeal hearing.

PRESENTATION BY THE ORIGINAL ETHICS HEARING PANEL CHAIR

At this time, the original ethics Hearing Panel Chair will summarize the case (or provide copies of transcripts).

PRESENTATION BY THE APPELLANT

At this time, the appellant may offer any corrections or modifications to the original ethics Hearing Panel Chair's summary, introduce supporting evidence and/or the statements of any other parties to the original ethics hearing, and explain the basis for this appeal.

REBUTTAL BY THE ORIGINAL ETHICS HEARING CHAIR

At this time the original ethics hearing Chair will have an opportunity to explain why the original ethics Hearing Panel's decision should be upheld by this Appeal Hearing Panel.

TESTIMONY OF THE OTHER PARTIES TO THE ORIGINAL ETHICS HEARING

At this time, any other parties to the original ethics hearing will have an opportunity to address whether the decision of the ethics Hearing Panel should be upheld by the Appeal Hearing Panel. Any party testifying must restrict his/her discussion to the issues raised in the appeal.

CONFIRMATION OF FAIRNESS/OPPORTUNITY TO TESTIFY

Does each of you feel that this hearing has been conducted fairly? (*If yes*) Let the record show that all parties to this appeal indicated that they feel this appeal hearing has been conducted fairly. Have each of you had adequate opportunity to state why you believe that the decision of the original ethics Hearing Panel should or should not be upheld by this Appeal Hearing Panel? (*If yes*) Let the record show that both the complainant and the respondent have indicated that they have had the opportunity to state why they believe that the decision of the original ethics Hearing Panel should or should not be upheld by this Appeal Hearing Panel. (*If any party answers "no," ask him/her to explain his/her concern for the record.*)

CLOSING STATEMENT

The decision of this appeal Hearing Panel is final. Before we adjourn this appeal hearing, all persons present are advised that the report and findings of this Appeal Hearing Panel are considered confidential. Upon final action by this Appeal Hearing Panel, the SIOR president will disseminate to the complainant and the respondent in the original ethics hearing, the Board of Directors, the Chair and members of the original ethics Hearing Panel, SIOR legal counsel, and any governmental agency or real estate professional organization, as directed by the Board of Directors, such notice as the president deems appropriate under the circumstances to effectuate any discipline prescribed, provided, however, that the nature, form, content, and extent of the notice will be specifically approved by SIOR legal counsel prior to dissemination.

There being no further business to be considered in this appeal hearing, this hearing stands adjourned.

EXECUTIVE SESSION

After adjournment, the Appeal Hearing Panel will enter executive session to examine the issues raised on appeal. The Appeal Hearing Panel may adopt or modify the recommendation of the original Hearing Panel, including the discipline imposed, or, alternatively, the Appeal Hearing Panel may dismiss the complaint if it concludes that the findings of fact do not support the original Hearing Panel's conclusion as to unethical conduct. If the Appeal Hearing Panel is concerned with a possible procedural deficiency or error, it may refer the case back to the Professional Standards Committee Chair for a new hearing before a different ethics Hearing Panel. If the Appeal Hearing Panel is concerned with the appropriateness of the recommendation for discipline, it may impose alternative discipline that does not exceed that discipline recommended by the original ethics Hearing Panel or, alternatively, the Appeal Hearing Panel may refer the decision back to the original ethics Hearing Panel for further consideration. If the matter is referred back to the original ethics Hearing Panel for further consideration, it should be accompanied by the Appeal Hearing Panel's concerns regarding the proposed discipline. However, while the Appeal Hearing Panel may recommend that the discipline be increased, the original ethics Hearing Panel is not required to increase the discipline beyond that originally recommended. The Appeal Hearing Panel's decision will be reported using SIOR-PS Form 12. "Ethics Hearing Action by the Board of Directors."

SOCIETY OF INDUSTRIAL AND OFFICE REALTORS® (SIOR) Ethics, Mediation & Arbitration Procedures Manual

PART III: MEDIATION AND ARBITRATION OF FINANCIAL DISPUTES

A.	Duty to Mediate/Arbitrate	33
В.	Manner of Invoking Mediation/Arbitration	33
C.	Timely Filing of Mediation and Arbitration Requests	33
D.	Review of a Mediation/Arbitration Request	33
E.	Mediation by SIOR of Arbitrable Disputes	36
F.	Arbitration Hearing Preparations	37
G.	Postponement of a Hearing	38
H.	Conduct of an Arbitration Hearing	38
I.	Hearing Format	38
J.	Role of Legal Counsel	39
K.	SIOR's "Arbitration Guidelines"	39
L.	Settlement	41
M.	Decision and Award Following Hearing	41
N.	Procedural Review by the Board of Directors	41
Ο.	Confidentiality of Arbitration Proceedings and Results	43
P.	Enforcement of Awards	43
HEA	ARING PANEL CHAIR PROCEDURAL GUIDE FOR ARBITRATION HEARINGS	44
	ARING PANEL CHAIR PROCEDURAL GUIDE FOR ARBITRATION PROCEDURAL REVIEW	48

PART III: MEDIATION AND ARBITRATION OF FINANCIAL DISPUTES

A. Duty to Mediate/Arbitrate

Principle 16 of the Code of Ethical Standards and Principles of Professional Practice calls for mandatory mediation and arbitration of financial disputes when they involve SIOR Adherents from different firms. (See Standard of Practice 16.1.) An arbitration decision is binding on the parties. Under Principle 16, all affected parties may otherwise agree to a voluntary settlement of their dispute, and settlement is encouraged. The mediation and arbitration process will be conducted in accordance with the procedures and policies set forth in this manual.

B. Manner of Invoking Mediation/Arbitration

Any person authorized by Principle 16 of the *Code of Ethical Standards and Principles of Professional Practice* may request mediation and/or arbitration by SIOR, provided that such person had not previously initiated litigation concerning the controversy and amounts in dispute, regardless of the pendency or, if already adjudicated, the outcome, of any such litigation.

Mediation is the required first step in dispute resolution, which is invoked by filing a Mediation/Arbitration Request (SIOR-PS Form 3). This form must be signed by the complainant, must indicate the nature of the controversy and the amount in dispute, and must be accompanied by the required filing fee, currently \$3,500. An additional \$4,000 filing fee, of which \$500 is a nonrefundable administrative fee, will be imposed if mediation fails and arbitration is required. (Where one or more parties from the same firm are involved in more than one related request for arbitration, and the claims are consolidated and resolved in a single hearing, no more than one fee will be required of those parties. When more than one claimant is involved in a request for mediation/arbitration, which results in one of the claimant's filing a separate arbitration petition to determine the distribution of the disputed funds, any claimant making no claim to the funds shall not be assessed an arbitration filing fee.)

SIOR Adherents' clients or non-SIOR member real estate professionals involved in a dispute with a SIOR Adherent may also request mediation/arbitration within SIOR's system, provided that they agree to be bound by an arbitration decision. SIOR Adherents may also invoke mediation/arbitration against clients or non-member real estate professionals, but no mediation or arbitration may be held without the client or non-member's voluntary agreement to mediate/arbitrate and be bound by an arbitration decision. (See Standards of Practice 16.2. and 16.3.)

C. Timely Filing of Mediation and Arbitration Requests

Requests for the mediation/arbitration of financial disputes in which an Adherent is involved must be filed within one (1) year of the later of: (a) the date of the closing of a sale from which the matter arises or to which it is related; (b) the commencement date of a lease (or its renewal or extension) from which it is related; (c) the rental commencement date of a lease (or its renewal or extension) from which the matter arises or to which it is related; or (d) the date on which the basis of the dispute is known, or could be known, in the exercise of reasonable diligence.

D. Review of a Mediation/Arbitration Request

Upon receipt of a completed Mediation/Arbitration Request (SIOR-PS Form 3), the Professional Standards Staff Officer will promptly refer the request to the Professional Standards Committee Chair, who will supervise assembly of a file to facilitate a Grievance Review Panel.

The Professional Standards Staff Officer shall promptly request a written response from the party complained of (called the "Respondent") by sending a copy of the complaint to the respondent and requesting the respondent to submit a response (see SIOR-PS Forms 5, "Request for Information," and 7, "Response to Request for Mediation/Arbitration"). The respondent will be advised that failure to respond

to the complaint within thirty (30) days may result in the complaint being forwarded for a hearing and may subject the respondent to a charge of violating the *Code of Ethical Principles and Standards of Professional Practice* for failing to submit pertinent facts to an appropriate body. If no response is filed within the time allotted, a Grievance Review Panel will be formed to make a determination as to whether arbitration is appropriate and whether a hearing should be scheduled based upon the information set forth in the request.

Within fifteen (15) days of receipt of a response from the respondent, the Professional Standards Staff Officer will provide a copy of the response to the complainant.

A Grievance Review Panel will be appointed by the Professional Standards Committee Chair which shall consist of no less than three (3) nor more than five (5) members (but must be an odd number) of the Professional Standards Committee. The function of the Grievance Review Panel is to make only such preliminary investigation and evaluation of a request as required to determine whether the matter is properly subject to mediation/arbitration and whether the proper parties are named in the request. The Grievance Review Panel does not hold hearings and does not determine entitlement to awards.

1. Initial Action for Mediation/Arbitration Review

The Professional Standards Committee Chair will schedule a meeting of the Grievance Review Panel, which may meet by conference call. The Professional Standards Staff Officer will provide members of the Grievance Review Panel with copies of the case file, including copies of the complaint and the reviewer's reports, if any, prior to the meeting. The Grievance Review Panel may, at the discretion of the Chair, also be provided with a copy of the response to consider it for purposes of evaluating the request.

2. Consideration of a Mediation/Arbitration Request by the Grievance Review Panel

In reviewing a mediation/ arbitration request, the Grievance Review Panel will consider the following:

a. Is the request acceptable in the form as received by the Panel?

If not in proper form, the Grievance Review Panel may request that the Professional Standards Staff Officer contact the complainant to advise that the request must be submitted in proper form.

Note: If deemed appropriate, a member of the Grievance Review Panel may be assigned to contact the complainant and provide procedural assistance to amend the request or resubmit a new request in proper form and with proper content. The Grievance Review Panel member providing such assistance will ensure that only procedural assistance is provided to the complainant and that the complainant understands that the member is not representing the complainant.

- b. Are all necessary parties named in the request for mediation/arbitration?
- c. Was the request for mediation/arbitration filed within one (1) year of the later of: (i) the date of the closing of a sale from which the matter arises or to which it is related; (ii) the commencement date of a lease (or its renewal or extension) from which the matter arises or to which it is related; the rental commencement date of a lease (or its renewal or extension) from which the matter arises or to which it is related; or (iv) the date on which the basis of the dispute is known, or could be known, in the exercise of reasonable diligence?
- d. Are the parties entitled to invoke mediation/arbitration through SIOR's facilities? Were the parties so entitled at the time the facts giving rise to the dispute occurred?

- e. Is litigation pending, or has litigation been finally adjudicated, in connection with the same transaction?
 - Note: No mediation or arbitration will be provided on matters in pending litigation unless the court refers the matter to SIOR for mediation/arbitration. Adherents have a duty under Principle 16 of the Code of Ethical Standards and Principles of Professional Practice not to litigate matters that they should mediate/arbitrate.
- f. Is there any reason to conclude that SIOR would be unable to provide an impartial Hearing Panel?
- g. If the facts alleged in the request for mediation/arbitration were taken as true on their face, is the matter at issue a financial dispute related to a real estate-related professional or business activity,
 - and is it properly subject to mediation/arbitration, e.g., is there some basis on which an arbitration award could be based?
- h. If an arbitrable issue exists, are the parties required to arbitrate or is their participation voluntary? Where participation is voluntary, have those parties agreed to in writing be bound by the arbitration outcome or award?
- i. Is the amount in dispute too small for SIOR to mediate/arbitrate?
- j. Is the matter too legally complex, involving central issues that the mediator/arbitrators may not be able to address in a knowledgeable way?
- k. Are there a sufficient number of knowledgeable mediators/arbitrators available?

If all of the relevant questions have been considered, and a majority of the Grievance Review Panel concludes that the matter is properly subject to mediation/arbitration by SIOR, the Grievance Review Panel will send the request to the Professional Standards Committee Chair for initiation of mediation procedures.

3. Appeals from the Decision of the Grievance Review Panel Related to a Mediation/Arbitration Request

If the Grievance Review Panel determines that a matter should not be mediated/arbitrated by SIOR due to the parties to the dispute, the amount involved, the legal complexity, or for any other reason specified in the Grievance Review Panel decision, either of the parties may appeal the decision to the Board of Directors within twenty (20) days of the date of notice of the Grievance Review Panel decision.

Only those materials which were presented to the Grievance Review Panel when the Grievance Review Panel made its decision will be presented to the Board of Directors and considered with the appeal. The parties to the arbitration (complainant and respondent) do not have the right to appear at the appeal hearing before the Board of Directors. The SIOR President may appoint a panel of directors, acting on behalf of the Board of Directors, to hear the appeal. Any appeal panel so appointed must be composed of at least five (5) directors. Alternatively, the appeal may be heard by SIOR's Governance Committee. The decision of the appeal panel or Governance Committee is final and not subject to further review by the Board of Directors.

- 4. If the Grievance Review Panel finds the matter properly subject to mediation/arbitration, the Chair will contact both the complainant(s) and the respondent(s) within five (5) business days of receipt of the Grievance Review Panel's findings to implement mediation procedures.
- 5. If the Grievance Review Panel dismisses the case, the deposit will be returned to the complainant, less a \$500 fee for administrative expenses.

E. Mediation by SIOR of Arbitrable Disputes

A completed SIOR-PS Form 7 agreeing to mediate and arbitrate must be accompanied by a \$3,500 fee to help cover mediator expenses. The filing fee of \$3,500 shall be used for expenses incurred in the mediation process. If the costs of mediation are less than the filing fees, the excess funds shall be placed in an escrow account for use to cover costs of other mediation services. If the costs exceed the filing fees for mediation there will be no claim by SIOR for additional fees, and any available escrow funds will be used to offset the additional costs.

If mediation fails to resolve the dispute, the matter will be handled through arbitration procedures and both parties to the dispute will be assessed an additional \$4,000 fee to help cover arbitration expenses.

The Professional Standards Staff Officer will send each party a SIOR-PS Form 17, "Mediation Officer Selection Form," on which will be listed the names of several potential mediators identified by the Professional Standards Committee Chair. Each party will be requested to return that form indicating whether the listed persons are acceptable or not acceptable. A mediation officer will be designated by the Professional Standards Committee Chair from among the listed persons shown as acceptable to all parties. Whenever possible, mediation officers will be designated from Active members who have substantial training and/or experience in matters relevant to the mediation in question. The parties will be notified by the Professional Standards Staff Officer, using SIOR-PS Form 17A, "Mediation Officer Selection Notice," as to the identity of the mediation officer that is appointed by the Professional Standards Committee Chair. The parties will then be contacted by the mediation officer to arrange a mediation conference which may be in the form of a meeting or a teleconference. Every effort will be made to ensure that the time and conference arrangements are convenient for all the parties.

The mediation officer will attempt to resolve the dispute prior to its referral to an arbitration panel. No party to an arbitrable matter is required to agree to be bound by any determination reached through mediation. Mediation is not intended to be a substitute for the arbitration procedures described in this manual. Mediation is required as a service to SIOR Adherents and other parties to resolve business financial disputes.

Although mediation is not binding upon the parties, the need for procedural fairness remains. There generally is no need for the parties to be represented by legal counsel; however, parties to the mediation may be accompanied by and represented by legal counsel. Parties intending to have legal counsel present at the mediation conference should so advise the mediation officer or the Professional Standards Staff Officer and the other parties at least ten (10) calendar days prior to the conference.

No aspect of a mediation conference may be relied upon or introduced as evidence in any arbitration, judicial, or other proceeding, including but not limited to views expressed or suggestions made by the parties with respect to possible settlement of the dispute, admissions made in the course of the mediation, proposals made or views expressed by the mediation officer, or the responses of any party to any of the aforementioned. No privilege will be affected by disclosures made in the course of mediation. No records, reports, or other documents received or prepared by the mediator will be released to anyone after the mediation is completed. Neither SIOR nor the mediation officer may be compelled to disclose or testify in any proceeding as to information disclosed or representations made in the course of the mediation or confidential communication to the mediation officer.

If a mediated resolution of a dispute is reached, each party will be asked to sign a Mediation Resolution Agreement (SIOR-PS Form 16) as drawn by the mediation officer. If one or both parties fail to sign the Mediation Resolution Agreement and return it to the mediation officer within the time specified by the mediation officer, or if the mediation officer determines that mediation will not successfully resolve the dispute, the matter will be referred to the Professional Standards Committee Chair to be scheduled for an arbitration hearing. The SIOR Member who serves as the mediator may not serve on the arbitration panel.

Experience has shown that the vast majority of business disputes resolved within REALTOR ^(R)-related organizations can be settled at the mediation stage. Most parties prefer to settle their disputes via mediation, as they maintain control of the process throughout and any settlement that is reached is voluntary in nature, and is reached by agreement of both parties. By contrast, in an arbitration hearing, the Hearing Panel controls the proceedings and determines the settlement.

F. Arbitration Hearing Preparations

1. If mediation is unsuccessful, the Professional Standards Staff Officer will send to each of the parties a list of members of the Professional Standards Committee for the purpose of determining whether the parties wish to challenge one or more of those persons serving on a panel arbitrating their dispute. Challenges may be filed within ten (10) days using SIOR-PS Form 1A ("Notice of Right to Challenge Hearing Panel Members"). A person will automatically be disqualified as a member of a Hearing Panel in any proceeding in which the person is (a) related by blood or other familial relationship to a party; or (b) an employer, partner, employee, or directly associated in business with a party; or affiliated with any person or entity who is directly associated in business with a party; (c) a party to the hearing, or a party or a witness in any other pending case involving a party to the hearing; or (d) involved in any other relationship or has knowledge which may prevent him/her from rendering an impartial decision. Other grounds for disqualification may also be asserted and will be considered in light of their nature and the supporting evidence offered. The validity and legitimacy of challenges will be determined by the Professional Standards Committee Chair.

The Professional Standards Committee Chair will appoint, from the names not successfully challenged by either party, three (3) arbitrators who will hear the dispute. The Chair will also select one of the panel members to serve as of the Hearing Panel. Any Hearing Panel must have an odd number of members. No person who has been a member of the Grievance Review Panel on the arbitration may be appointed to the Hearing Panel. It is a membership duty of anyone so appointed to serve as an arbitrator unless disqualified. Parties will be deemed to have waived all objections to any potential panelist whom they do not challenge. If challenges to members of the Professional Standards Committee result in there being an insufficient number of members to constitute a panel, SIOR's president may appoint other qualified SIOR members to serve as panel members. No arbitration may proceed without three (3) or more arbitrators. Each person chosen as a Hearing Panel member will promptly complete and sign a Certificate of Hearing Panel Qualification (SIOR-PS Form 1).

- 2. The Professional Standards Staff Officer will inform the parties of the date, time, and place of the hearing established by the arbitrators or the Professional Standards Committee Chair (on SIOR-PS Form 9, "Official Notice of Arbitration Hearing"). The parties will be given at least thirty (30) days prior notice of the scheduled hearing, but appearance at a hearing without objection by a party will constitute a waiver of any defective notice of the hearing. Hearing procedure guidelines (SIOR-PS Form 10, "Outline of Procedure for Ethics or Arbitration Hearings") and Arbitration Guidelines (SIOR-PS Form 14) will accompany the notice of hearing to the parties. The arbitration request and response, as well as supporting materials, will be provided to Hearing Panel members at least ten (10) business days prior to the scheduled hearing. The arbitrators may recess the hearing from time to time as necessary and, on request of a party or upon the arbitrators' own motion, may postpone the hearing until the next semi-annual meeting or such other date as is agreed to by the parties.
- 3. Upon notice by the staff officer, the parties to the dispute will, with diligence, present in writing for use by the arbitrators such statements and proof as they deem appropriate to support their claims or positions. Proof may be submitted in the form of affidavits or otherwise. The arbitration Hearing Panel Chair may require that statements be verified by affidavits or that accuracy or authenticity of any documents or other papers submitted be verified by sworn affidavit. At the hearing, the arbitrators will receive any further written statements, documents, or other papers, will hear oral testimony, and will determine what personal appearances will be made by the parties and regulate the holding of hearings. The Hearing Panel may receive and consider any evidence it deems material and proper,

including evidence of accountants and other experts. Each party is responsible for the expenses of witnesses it calls. Parties to arbitrations are entitled to have legal counsel present at any hearing. Each party is responsible for the expenses of its own respective counsel.

If any party, witness, or counsel involved in the proceedings requires accommodation because of a disability, SIOR will make reasonable efforts to provide such accommodation. Individuals requiring such accommodation must contact the Professional Standards Staff Officer at least fifteen (15) days before the hearing.

G. Postponement of a Hearing

Postponement may be granted if there are extenuating circumstances. Requests for postponement must be made in writing to the Hearing Panel Chair or the Professional Standards Staff Officer. Postponement requests can be granted by the Hearing Panel Chair. All parties will be advised of the date of the rescheduled hearing at least ten (10) days in advance of the rescheduled hearing.

H. Conduct of an Arbitration Hearing

An arbitration hearing involves a series of procedural steps or events that are coordinated by the Hearing Panel Chair. While each proceeding will present somewhat different circumstances, the Hearing Panel Chair will apply the standard procedures in a manner to accommodate the fair and efficient presentation of the parties' respective cases. In most cases, the hearing will generally follow the procedural steps outlined in the "Hearing Panel Chair Procedural Guide for Arbitration Hearings," which appears as Annex A at the end of this part of the manual. It is essentially a form that the panel Chair uses as a starting point in preparing to direct the hearing. The parties are urged to consider these procedures and present any resulting issues or requests in writing to the Professional Standards Staff Officer well prior to the start of the hearing, so that the resolution of these issues will not cause unnecessary delay or inconvenience to other participants.

SIOR will, and any party may at its own expense, have a court reporter present at hearings, or may record hearings. If a party prepares a transcript, that party must provide a copy to the Professional Standards Staff Officer, who may provide a copy thereof to all other parties.

I. Hearing Format

An arbitration Hearing Panel will not be bound by the rules of evidence applicable in courts of law, but will afford all parties a reasonably full opportunity to be heard, present witnesses, and offer evidence.

The arbitration hearing procedure will follow these general guidelines:

- 1. Opening statement by the Hearing Panel Chair, citing the panel's authority to hear the case and explaining the reasons for the hearing.
- 2. The arbitration request or petition will be read into the record.
- 3. The testimony of all parties and witnesses will be affirmed.
- 4. The parties will be given an opportunity to present evidence and testimony on their behalf. All witnesses will be excused from the hearing except while testifying.
- 5. The parties will be afforded an opportunity to examine and cross-examine all witnesses and parties.
- 6. Panel members may ask questions at any time during the hearing.
- 7. The panel Chair may exclude any questions which the Chair deems irrelevant or argumentative.

- 8. Each side may make a closing statement. The complainant will make the first closing statement and the respondent will make the final closing statement.
- 9. The hearing will adjourn.
- 10. The Hearing Panel will go into executive session to decide the case.

J. Role of Legal Counsel

A party may be represented in any arbitration hearing by legal counsel. However, parties may not refuse to respond directly to requests for information or questions addressed to them by members of the Hearing Panel, except on the grounds of self-incrimination, or on other grounds deemed by the panel to be appropriate. In this connection, the panel need not accept the statements of legal counsel as being statements of counsel's client if the panel desires direct testimony. Parties will be held responsible for the conduct of their legal counsel. Any effort by legal counsel to harass, intimidate, coerce, or confuse the panel members or any party to the proceedings or any action by legal counsel which is viewed by the panel as disruptive of the proceedings, will be grounds for exclusion of legal counsel from the hearing. The decision to exclude legal counsel for any of the foregoing reasons must be based on a majority vote of the members of the panel and may not be appealed. In the event legal counsel is excluded, the hearing will be postponed, to a date certain not less than thirty (30) nor more than ninety (90) days from the date of adjournment, to enable the party to obtain alternate legal counsel, provided, however, that such postponement will not be authorized if it appears to members of the panel that the action of legal counsel has been undertaken to obtain a postponement or delay of the hearing.

K. SIOR's "Arbitration Guidelines"

All parties to arbitration should become familiar with SIOR's Arbitration Guidelines (SIOR-PS Form 14), because they provide Hearing Panels with guidance as to general issues and questions involved in many arbitration hearings.

1. Procuring Cause

Determination of "procuring cause" is often the foundation of an arbitration dispute over entitlement to a commission or a portion of a commission. While a number of definitions of procuring cause exist, and a myriad of factors may ultimately enter into any determination of procuring cause, for purposes of arbitration conducted within the Society of Industrial and Office Realtors®, procuring cause can be understood as the uninterrupted series of causal events which results in a successful real estate transaction. Or, in other words, what causes a successful real estate transaction to occur.

Many have tried to define a single, comprehensive template that can be used in all procuring cause disputes to determine entitlement to a commission in dispute without a comprehensive analysis of all relevant details of an underlying transaction. Unfortunately, there is no typical transaction, and fair and equitable decisions require a comprehensive understanding of the events that led to a transaction.

2. Factors for Consideration by Arbitration Hearing Panels

a. No Predetermined Rule of Entitlement

Every arbitration hearing is considered in light of all of the relevant facts and circumstances as presented by the parties and their witnesses. Rules of thumb, prior decisions by other panels in other matters, and other predeterminants must be disregarded.

b. Arbitrability and Appropriate Parties

While primarily the responsibility of the Grievance Review Panel, arbitration Hearing Panels may consider in light of the evidence (as distinguished from the mere allegations) questions of whether an arbitrable issue actually exists, and whether the parties named are appropriate to arbitration.

c. Relevance and Admissibility

Frequently, Hearing Panels are asked to rule on questions of admissibility and relevancy. The general rule states that anything a Hearing Panel believes may assist it in reaching a fair, equitable, and knowledgeable decision is admissible. If a Hearing Panel believes that an SIOR Adherent involved in the arbitration may have violated the Code of Ethical Principles and Standards of Professional Practice, the panel may cite the relevant testimony and refer it to the Professional Standards Committee Chair for consideration in the same manner as other alleged violations of the Code of Ethical Principles and Standards of Professional Practice. Under no circumstances may an ethics hearing be conducted, based on concerns identified by an arbitration Hearing Panel, until after the arbitration process has been completed and the award, if any, has been made.

d. Communication and Contact — Abandonment and Estrangement

Many arbitrable disputes turn on the relationship, or lack thereof, between a broker (often a cooperating broker) and a prospective buyer/lessee. Panels need to consider whether, under the circumstances and in accord with general custom and practice, a broker made reasonable efforts to develop and maintain an ongoing relationship with the buyer/lessee.

Panels need to determine, in cases where two cooperating brokers have competing claims against a listing broker, whether the first cooperating broker actively maintained ongoing contact with the buyer/lessee or, alternatively, whether the broker's inactivity, or perceived inactivity, may have caused the buyer/lessee to reasonably conclude that the broker had lost interest or disengaged from the transaction (abandonment). In other instances, a buyer/lessee, despite reasonable efforts by a broker to maintain ongoing contact, may seek assistance from another broker.

The panel needs to consider why the buyer/lessee abandoned the first broker. In still other instances, there may be no question that there was an ongoing relationship between a broker and a buyer/lessee; the issue then becomes whether the broker engaged in conduct which caused the buyer/lessee to terminate the relationship (estrangement). Panels need to consider whether such conduct caused a break in the series of events leading to a transaction and whether the successful transaction was actually brought about through the initiation of a separate, subsequent series of events by the second cooperating broker.

In assessing issues in this area, panels should be mindful of the rules imposed on Adherents by Code Principle 14 ("Do Not Interfere with the Relationships or Rights of Other Real Estate Professionals"), Standard of Practice 14.1 ("Abandonment"), Standard of Practice 14.2 ("Alienation and Estrangement"), Standard of Practice 14.3 ("Termination"), Standard of Practice 14.4 ("No Other Exceptions"), and Standard of Practice 14.5 ("Compensation Rights").

e. Conformity with Law

The procedures by which arbitration requests are reviewed, hearings are conducted, and awards are made should be selected in conformity with governing law. The parties have the obligation to explain their perception of the requirements of applicable law. The advice of SIOR legal counsel may be sought in any case where the panel believes that the procedural requirements of law may be critical or where the parties insist that the procedures being followed are not adequate.

f. Consideration of the Entire Course of Events

The standard of proof in arbitration hearings is a preponderance of the evidence and the initial burden of proof rests with the party requesting arbitration. This does not, however, preclude panel members from asking questions of the parties or witnesses to confirm their understanding of testimony presented or to ensure that panel members have a clear understanding of the events that led to the transaction and to the request for arbitration.

L. Settlement

The parties to an arbitration may settle the issue between them by agreement at any time. In such event, upon notification to the Professional Standards Staff Officer, the arbitration proceedings will be terminated and the termination will be recorded in the file.

A portion or all of each party's deposit may be retained by SIOR to cover the costs incurred by SIOR up to the point of settlement of the dispute. The decision on that question will be made by the Professional Standards Committee Chair.

M. Decision and Award Following Hearing

- 1. The award of the arbitrators (SIOR-PS Form 13, "Arbitration Decision and Award") will be made as soon as possible after the hearing is concluded. The award will be in writing and signed by the arbitrators or a majority of them that joined in the decision, will state only the amount of the award, and when so signed and provided to each of the parties, will be valid and binding. Notwithstanding the foregoing, a party to an arbitration proceeding may appeal to the Board of Directors. Such appeals may be based only on claims of procedural deficiency or procedural irregularities.
- 2. The award will direct the disposition of the arbitration fees paid by the parties. Any amount retained from the deposits may be used by the staff officer to cover the costs of the arbitration. Any portion not used specifically to cover the costs of the arbitration will go into the general operating funds of SIOR. In the event the award of the arbitrators is an amount other than that requested by any of the parties, the disposition of the fees will be directed by the arbitrators. When a SIOR Adherent requests arbitration to determine which of multiple respondents is entitled to disputed funds, or where a party makes no claim to the disputed funds, that party shall not be assessed an arbitration filing fee.
- 3. After notice of an award has been sent to each of the parties, they have twenty (20) days to request procedural review of the arbitration hearing procedure as described in this manual. If no such review is requested, the award becomes final and binding following the twenty (20) day period. However, if procedural review is requested, the award is not considered final and binding until completion of Board of Directors review.

N. Procedural Review by the Board of Directors

1. A written request for procedural review of arbitration hearing procedures must be filed with the Professional Standards Staff Officer within twenty (20) days after notice of an award has been served on the parties using SIOR-PS Form 21, "Request for Arbitration Hearing Procedural Review." A request for procedural review must cite the alleged procedural deficiencies or other procedural irregularities the party believes deprived the party of a fair hearing (e.g., exclusion of witnesses, fraud, coercion, bias, prejudice, evident partiality). Only the bases raised in the written request for procedural review may be raised during the review before the Board of Directors. The request for procedural review will be reviewed by the Professional Standards Staff Officer only for the purpose of determining whether the request states an appropriate type of objection for consideration by the Board. If determined to be insufficient, the request will be returned to the requestor accompanied by an explanation and a request for an amendment providing additional detail, to be received by the Professional Standards Staff Officer within ten (10) days. This initial administrative review is not a

- decision on the merits of the request for procedural review but is intended only to ensure compliance with the requirement that the request describe the alleged procedural deficiency or irregularity on which the appeal is based.
- 2. When a request for procedural review (as originally filed if in proper form, as originally filed if no amendment is submitted, or as amended even if still deemed to be lacking) is received, the Professional Standards Staff Officer will immediately send a copy to the other party and provide to the SIOR president a copy of the request for procedural review and the related correspondence, if any. The Professional Standards Staff Officer will notify all parties of the time and place of the review by the Board of Directors at least thirty (30) days in advance. The time set for the hearing will be a time which coincides with the Board's next regular meeting or at a special meeting called for that purpose. Notice will be given using SIOR-PS Form 22, "Official Notice of Arbitration Procedural Review." That notice will be accompanied by SIOR-PS Form 20 ("Notice of Right to Challenge Hearing Panel Members in an Ethics or Arbitration Appeal Hearing"). A challenge may be submitted, using that form, within ten (10) days after the notice is sent. Anyone who is a party to or testified during the original hearing, was a member of the Grievance Review Panel which reviewed this request, was a member of the original Hearing Panel, or who was otherwise directly involved in the matter prior to the appeal. is disqualified from serving on the Appeal Hearing Panel. Furthermore, anyone who is related by blood or other family relationship to any party to the hearing, or who is an employer, partner, employee, or directly associated in business with any of the parties, or affiliated with any person or entity who is directly associated in business with any of the parties, or who is a party to the hearing or a party or a witness in any other pending case involving the complainant or respondent, or who is involved in any other relationship or has knowledge which may prevent him/her from rendering an impartial decision, is also disqualified. A party to the appeal may also cite any other ground that may prevent an appeal panel member from rendering an impartial decision. Challenges will be ruled upon by the SIOR President.
- 3. Any party may be represented by legal counsel and each party is requested to provide written notice of an intention to have counsel present at least ten (10) days before the scheduled hearing date. Failure to provide this notice will not invalidate a party's right to legal representation but may result in a continuance of the hearing if the appeal panel determines that the rights of any other party require representation by legal counsel, or if such other party requests a continuance in order to retain counsel.
- 4. The request for procedural review may be heard by a panel of directors appointed by the SIOR president for that purpose (or, alternatively, by SIOR's Governance Committee). Five (5) directors will constitute such a procedural review panel, which will act on behalf of the Board of Directors. The decision of the panel (or Governance Committee) will be final and binding and will not be subject to further review by the Board of Directors.
- 5. At the procedural review appeal hearing, the party filing the request will have an opportunity to explain the bases on which the party is requesting that the award of the arbitrators be overturned. The Chair of the arbitration panel will have an opportunity to respond to the appellant's statement. The other party will have the opportunity to address the appellant's arguments and support materials. The hearing will be generally organized by and will follow the procedural steps outlined in the "Appeal Panel Chair Procedural Guide for Arbitration Appeal Hearings," which appears as Annex B at the end of this part of the manual. It is essentially a form that the panel Chair uses as a starting point in preparing to direct the hearing. The parties are urged to consider these procedures and present any resulting issues or requests in writing to the Professional Standards Staff Officer well prior to the start of the hearing, so that their resolution will not cause unnecessary delay or inconvenience to other participants. Issues raised in this manner will be ruled upon by the appeal panel Chair.
- 6. The Board of Directors will not hear an appeal with respect to the merits of an arbitration award, and will not, on appeal, review evidence offered with respect to the merits of that award, except as such evidence may bear upon a claim of deprivation of procedural fairness. The directors will render their

decision promptly. This decision may be to adopt the award of the arbitrators or to overturn the award based on a substantial procedural error in the arbitration hearing process that resulted in denial of fair consideration or fair determination of the issues.

7. If the appeal panel determines that a substantial procedural error occurred, which resulted in the denial of due process to one or more parties, the appeal panel may invalidate the original arbitration award and direct that the matter be referred to the Professional Standards Committee for a hearing on the merits before a different Hearing Panel. Alternatively, the appeal panel may release the parties from their obligation to arbitrate if the directors conclude that SIOR will be unable to appoint an impartial Hearing Panel.

O. Confidentiality of Arbitration Proceedings and Results

All members of any Hearing Panel will have an obligation to maintain and protect the confidentiality of the proceedings and deliberations of the Hearing Panel before, during, and after its determinations and recommendations. Hearing Panel members will not discuss Hearing Panel proceedings and deliberations with any person(s) except as required by these procedures or by law.

All other participants in an arbitration proceeding also have an obligation to maintain and protect the confidentiality of the proceedings and their results.

Unauthorized disclosure includes any report or publication under any circumstances not established in this manual. The following are circumstances where disclosure by a party to an ethics and/or arbitration proceeding is authorized:

- i. where the dissemination of a decision to individuals who have some knowledge of the proceeding might vindicate an Adherent's professional reputation; or
- ii. where there is a civil, criminal or administrative proceeding involving the same facts and circumstances which gave rise to the proceedings.

P. Enforcement of Awards

The judgment of any competent court of record, either state or federal, may be rendered upon the award. If an Adherent fails to comply with an award, the party to whom the award has been rendered will be advised by SIOR to seek judicial enforcement and request reimbursement of legal fees incurred in seeking enforcement. At the discretion of the Board of Directors, SIOR may support the request for judicial enforcement in the court.

HEARING PANEL CHAIR PROCEDURAL GUIDE FOR ARBITRATION HEARINGS

The hearing will start promptly at the appointed time. All parties to the hearing are asked to enter the hearing room.

(Recording by court reporter and/or recorder begins.)

CHAIR'S OPENING STATEMENT

Mu nama ia

Ladies and gentlemen, I now call this hearing to order. The Hearing Panel of the Professional Standards Committee is charged under the Bylaws of the Society of Industrial AND Office Realtors® with holding hearings in matters concerning alleged unethical conduct, or in the arbitration of business disputes arising out of the real estate business. Hearings follow the rules described in SIOR's *Ethics, Mediation & Arbitration Procedures Manual.* The body meeting here is an impartial panel of the Professional Standards Committee that has been selected and called here today to ascertain the truth in the particular matter at hand, which is an arbitration proceeding, and to render a decision on the testimony and evidence presented. An ethics proceeding is different from an arbitration proceeding, and the two types of proceedings should not be confused. The particular matter to be considered by this panel at this time is an arbitration proceeding.

The Professional Standards Committee is a body duly constituted under the authority of the Bylaws of The Society of Industrial and Office Realtors®. The Committee members have been duly appointed by SIOR President and approved by the Board of Directors. At this time, I would like to introduce the members of this panel, all of whom are members of the Professional Standards Committee.

and Lyvill comes as Chair of this Hearing

Panel.	, aı	id I will serve as Chair of this realing
The other members of this pa	anel are:	, , and
		, and
Present at this hearing isbusiness associate(s),		
The complainant is accompanied by leg who include	al counsel,	, and witnesses
who include	, and	·
(If the complainant is represented by legal If not, ask if there is an objection. If none,		
Also present at this hearing isbusiness associate(s),		, the respondent, and his/her
The respondent is accompanied by lega who include		
who include	, and	
(If the respondent is represented by legal If not, ask if there is an objection. If none,		
SIOR's legal counsel,		
Officer,	, also are prese	ent today.

This proceeding is being recorded. I ask all parties to cooperate by speaking clearly into the microphones so that we have an accurate and complete record of this proceeding.

BASIS OF THE HEARING

This hearing is to arbitrate a business dispute arising out of the real estate business in accordance with the request of the complainant that he/she be awarded [specify the amount and type of payment in dispute, *such as* commission or part of a commission] in connection with [identify the real estate transaction or the professional assignment giving rise to the claim]. The request for arbitration is as follows:

(Read the Arbitration Request, SIOR-PS Form 3, into the record.)

Any prior offers for settlement or proposed mediation resolutions of this case will not be considered by the Hearing Panel.

This panel is not governed by the technical rules of evidence which may apply in a court of law. This panel will seek to determine all ascertainable and relevant facts pertaining to the matter under consideration to arrive at a peer judgment and decision that is fair to all of the parties. The panel is governed and directed by the Bylaws of the Society of Industrial And Office Realtors®, SIOR's Code of Ethical Principles and Standards of Professional Practice, and SIOR's Ethics, Mediation and Arbitration Procedures Manual. The panel determines its own rules or evidence and its own procedures, with the objectives of equity and due process. The following has been generally accepted and ruled on by this panel as to the procedures to be followed during this hearing:

- All parties and their representatives to these proceedings will be allowed full opportunities to be heard on matters relevant to the issue. The panel may rule at any time during the hearing on the relevance of testimony being given or on questions being directed to any party or his/her representative or to witnesses providing testimony. All parties and witnesses will be asked to affirm that testimony given is the truth to the best of their knowledge.
- 2. (Read only if either or both parties has legal counsel present; otherwise, proceed to 3.)

A party may be represented by legal counsel. However, no party may refuse to respond directly to requests for information or questions addressed to him/her by members of the panel except on grounds of self-incrimination or other grounds which the panel deems appropriate. In this connection, the panel need not accept the statements of legal counsel as being the statements of his/her client if the panel desires direct testimony. Legal counsel is present to advise and consult with his/her client, and to speak for him/her subject to appropriate rulings or determinations by the panel. This panel will countenance no effort by any party or by any legal counsel to any party to harass, intimidate, coerce, or confuse the panel members or any party to the proceedings.

- 3. The panel may rule at any time on the admissibility of evidence. As Chair, I will act as keeper of the evidence introduced at this hearing and mark each exhibit with an exhibit identification number and date.
- 4. The members of this panel are authorized, individually, to ask questions as they deem pertinent and significant at any time during this hearing. To preserve order, I will rule on questions or testimony by the parties and their representatives, or by witnesses in these proceedings. If deemed necessary, I will consult with the members of this panel and with SIOR legal counsel concerning such rulings.

5. At this time, I request that all persons present in the room who expect to testify at this hearing stand and make appropriate affirmation.

(Ask each party and witness to stand and affirm.)

Raise your right hand, and following the question I now pose, answer in the affirmative if you do so affirm:

Do you affirm that the testimony you are about to give in this proceeding will be the truth, the whole truth, and nothing but the truth?

(If all parties affirm) Let the record show that all parties have answered in the affirmative. Please be seated. Witnesses are excused from the hearing at this time until called to provide testimony.

OUTLINE OF PROCEDURE FOR HEARING

Both the complainant and the respondent were sent a copy of the Outline of Procedure for an Ethics or Arbitration Hearing (SIOR-PS Form 10) and SIOR's Arbitration Guidelines (SIOR-PS Form 14). Did each of you receive the Outline and Guidelines? (If yes) Let the record show that both the complainant and respondent have stated that they did receive the Outline and Guidelines. (If no, the party should be given a copy and the Chair should inquire if that party has any objections to proceeding.) Do you have any questions concerning the Outline of Procedure or the Arbitration Guidelines? (If none) Let the record show that neither the complainant nor the respondent has any questions concerning the Outline of Procedure or the Arbitration Guidelines.

We will now proceed with the hearing.

PRESENTATION BY THE COMPLAINANT

The complainant will now state his/her case and present any evidence or witnesses that he/she may desire.

(Before each witness is excused)

CROSS EXAMINATION BY RESPONDENT

QUESTIONS FROM PANEL MEMBERS

PRESENTATION BY THE RESPONDENT

The respondent will now state his/her case and present any evidence or witnesses that he/she may desire.

(Before each witness is excused)

CROSS EXAMINATION BY COMPLAINANT

QUESTIONS FROM PANEL MEMBERS

CLOSING STATEMENTS

At this time, both the complainant and the respondent will be given opportunities to make a summary or closing statement if they so desire. The complainant's closing statement will be heard first.

CLOSING STATEMENT BY THE COMPLAINANT

CLOSING STATEMENT BY THE RESPONDENT

Does each of you feel that this hearing has been conducted fairly? (If yes) Let the record show that both the complainant and the respondent have indicated that they feel this hearing has been conducted fairly. Have each of you had adequate opportunity to testify, present evidence and witnesses, and conduct cross-examination? (If yes) Let the record show that both the complainant and the respondent have indicated that they have had the opportunity to testify, present evidence and witnesses, and conduct cross-examination. (If any party answers "no," ask him/her to explain his/her concern for the record.)

Before we adjourn the hearing, all persons are advised that the award of the panel is considered confidential. It will be available only to members of this panel, the parties, counsel, and staff as required or as otherwise specified in SIOR's Ethics, Mediation and Arbitration Procedures Manual. Upon final action by the Hearing Panel in an arbitration proceeding, the decision, when signed by the members of the Hearing Panel, or a majority of Hearing Panel members, will be served upon the parties to the dispute. The parties will be notified of the decision within the required time after this hearing is adjourned.

There being no further business to be considered in this hearing, this hearing stands adjourned.

EXECUTIVE SESSION

After adjournment, the panel will remain in executive session and determine the award. The panel will follow the procedure set forth in the Ethics, Mediation and Arbitration Procedures Manual, including the Arbitration Guidelines set forth in SIOR-PS Form 14. SIOR counsel should review all decisions of any hearing conducted by a professional standards Hearing Panel involving the resolution of business disputes. Such review should occur prior to any action of the panel becoming final.

HEARING PANEL CHAIR PROCEDURAL GUIDE FOR ARBITRATION PROCEDURAL REVIEW HEARINGS

The hearing will start promptly at the appointed time. All parties to the hearing are asked to enter the hearing room.

CHAIR'S OPENING STATEMENT

Ladies and gentlemen, I now call this procedural review hearing to order. SIOR's *Ethics, Mediation & Arbitration Procedures Manual* provides for the right to a review of the procedures used in arbitration hearings. Both the complainant and the respondent in an arbitration hearing have the right to request a procedural review. This particular request for procedural review will be heard by [the SIOR Board of Directors - or - a panel of directors appointed by the President - or - SIOR's Governance Committee], hereinafter referred to as the procedural review Hearing Panel.

The particular matter to be considered by this procedural review Hearing Panel is a request for review of the hearing procedures used in an arbitration hearing on (date). The original arbitration Hearing Panel was composed of,
, and chaired by
The basis of the original arbitration hearing was an arbitration request filed by, the complainant(s), against, the respondent(s), claiming that a dispute
arising out of the real estate business existed between the complainant(s) and the respondent(s). The Arbitration Decision and Award announced following the arbitration hearing was as follows:
(Read the Arbitration Decision and Award, SIOR-PS Form 13, into the record.)
Prior to the original arbitration hearing, the Arbitration Request was reviewed by a SIOR Grievance Review Panel and referred to SIOR's Professional Standards Committee for a hearing.
The procedural review Hearing Panel present here to day is a body duly constituted under the authority of the Bylaws of Society of Industrial and Office Realtors® and SIOR's <i>Ethics, Mediation & Arbitration Procedures Manual.</i>
At this time, I would like to introduce the members of this appeal Hearing Panel.
My name is, and I will serve as Chair of this procedural review Hearing Panel.
The other members of this panel are:,
, and

Anyone who was a party to the original arbitration hearing, a member of the Grievance Review Panel that reviewed this arbitration request, a member of the original arbitration Hearing Panel, or who was otherwise involved in this matter prior to the appeal, is disqualified from serving on this procedural review Hearing Panel. Furthermore, anyone who is related by blood or other family relationship to any party to the hearing, or who is an employer, partner, employee, or directly associated in business with any of the parties, or affiliated with any person or entity who is directly associated in business with any of the parties, a party to the hearing or a party or a witness in any other pending case involving the complainant or the respondent, or who is involved in any other relationship or has knowledge which may prevent him/her from rendering an impartial decision, also is disqualified. I will now ask the members of the procedural review Hearing Panel if they know of any reason why they should be disqualified. (*If none*) Let the record show that the members of the procedural review Hearing Panel have not indicated any reason why they

are not qualified to serve. Although the parties have been previously notified of their right to challenge members of this procedural review Hearing Panel, I will now ask the parties if they are aware of any reason why any member of this procedural review Hearing Panel is not qualified to serve. (*If none*) Let the record show that the parties have not indicated any reason why any member of this procedural review Hearing Panel is not qualified to serve.

Present at this procedural review hearing are the complainant(s) in the original arbitration he	
,	ng and his/her counsel (if any) the original arbitration Hearing Panel Chair, SIOR's legal counsel, and SIOR's Professional Standards Staff Officer,
The basis for the procedural review is I irregularities the appellant believes prevented	filed a Request for Arbitration Hearing Procedural Review imited to alleged procedural deficiencies or procedural a fair hearing (e.g., exclusion of witnesses, fraud, coercion rocedural review is further limited to the basis(es) set forth

(Read the Request for Arbitration Hearing Procedural Review, SIOR-PS Form 21, into the record.)

GUIDELINES

This procedural review panel is not construing issues of law and is not governed by the technical rules of evidence which may apply in courts of law. This procedural review panel will seek to determine all ascertainable and relevant facts pertaining to the matter under consideration to arrive at a peer judgment and decision that is fair to all of the parties. The decision will be based solely upon the arguments, evidence, and testimony offered during this procedural review hearing. All parties or their representatives to these proceedings will be allowed a full opportunity to be heard on matters relevant to the issue. The procedural review Hearing Panel may rule at any time during this hearing on the relevance of testimony being given. All parties giving testimony will be asked to affirm that testimony given is the truth to the best of their knowledge. A party may be represented by legal counsel. The procedural review Hearing Panel need not accept the statements of counsel as being the statements of their clients if this Hearing Panel desires direct testimony. Counsel are present to advise and consult with their clients, and to speak for them, subject to appropriate rulings or determinations by the procedural review Hearing Panel. This procedural review Hearing Panel will countenance no effort by any party or by counsel to harass, intimidate, coerce, or confuse the appeal Hearing Panel or any party to the proceedings.

OUTLINE OF PROCEDURE

At this time, I would like to explain the order of proceedings for this hearing. First, the appellant will have an opportunity to explain the basis (es) upon which he/she is requesting that the Arbitration Decision and Award be overturned. Next, the original arbitration Hearing Panel Chair will have an opportunity to respond to the matters raised by the appellant. Then, any other party to the original arbitration hearing will have an opportunity to address the appellant's arguments. At any time during this proceeding, members of the panel may ask questions. Finally, following the procedural review hearing, the Appeal Panel will go into executive session to reach a decision. Do any of the parties have any questions regarding that outline of the procedure? (If none) Let the record show that none of the parties has any questions regarding the outline of the procedure.

(Ask each party and witness to affirm.)

Raise your right hand, and following the question I now pose, answer in the affirmative if you do so affirm:

Do you affirm that the testimony you are about to give in this proceeding will be the truth, the whole truth, and nothing but the truth?

(If all parties affirm) Let the record show that all parties have answered in the affirmative. Please be seated.

PROCEDURAL REVIEW HEARING

We shall now proceed with the procedural review hearing.

PRESENTATION BY THE APPELLANT

At this time, the appellant may present the basis (es) upon which he/she is requesting that the Arbitration Decision and Award be overturned.

PRESENTATION BY THE ORIGINAL ARBITRATION HEARING PANEL CHAIR

At this time, the original arbitration Hearing Panel Chair will have an opportunity to demonstrate why the Arbitration Decision and Award should be upheld by the procedural review Hearing Panel.

PRESENTATION BY THE OTHER PARTIES TO THE ORIGINAL ARBITRATION HEARING

At this time, any other parties to the original arbitration hearing will have an opportunity to address whether the Arbitration Decision and Award should be upheld by the procedural review Hearing Panel. Any such party must restrict his/her presentation to the issues raised in the Request for Arbitration Hearing Procedural Review.

CONFIRMATION OF FAIRNESS/OPPORTUNITY TO TESTIFY

Does each of you feel that this hearing has been conducted fairly? (If yes) Let the record show that all parties to this procedural review indicated that they feel this hearing has been conducted fairly. Have each of you had an adequate opportunity to state why you believe that the decision of the original arbitration Hearing Panel should or should not be upheld by this procedural review Hearing Panel? (If yes) Let the record show that the appellant, the original panel Chair and the other party have indicated that they have had the opportunity to state why they believe that the decision of the original arbitration Hearing Panel should or should not be upheld by this procedural review Hearing Panel. (If any party answers "no," ask him/her to explain his/her concern for the record.)

CLOSING STATEMENT

The decision of this procedural review appeal panel is final. Before we adjourn this procedural review hearing, all persons present are advised that the final determination of this panel is considered confidential. Upon final action by this panel, the SIOR President will disseminate to the complainant and the respondent in the original arbitration hearing, the Board of Directors, the Chair and members of the original arbitration Hearing Panel, SIOR legal counsel, and the Professional Standards Staff Officer such notice as the President deems appropriate under the circumstances.

There being no further business to be considered in this procedural review hearing, this hearing stands adjourned. The panel will now go into executive session to render its final decision, which will be based solely on the arguments, evidence, and testimony offered at this hearing.

EXECUTIVE SESSION

After adjournment, the sole issue for the panel to decide is whether the arbitration procedures utilized in rendering the Arbitration Decision and Award afforded the appellant a fair hearing, in light of the objections raised on the SIOR-PS Form 21. If the procedural review Hearing Panel determines that the original arbitration hearing process was defective in any way that prevented a fair hearing, the Arbitration Decision and Award should be invalidated and the matter referred to a different arbitration Hearing Panel for a new hearing. Alternatively, the procedural review Hearing Panel may release the parties from their obligation to arbitrate, if the procedural review Hearing Panel concludes that SIOR will be unable to appoint an impartial arbitration Hearing Panel. Hearing Panel decisions will be made by majority vote.

In the event the respondent improperly declines to arbitrate, the procedures described in Part II under the heading "Initial Action by the Board of Directors for Failure to Arbitrate" become applicable.

SOCIETY OF INDUSTRIAL AND OFFICE REALTORS® (SIOR) Ethics, Mediation & Arbitration Procedures Manual

PART IV: FORMS USED IN PROFESSIONAL STANDARDS

A.	Forms Overview	54
В.	Forms Used in Ethics Enforcement Proceedings (in the order in which they are used)	54
C.	Forms Used in Resolving Financial Disputes (in the order in which they are used)	54
D.	All Forms Listed by Form Number	55
	CERTIFICATE OF HEARING PANEL QUALIFICATION	56
	NOTICE OF RIGHT TO CHALLENGE HEARING PANEL MEMBERS IN ETHICS OR ARBITRATION PROCEEDING	57
	ETHICS COMPLAINT	59
	REPORT OF POTENTIAL ETHICS VIOLATION	60
	MEDIATION/ARBITRATION REQUEST	61
	REQUEST FOR INFORMATION IN RESPONSE TO AN ETHICS COMPLAINT	63
	REQUEST FOR INFORMATION IN RESPONSE TO MEDIATION/ARBITRATION REQUEST	64
	RESPONSE TO ETHICS COMPLAINT	65
	RESPONSE AND AGREEMENT TO MEDIATE OR ARBITRATE	66
	OFFICIAL NOTICE OF ETHICS HEARING	68
	OFFICIAL NOTICE OF ARBITRATION HEARING	69
	OUTLINE OF PROCEDURE FOR ETHICS OR ARBITRATION HEARINGS	70
	ETHICS HEARING PANEL DECISION	74
	ETHICS HEARING ACTION BY THE BOARD OF DIRECTORS	76
	ARBITRATION DECISION AND AWARD	78
	ARBITRATION GUIDELINES	80

MEDIATION RESOLUTION AGREEMENT	84
MEDIATION OFFICER SELECTION FORM	85
MEDIATION OFFICER SELECTION NOTICE	86
REQUEST FOR APPEAL OF AN ETHICS HEARING DECISION	87
OFFICIAL NOTICE OF ETHICS APPEAL HEARING	88
NOTICE OF RIGHT TO CHALLENGE HEARING PANEL MEMBERS IN AN ETHICS OR ARBITRATION APPEAL HEARING	90
REQUEST FOR ARBITRATION HEARING PROCEDURAL REVIEW	91
OFFICIAL NOTICE OF ARBITRATION PROCEDURAL REVIEW	92
SIOR CONFIDENTIALITY STATEMENT	94

PART IV: FORMS USED IN PROFESSIONAL STANDARDS PROCEEDINGS

A. Forms Overview

SIOR's Professional Standards system, like the National Association of REALTORS system from which it has been derived, uses a number of forms to document formal requests, commitments, and determinations reached in connection with ethics enforcement and financial dispute resolution. The forms are constructed to help describe their function in the process and often include instructions that provide related guidance. Nevertheless, it is essential for participants to understand, as applicable, the ethics enforcement procedures described in Part II of this manual or the financial dispute resolution procedures described in Part III. In the event of any conflict between the forms and the procedures specified in Parts II and III of this manual, the procedures are controlling.

The forms appearing in this part of the manual may be copied and used. Copies can be obtained by any interested person from SIOR's Washington, D.C. office, through the Professional Standards Staff Officer. Adherents also may download or print copies from SIOR's website (www.sior.com).

Questions concerning the forms, or suggestions for amendments or additions, may be directed to the Professional Standards Staff Officer. Policy issues arising in this context will be referred by the Professional Standards Staff Officer to the Professional Standards Committee and/or another appropriate SIOR body.

The forms are listed below in three separate lists. The first list identifies the forms intended for use in ethics enforcement proceedings. Some of those forms will be used in virtually every proceeding, while others may or may not be needed, depending on the course of the proceeding. The forms appearing in the second list are in the approximate order in which they might be used. The second list identifies the forms intended for use in resolving financial disputes. The third list consists of all forms organized by form number. The forms themselves, which make up the balance of this part of the manual, are ordered by number and are shown following the third list.

B. Forms Used in Ethics Enforcement Proceedings (in the order in which they are used)

Ethics Complaint (SIOR-PS Form 2)

Report of Potential Ethics Violation (SIOR-PS Form 2A)

Request for Information in Response to an Ethics Complaint (SIOR-PS Form 4)

Response to Ethics Complaint (SIOR-PS Form 6)

Notice of Right to Challenge Hearing Panel Members (SIOR-PS Form 1A)

Official Notice of Ethics Hearing (SIOR-PS Form 8)

Outline of Procedure for Ethics or Arbitration Hearing (SIOR-PS Form 10)

Certificate of Hearing Panel Qualification (SIOR-PS Form 1)

Ethics Hearing Panel Decision (SIOR-PS Form 11)

Request for Appeal of Ethics Hearing Decision (SIOR-PS Form 18)

Official Notice of Ethics Appeal Hearing (SIOR-PS Form 19)

Notice of Right to Challenge Hearing Panel Members in an Ethics or Arbitration Appeal Hearing (SIOR-PS Form 20)

Ethics Hearing Action by the Board of Directors (SIOR-PS Form 12)

C. Forms Used in Resolving Financial Disputes (in the order in which they are used)

Mediation/Arbitration Request (SIOR-PS Form 3)

Request for Information in Response to Mediation/Arbitration Request (SIOR-PS Form 5)

Response to Request for Mediation/Arbitration (SIOR-PS Form 7)

Mediation Officer Selection Form (SIOR-PS Form 17)

Mediation Officer Selection Notice (SIOR-PS Form 17A)

Mediation Resolution Agreement (SIOR-PS Form 16)

SIOR Ethics, Mediation & Arbitration Procedures Manual – revised 2021 Part IV: Forms – page 54

Official Notice of Arbitration Hearing (SIOR-PS Form 9)

Notice of Right to Challenge Hearing Panel Members (SIOR-PS Form 1A)

Outline of Procedure for Ethics or Arbitration Hearing (SIOR-PS Form 10)

Arbitration Guidelines (SIOR-PS Form 14)

Certificate of Hearing Panel Qualification (SIOR-PS Form 1)

Arbitration Decision and Award (SIOR-PS Form 13)

Request for Arbitration Hearing Procedural Review (SIOR-PS Form 21)

Official Notice of Arbitration Procedural Review (SIOR-PS Form 22)

Notice of Right to Challenge Hearing Panel Members in an Ethics or Arbitration Appeal Hearing (SIOR-PS Form 20)

D. All Forms Listed by Form Number

Certificate of Hearing Panel Qualification (SIOR-PS Form 1)

Notice of Right to Challenge Hearing Panel Members (SIOR-PS Form 1A)

Ethics Complaint (SIOR-PS Form 2)

Report of Potential Ethics Violation (SIOR-PS Form 2A)

Mediation/Arbitration Request (SIOR-PS Form 3)

Request for Information in Response to an Ethics Complaint (SIOR-PS Form 4)

Request for Information in Response to Mediation/Arbitration Request (SIOR-PS Form 5)

Response to Ethics Complaint (SIOR-PS Form 6)

Response to Request for Mediation/Arbitration (SIOR-PS Form 7)

Official Notice of Ethics Hearing (SIOR-PS Form 8)

Official Notice of Arbitration Hearing (SIOR-PS Form 9)

Outline of Procedure for Ethics or Arbitration Hearing (SIOR-PS Form 10)

Ethics Hearing Panel Decision (SIOR-PS Form 11)

Ethics Hearing Action by the Board of Directors (SIOR-PS Form 12)

Arbitration Decision and Award (SIOR-PS Form 13)

Arbitration Guidelines (SIOR-PS Form 14)

Mediation Resolution Agreement (SIOR-PS Form 16)

Mediation Officer Selection Form (SIOR-PS Form 17)

Mediation Officer Selection Notice (SIOR-PS Form 17A)

Request for Appeal of Ethics Hearing Decision (SIOR-PS Form 18)

Official Notice of Ethics Appeal Hearing (SIOR-PS Form 19)

Notice of Right to Challenge Hearing Panel Members in an Ethics or Arbitration Appeal Hearing (SIOR-PS Form 20)

Request for Arbitration Hearing Procedural Review (SIOR-PS Form 21)

Official Notice of Arbitration Procedural Review (SIOR-PS Form 22)

Confidentiality Statement (SIOR-PS Form 23)

Case #	SIOR-PS Form 1
--------	----------------

Society of Industrial and Office Realtors[®] Professional Standards Process 1201 New York Avenue, NW, Suite 350 Washington, DC 20005 USA 202.449.8200 phone | 202.216.9325 fax

CERTIFICATE OF HEARING PANEL QUALIFICATION

OF IND	indersigned member of a Hearing Panel of the Professional Standards Committee of the SOCIETY JSTRIAL AND OFFICE REALTORS $^{\textcircled{R}}$, do hereby certify that I am not disqualified by any reason stated from hearing the case cited below as:			
	vs			
Cited o	ase is a hearing in (check one only):			
	an ethics enforcement proceeding concerning alleged violation of the SOCIETY OF INDUSTRIAL AND			
_	OFFICE REALTORS [®] Code of Ethical Principles and Standards of Professional Practice. an arbitration of a financial dispute pursuant to the Code of Ethical Principles and Standards of Professional Practice.			
Reaso	ns for disqualification:			
	ember of the Hearing Panel will be disqualified from hearing any case for any of the following s if the Hearing Panel member:			
a.	is related by blood or other family relationship to either the Complainant or the Respondent.			
b.	 is an employer, partner, employee, or directly associated in business with, or affiliated with any person or entity who is directly associated in business with, either the Complainant or the Respondent. 			
C.	c. is a party to the hearing or a party or witness in any other pending case involving any party to this hearing.			
d.	 knows of any other reason that the Hearing Panel majority might determine would prevent him/her from rendering an impartial judgment. 			
Typed/	Printed Name of Panel Member Signature			
Date:				

Case #	SIOR-PS Form 1A
--------	-----------------

SOCIETY OF INDUSTRIAL AND OFFICE REALTORS® **Professional Standards Process** 1201 New York Avenue, NW, Suite 350 Washington, DC 20005 USA 202.449.8200 phone | 202.216.9325 fax

NOTICE OF RIGHT TO CHALLENGE HEARING PANEL MEMBERS IN ETHICS OR ARBITRATION PROCEEDING

Notice is given herewith to all parties in the matter of:			
vs			,
in an \square ethics / \square arbitration proceeding, that any party has the right to chindividual who may be appointed to serve on a Hearing Panel. A list of position, state of business residence and network affiliation, if any, of each pelow. A person will be automatically disqualified as a member of a Hea which the person is: (a) related by blood or other familial relationship Respondent; (b) an employer, partner, employee or directly associated in any person or entity directly associated in business with either the Complexity to the hearing or a party or a witness in any other pending case involor any party to this hearing; or (d) in any other similar relationship or has him/her from rendering an impartial decision. Other grounds for disqualific will be considered in light of their nature and the supporting evidence.	otential appointed botential appoint octential appoin ring Panel in art to either the continuous business with, ainant or the Roving Complainaknowledge wh	ees, sho ntee, is ny proc complai or affili espond ant, Res ich ma	owing the provided eeding in int or the iated with lent; (c) a spondent, y prevent
Please indicate whether you challenge each listed person by circling the a a copy of this form to the Professional Standards Staff Officer, includir reason(s) for challenge and the related supporting evidence. If your reasupport your challenge, the individual challenged will not be appointed to the Challenges must be filed with SIOR within ten (10) days from the date the parties involved in a case.	ng a statement ason(s) is deen he Hearing Pan	explained surel for y	ning your fficient to our case.
Potential Hearing Panel Members		Challer	nge?
		YES	NO
		YES	NO
	_	YES	NO
	_	YES	NO
	_	YES	NO
		YES	NO
	-	YES	NO

Case #	SIOR-PS Form 1A
Separately for each person challenged, state the grounds for challenged	llenge:
(The facts supporting the challenge should be detailed in a separany documents relied upon should be appended to that statemen	
Name (typed/printed):	
Signature:	
Challenge Date:	
OFFICE USE ONLY Date Received by SIOR:	

Case # _____ SIOR-PS Form 2

Society of Industrial and Office Realtors[®] Professional Standards Process 1201 New York Avenue, NW, Suite 350 Washington, DC 20005 USA 202.449.8200 phone | 202.216.9325 fax

ETHICS COMPLAINT

To the P	rofessional Standar	ds Committee	of The Society of Industrial and Office Realtors [®] :
Date Su	bmitted:		Date Received by SIOR:
Complai <i>Those m</i>	nant(s): naking the complaint		
Respond	dent(s): ject(s) of the compla	int.	
There has or Stand the attace may be	lard(s) of Practice _ ched statement which	ch is signed a	and dated by the listed Complainant(s). The attached statemer tation as the Complainant(s) believe(s) should be considered it
This con	nplaint is true and co	rrect to the b	est knowledge and belief of the undersigned.
	enalty of perjury, the ef, the allegations in		Complainant(s) declare that to the best of my (our) knowledge complaint are true.
;		ore the state	to this ethics complaint involved in civil or criminal litigation or in real estate licensing authority or any professional or governmen or agency?
	YES	NO	If yes, explain:
	Were the listed Res Practice at the time		bject to SIOR's Code of Ethics and Standards of Professional ethics violation?
	YES	NO	If no, explain:
Complai	nant(s)		
	Typed/printed:		Signature:
	Typed/printed:		Signature:

Case #	SIOR-PS Form 2A
<u></u>	0.01(1.01.011)

Professional Standards Process
1201 New York Avenue, NW, Suite 350
Washington, DC 20005 USA
202.449.8200 phone | 202.216.9325 fax

REPORT OF POTENTIAL ETHICS VIOLATION

Standard of Practice 15.4 obligates all Adherents to report to SIOR all potential violations of SIOR's Code of Ethical Principles and Standards of Professional Practice by other Adherents of which the reporting Adherent has firsthand knowledge. This form is used to submit such reports. This form need not be filed where an Adherent has filed an Ethics Complaint using SIOR-PS-Form 2 related to the same potential violation. Reports of potential ethics violations are subject to review and investigation by the Professional Standards Committee, which may initiate an enforcement proceeding or take other appropriate action.

To the Professional Standards Committee of the Society of Industrial and Office Realtors [®] :		
Date Submitted:		Date Received by SIOR:
Reporting Adherent(s) (name, address, phone):		
Adherent(s) Potentially at F	ault (name	, address, phone):
		nd knowledge of a potential violation of SIOR's Code Principle(s) _or Standard(s) of Practice
identify other persons who detailed explanations req	the reporte uested be	ached signed and dated statement. (The attached statement should be believes may have relevant knowledge. It should also include any slow. The attached statement may be accompanied by such e(s) should be considered in determining whether and what further
		his report involved in civil or criminal litigation or in any proceeding uthority or any government regulatory or administrative agency?
YES	NO	If yes, explain:
		ally at fault subject to SIOR's Code of Ethics and Standards of e potential ethics violation(s)?
YES	NO	If no, explain:
This report is true and corre	ect to the be	est knowledge and belief of the undersigned.
Reporting Adherent(s)		
Name (Typed/printed):		Signature:

Case #	SIOR-PS Form 3

SOCIETY OF INDUSTRIAL AND OFFICE REALTORS[®]
Professional Standards Process
1201 New York Avenue, NW, Suite 350
Washington, DC 20005 USA
202.449.8200 phone | 202.216.9325 fax

MEDIATION/ARBITRATION REQUEST

A real estate-related business dispute exists between the Complainant(s) and the Respondent(s) listed below. Complainant(s) understand that principal Respondent(s) named below was(were) an Active member or held Member Associate status in the Society of Industrial and Office REALTORS® ("SIOR") at the time the dispute arose.

Principal Respondent:

Other Respondents:

There is due, unpaid and owing to Complainant(s) from the above named Respondent(s) the sum of

This claim is predicated upon the attached statement, and supporting

documentation, which is incorporated by reference into this request.

Complainant(s) request and consent to mediation and, if necessary, arbitration through SIOR in accordance with Principle 16 of the SIOR's Code of Ethical Principles and Standards of Professional Practice, and agree to mediate in good faith, and to abide by and promptly comply with any mediation settlement or arbitration award. In the event the prevailing party(ies) does(do) not comply with the mediated settlement or arbitration award and it is necessary for any party to this mediation/arbitration to obtain judicial confirmation and enforcement of an award against the losing party(ies), The losing party(ies) agree to pay the prevailing party the costs and reasonable attorney fees incurred in obtaining such confirmation and enforcement.

Enclosed is Complainant(s) check, payable to Society of Industrial and Office REALTORS[®], in the amount of \$3,500 for the mediation filing fee to help cover mediation expenses.

Complainant(s) agree(s) to comply with the procedures for resolving financial disputes provided in the Society's *Ethics, Mediation & Arbitration Procedures Manual*, including the use of mediation prior to entering into a formal arbitration. Mediation is the first step in SIOR's arbitration process. If mediation is not successful, the matter will be referred for an arbitration hearing in accordance with the guidelines set forth in SIOR's Ethics, Mediation and Arbitration Procedures Manual.

If the parties to the mediation conference agree to a settlement of the dispute and the settlement is reduced to writing and is fully signed by all parties, the matter will be considered resolved and will not be the subject of a subsequent arbitration hearing. In the event that either of the parties fails to abide by the terms of the settlement, the matter may not be arbitrated; instead, the other party is encouraged to have the settlement agreement judicially enforced by a court or competent jurisdiction.

No aspect of a mediation conference may be relied upon or introduced as evidence in any arbitration, judicial, or other proceeding, including, but not limited to, views expressed or suggestions made by the parties with respect to possible settlement of the dispute, admissions made in the course of the mediation, proposals made or views expressed by the mediation officer, or the responses of any party to any of the aforementioned. No privilege will be affected by disclosures made in the course of mediation. Disclosure of any records, reports, or other documents received or prepared by the mediation officer will not be compelled. Neither SIOR nor the mediation officer may be compelled to disclose or testify in any proceeding as to information disclosed or representations made in the course of the mediation or confidential communication to the mediation officer.

Case # SIOR-PS Form 3
Complainant(s) agree(s) that if mediation fails to resolve the dispute, SIOR arbitration procedures will be invoked and Complainant(s) will remit an additional \$ \$4,000 arbitration filing fee to cover arbitration expenses.
Under penalty of perjury, Complainant(s) declare that this request and the allegations contained herein are true and correct to the best of their knowledge and belief.
Are the circumstances giving rise to this mediation/arbitration request the subject of pending or completed civil litigation or administrative agency action? YES NO
Is the principal Complainant the broker of record for your firm in this transaction? YES NO NOT APPLICABLE
If the broker of record for your firm in this transaction is other than the principal Complainant, it will be necessary for that broker of record to agree that this matter may be resolved through the requested mediation/arbitration, sign this form on the "Other Complainants" line and indicate that she/he is the broker of record.
My signature to this Agreement confirms my acknowledgment of my rights and the terms of the mediation/arbitration procedures outlined herein. I hereby affirm that I have authority to enter into and sign a binding written agreement to settle this dispute and that any agreement signed by the parties pursuant to the mediation conference will be binding.
Complainant(s):
Principal Complainant:
Other Complainants:
Date:

SIOR-PS Form 4

Society of Industrial and Office Realtors[®] Professional Standards Process 1201 New York Avenue, NW, Suite 350 Washington, DC 20005 USA 202.449.8200 phone | 202.216.9325 fax

REQUEST FOR INFORMATION IN RESPONSE TO AN ETHICS COMPLAINT

Date:	_
То:	_
In the case of	(Complainant)
vs.	(Respondent).
The Complainant has named you as Respondent in the an SIOR Grievance Review Panel in accordance with Ethics, Mediation & Arbitration Procedures Manual.	
Please be advised that you have thirty (30) days after t Professional Standards Staff Officer at the address al Response to Ethics Complaint (SIOR-PS Form 6) a response will be provided to the Complainant.	bove. Your response should include the attached
Failure to provide a response in the designated time profer a hearing and may also result in the Professional syou, alleging a violation of the Code of Ethics for failurgrievance proceeding.	Standards Committee filing of a complaint against
YOUR RESPONSE IS DUE BY Please attach your response to SIOR-PS Form	n 6 and return by the indicated date.
Please contact me at 202.449.8204 if you have question	ons. Thank you for your cooperation.
	Sincerely,
	Professional Standards Staff Officer

Case #	SIOR-PS Form 5
--------	----------------

Society of Industrial and Office Realtors[®] Professional Standards Process 1201 New York Avenue, NW, Suite 350 Washington, DC 20005 USA 202.449.8200 phone | 202.216.9325 fax

REQUEST FOR INFORMATION IN RESPONSE TO MEDIATION/ARBITRATION REQUEST

Date:	
То:	, named by
	as Respondent.
filed with SIOR. The matter will be reviewe	/Arbitration Request which names you as the Respondent, as ed by an SIOR Grievance Review Panel to determine whether it ation hearing, in accordance with the policies and procedures iteration Procedures Manual.
Mediation/Arbitration Request within thirty submitted using SIOR-PS Form 7, "Responsithin the time allotted, the Grievance Re	equests that the Respondent provide a written response to this (30) days of receipt of this request. Your response should be use to Request for Mediation/Arbitration." If no response is filed eview Panel will make its determination whether an arbitration the information set forth in the Arbitration Request.
YOUR RESPONSE IS DUE BY	OR-PS Form 7 and return by the indicated date.
If you have questions, please contact me a	t 202-449-8204. Thank you for your cooperation.
	Sincerely,
	Professional Standards Staff Officer

Case #	SIOR-PS Form 6

Society of Industrial and Office Realtors®
Professional Standards Process
1201 New York Avenue, NW, Suite 350
Washington, DC 20005 USA
202.449.8200 phone | 202.216.9325 fax

RESPONSE TO ETHICS COMPLAINT

To the SOCIETY OF INDUSTRIAL AND OFFICE REALTORS®:
Complainant(s):
Respondent(s):
In response to an alleged violation of SIOR's <i>Code of Ethical Principles and Standards of Professional Practice</i> , the undersigned Respondent(s) responds and substantiates such response by the attached statementThe statement should address whether the Respondent(s) agrees or disagrees with the facts asserted by the Complainant(s) and should address whether the facts are understood by the Respondent(s) do or do not involve one or more ethics violations. The statement may be accompanied by such documents as the Respondent(s) may believe will prove helpful.
Under penalty of perjury, I declare that, to the best of my knowledge and belief, the statements contained in this response are true and correct.
Respondent(s):
Printed Name(s):
Signature(s):
Date Signed:
OFFICE USE: Date Received by SIOR:

Case #	SIOR-PS Form 7

Society of Industrial and Office Realtors®
Professional Standards Process
1201 New York Avenue, NW, Suite 350
Washington, DC 20005 USA
202.449.8200 phone | 202.216.9325 fax

RESPONSE AND AGREEMENT TO MEDIATE OR ARBITRATE

In response to a M	ediation/Arbitration Request naming
'	as Complainant(s)
and	as Respondent(s),
Ethical Principles	existence of a dispute arising out of the real estate business as defined in the Code of and Standards of Professional Practice but deny any indebtedness as claimed. My ed upon the attached statement, which is hereby incorporated by reference into this
exists be	le: Affirmatively, I claim that a monetary dispute arising out of the real estate business ween the parties wherein there is due, owing and unpaid to me the sum of My claim is predicated upon the attached statement, which is hereby ed into this response.
	est step in SIOR's arbitration process. If mediation is not successful, the matter will be

Mediation is the first step in SIOR's arbitration process. If mediation is not successful, the matter will be referred for an arbitration hearing in accordance with the guidelines set forth in SIOR's Ethics, Mediation and Arbitration Procedures Manual.

If the parties to the mediation conference agree to a settlement of the dispute and the settlement is reduced to writing and has been signed by all of the parties, the matter will be considered resolved, and will not be the subject of a subsequent arbitration hearing. In the event that either of the parties fails to abide by the terms of the settlement, the matter may not be arbitrated; instead, the other party is encouraged to have the settlement agreement judicially enforced by a court of competent jurisdiction.

No aspect of a mediation conference may be relied upon or introduced as evidence in any arbitration, judicial, or other proceeding, including, but not limited to, views expressed or suggestions made by the parties with respect to possible settlement of the dispute, admissions made in the course of the mediation, proposals made or views expressed by the mediation officer, or the responses of any party to any of the aforementioned. No privilege will be affected by disclosures made in the course of mediation. Disclosure of any records, reports, or other documents received or prepared by the mediation officer will not be compelled. Neither SIOR nor the mediation officer may be compelled to disclose or testify in any proceeding as to information disclosed or representations made in the course of the mediation or confidential communication to the mediation officer.

I consent to mediation, and to arbitration in the event that mediation fails, through SIOR in accordance with SIOR's *Ethics, Mediation & Arbitration Procedures Manual*; in the event of arbitration, I agree to abide by the arbitration award and to comply with it promptly.

In the event that I do not comply with an arbitration award and it is necessary for any party to this arbitration to obtain judicial confirmation and enforcement of the arbitration award against me, I agree to pay the party obtaining such enforcement the costs and reasonable attorneys' fees incurred in obtaining such confirmation and enforcement.

Case #	SIOR-PS Form 7 continue

Is (one of) the named Respondent(s) the broker of record for your firm in this transaction?

YES NO NOT APPLICABLE

If there is a broker of record for your firm in this transaction other than the named Respondent(s), it will be necessary for that broker of record to agree that this matter may be resolved through the requested arbitration, sign this form on the "Other Respondents" line and indicate that she/he is the broker of record.

I enclose my check, payable to the SOCIETY OF INDUSTRIAL AND OFFICE REALTORS $^{\circledR}$, in the amount of \$3,500 for the mediation filing fee. I agree that if mediation fails to resolve the dispute, SIOR arbitration procedures will be complied with and I will remit an additional \$4,000 arbitration filing fee to cover arbitrator expenses.

Under penalty of perjury, I declare that the information provided with this response is true and correct to the best of my knowledge and belief.

My signature to this Agreement confirms my acknowledgment of my rights and the terms of the mediation/arbitration procedures outlined herein. I hereby affirm that I have authority to enter into and sign a binding written agreement to settle this dispute and that any agreement signed by the parties pursuant to the mediation conference will be binding.

Date:		_	
Respondent(s):			
Principal Res	pondent Name:		
Principal Res	pondent Signature:		
Other Respon	ndents:		
OFFICE USE:	Date Received by SIOR:		

SIOR-PS Form 8
S

Society of Industrial and Office Realtors[®]
Professional Standards Process
1201 New York Avenue, NW, Suite 350
Washington, DC 20005 USA
202.449.8200 phone | 202.216.9325 fax

OFFICIAL NOTICE OF ETHICS HEARING

In the case of	
	vs.
Complainant(s)	vs Respondent(s)
following as the date, time, and place for	at the Chair of the Ethics Hearing Panel has designated the r hearing the above case. This is a hearing concerning the ons of SIOR's Code of Ethical Principles and Standards of
You are hereby notified to appear at the app	pointed time and place.
Date:	Time:
Place:	
The Chair and members of the Ethics Heari Chair:	
counsel, and each party is requested to prohave representation present, at least fiftee will not invalidate a party's right to represent Each party must provide a list of names of and SIOR not less than fifteen (15) days witnesses to be present at the time and pla witnesses are unable to be present for the the Ethics Hearing Panel for such witnesses	e SIOR member who is an Adherent to the Code or by legal devide written notice to the parties and SIOR of their intention to the price of the parties and SIOR of their intention to the provide this notice of the hearing. Failure to provide this notice of the hearing, but may result in a continuance of the hearing. If witnesses he/she intends to call at the hearing to all parties as prior to the hearing. Each party must arrange for his/her are designated for the hearing; or in the event that one or more the designated hearing, the party must request permission from the second to be allowed to testify remotely. The parties must not be Ethics Hearing Panel or the Board of Directors prior to the mouncement of the decision.
Date Issued:	Professional Standards Staff Officer

Case #	SIOR-PS Form 9
--------	----------------

Society of Industrial and Office Realtors[®]
Professional Standards Process
1201 New York Avenue, NW, Suite 350
Washington, DC 20005 USA
202.449.8200 phone | 202.216.9325 fax

OFFICIAL NOTICE OF ARBITRATION HEARING

In the case of				
VS.				
Complainant(s)	vs Respondent(s)			
following as the date, time, and pla	that the Chair of the Arbitration Hearing Panel has designated the ace for hearing the above case. This is an arbitration hearing spute, and the hearing will be conducted in accordance with SIOR's edures Manual.			
You are hereby notified to appear at th	e appointed time and place.			
Date:	Time:			
Place:				
The members of the Hearing Panel for	this case are:			
Chair:				
the parties and SIOR of an intention prior to the hearing. Failure to provide	gal counsel and each party is requested to provide written notice to to have legal counsel present, at least fifteen (15) calendar days this notice will not invalidate a party's right to legal representation hearing if the Arbitration Hearing Panel determines that the rights on by legal counsel.			
and SIOR not less than fifteen (15) his/her witnesses to be present at the or more witnesses are unable to be pr from the Arbitration Hearing Panel for not discuss the case with any membe the hearing or after the hearing and proceeding may settle the issue between	nes of witnesses he/she intends to call at the hearing to all parties calendar days prior to the hearing. Each party must arrange for time and place designated for the hearing; or in the event that one esent for the designated hearing, the party must request permission such witness(es) to be allowed to testify remotely. The parties must er of the Arbitration Hearing Panel or the Board of Directors prior to a prior to announcement of the decision. Parties to an arbitration een them by joint agreement at any time and, in such event, should ch settlement to the Professional Standards Staff Officer.			
Date Issued:	Due for a claused Ottom decode Ottoff Office of			
	Professional Standards Staff Officer			

Case # _____ SIOR-PS Form 10

SOCIETY OF INDUSTRIAL AND OFFICE REALTORS®
Professional Standards Process
1201 New York Avenue, NW, Suite 350
Washington, DC 20005 USA
202.449.8200 phone | 202.216.9325 fax

OUTLINE OF PROCEDURE FOR ETHICS OR ARBITRATION HEARINGS

Postponement of Hearing

Postponement may be granted if there are extenuating circumstances. Requests for postponement must be made in writing. Requests can be granted by the Hearing Panel Chair. All parties will be advised of the date of the rescheduled hearing.

Recording the Hearing

SIOR may, and any party may at its own expense, have a court reporter present or may make an audio recording of the proceeding. If a party prepares a hearing transcript, it must provide a copy to SIOR's Professional Standards Staff Officer. If SIOR utilizes a court reporter in lieu of an audio recording, the parties may still make their own audio recordings. The hearing may not be videotaped without the prior written consent of all parties and SIOR.

Hearing Arrangements and Procedure

Absent specific prior arrangements, hearings typically are held in rooms such as hotel meeting rooms and are scheduled for less than half of a day. If any party requires unusual physical arrangements or an unusually long period of time so that the party's case may be fully and fairly presented, the party should submit in advance a request for such accommodations to the Professional Standards Staff Officer in writing as soon as possible, and no less than fifteen (15) calendar days in advance of the hearing, so that there will be a reasonable opportunity to assess and, if appropriate, accommodate the request. Decisions on requests for unusual hearing physical arrangements or times will be ruled upon by the Hearing Panel Chair, in consultation with the Professional Standards Staff Officer.

The Hearing Panel will <u>not</u> be bound by the rules of evidence applicable in courts of law, but will afford all parties a full opportunity to be heard, present witnesses, and offer evidence, subject to the Hearing Panel Chair's judgment as to relevance. After the Official Notice of Ethics Hearing (SIOR-PS Form 8) has been issued, the parties may provide the Professional Standards Staff Officer with additional written materials and/or statements to support their positions. Written affidavits should include notarized proof of their authorship. At the hearing, panel members may also receive other written statements, documents, or other papers and will hear testimony from the parties and their witnesses. It is the responsibility of all parties to arrange for their witnesses to be present at hearing. In appropriate circumstances and subject to prior notice to the other parties and the Professional Standards Staff Officer, non-Adherent witnesses may participate by telephone or other electronic means. The Hearing Panel may receive and consider any evidence it deems material and proper, including evidence of experts. Each party is responsible for the expenses of their witnesses and legal counsel.

Hearing Format

The hearing procedure will include the following steps:

- An Opening Statement will be made by the Hearing Panel Chair, citing authority to hear the case and explaining the reasons for the hearing.
- Reading of the complaint into the record.
- The testimony of all parties and witnesses will be affirmed. All witnesses will be excused from the hearing except while testifying.
- The parties will be given an opportunity to present evidence and testimony on their behalf and they may call witnesses. The parties will be afforded an opportunity to examine and crossexamine all witnesses and parties.
- The panel members may ask questions at any time during the hearing.
- The Hearing Panel Chair may exclude any question ruled to be irrelevant or argumentative.
- Each party may make a closing statement. The Complainant will make the first closing statement and the Respondent will make the final closing statement.
- The hearing will be adjourned.
- The Hearing Panel will go into executive session to decide the case.

Finding in an Ethics Hearing

The finding and recommendation for discipline, if any, will be reduced to writing by the Hearing Panel and submitted to the parties and SIOR's Board of Directors in accordance with the procedures defined in SIOR's Ethics, Mediation and Arbitration Procedures Manual.

Disciplinary action that SIOR's Board of Directors may impose on Adherents for violation of their above duties may consist of one or more of the following:

- 1. a letter of warning with a copy placed in the Adherent's file;
- 2. a letter of reprimand with a copy placed in the Adherent's file;
- 3. a requirement that the Adherent attend a SIOR course or seminar or a program sponsored by another Realtor group that addresses ethics and can be reasonably attended taking into consideration cost, location, and duration;
- 4. an appropriate and reasonable fine not to exceed a maximum amount equal to ten times SIOR's annual dues in effect at the time the fine is imposed;
- 5. placing the Adherent on probation for a stated period of time, not less than thirty (30) days nor more than one (1) year in duration;

PROBATION means the discipline recommended will be held in abeyance for a stipulated period of time, which may not exceed one (1) year. Any subsequent finding of a violation of SIOR's Bylaws or Code of Ethical Principles during the probationary period may, at the discretion of SIOR's Board of Directors, result in the imposition of the suspended discipline. Absent any subsequent findings of a violation during the probationary period, both the probationary status and the suspended discipline shall be considered fulfilled, and the record shall reflect the fulfillment;

- 6. suspension of membership or Member Associate status for a period not less than thirty (30) days nor more than one (1) year, with automatic reinstatement to good standing at the end of the suspension period;
 - SUSPENSION of membership means temporary termination of membership rights and privileges and denial of SIOR services and benefits for a period not less than thirty (30) days nor more than one (1) year, on terms and conditions expressly stated for an established period of time, including, but not limited to, use of the SIOR designation. At the expiration of the suspension period, all withdrawn membership rights and privileges will be automatically restored. During the suspension period, membership per se continues and the suspended members remain obligated for payment of membership dues and to abide by SIOR's Bylaws and Code of Ethical Principles. Any failure to abide by the terms and conditions of the suspension, or any subsequent finding of a violation of SIOR's Bylaws or Code of Ethical Principles, shall be grounds for consideration as to possible extension of the suspension or expulsion from membership in SIOR by SIOR's Board of Directors. In lieu of suspension, at the option of the Board, an assessment not to exceed an amount equal to ten times the SIOR's annual dues in effect at the time the assessment is imposed, which assessment may be imposed on a given Adherent only once in any three (3) year period:
- 7. expulsion from membership or Member Associate status with no reinstatement opportunity for a specified period of one (1) to three (5) years, with any subsequent readmission to be by application only after the specified period of expulsion, with the decision on readmission to be made based on the merits of the application, and to be considered under the admissions standards in effect at the time.
- 8. Denial of membership and application for membership for up to (5) years from date of findings of the Hearing Paned

SIOR has adopted criteria limiting imposition of the more serious sanctions, suspension and expulsion. An Adherent may be suspended or expelled where the Hearing Panel finds:

- b. A violation that caused death or significant personal injury; or
- f. A violation that has caused substantial economic injury to the party represented or to any other party to whom a duty of honesty, faith, trust, or fiduciary is owed; or
- g. A violation that has produced governmental criticism or substantial media criticism of the Adherent and/or has cast aspersions on the real estate profession or SIOR; or
- h. That the Respondent's misconduct involved a violation of one or more laws, statutes, or ordinances, of if the misconduct is considered to be as egregious as those addressed by the above three criteria but not literally covered by any one of them; or
- i. That within the 10 years preceding the complaint, the Respondent has been found to have violated the Code on two (2) or more separate occasions.

The first two criteria are intended to identify violations that have caused serious harm to individuals or businesses. The third criteria encompasses certain violations that violate laws or have adversely affected the reputation of the profession or SIOR. The fourth criteria covers violations of equal or greater severity that do not fall literally within the language of any of the first three criteria.

Award in an Arbitration Hearing

The decision of the Hearing Panel in an arbitration proceeding will be reduced to writing, setting forth only the amount of the award, if any, and a copy will be furnished to each of the parties to the arbitration. A copy will be filed with the Professional Standards Staff Officer.

Use of Legal Counsel

A party may be represented in any arbitration hearing by legal counsel. However, parties may not refuse to respond directly to requests for information or questions addressed to them by Hearing Panel members except on the grounds of self-incrimination, or on other grounds deemed appropriate by the Hearing Panel Chair. In this connection, the panel need not accept the statements of legal counsel as being the statements of legal counsel's client if the panel desires direct testimony. Parties will be held responsible for the conduct of their legal counsel. Any efforts by legal counsel to harass, intimidate, coerce, or confuse the panel members or any party to the proceedings or any action by legal counsel which is viewed by the panel as disruptive of the proceedings will be grounds for exclusion of legal counsel from the hearing. The decision to exclude legal counsel for any of the foregoing reasons will be the result of a majority vote of the Hearing Panel members and will not be appealable. In the event legal counsel is excluded, the hearing will be postponed to a date certain not less than thirty (30) days nor more than sixty (60) days from the date of adjournment to enable the party to obtain alternate legal counsel, provided, however, that such postponement will not be authorized if it appears to members of the panel that the action of legal counsel has been undertaken to obtain a postponement or delay of the hearing.

Accommodation of Disabilities

If any party, witness, or counsel involved in the proceeding requires accommodation because of a disability, SIOR will do its best to provide such accommodation. Individuals requiring such accommodation must contact SIOR's Professional Standards Staff Officer at least fifteen (15) days before the hearing.

Questions about These Procedures

Questions about these procedures should be directed to SIOR's Professional Standards Staff Officer at the address and telephone number above. A more complete statement of the rules governing hearings and other proceedings, as well as a full set of potentially applicable forms, is found in SIOR's Ethics, Mediation and Arbitration Procedures Manual, a copy of which may be obtained from the Professional Standards Staff Officer.

Case #	SIOR-PS Form 11

Society of Industrial and Office Realtors[®] Professional Standards Process 1201 New York Avenue, NW, Suite 350 Washington, DC 20005 USA 202.449.8200 phone | 202.216.9325 fax

ETHICS HEARING PANEL DECISION

Date File	ed:				Hearing [Date: _						
Complai	nant(s): _											
Respond	lent(s):											
FINDING	S OF FA	СТ										
The Hea	ring Pane	l's dec	cision is b	ased on it	s having found	the fo	llowing	facts	to have	been est	tablished:	
CONCLU	JSIONS											
					he case describ f the Code of E						iolation o	r
Princ	ciple/Stan	dard o	f Practice))	[]	OIV V	LATION LATION LATION		N	OT IN VIC OT IN VIC OT IN VIC	DLATION	
The that:				above	conclusions,	in	light	of	the	above	facts,	is
	MENDAT	ION F	OR DISC	IPLINAR'	Y ACTION						-	
					ne following acti	on:						

The preceding Findings of Fact, Conclusions, and Recommendation(s) were rendered by an Ethics Hearing Panel comprised of the following members whose signatures are affixed below.

signature

NOTICE: This decision is not final and is subject to certain rights of both the Complainant(s) and the Respondent(s).

REHEARING AND APPEAL RIGHTS: Within twenty (20) days of this notification, either party may request a rehearing on an adverse decision by the original Hearing Panel. This request must be in writing and directed to the Hearing Panel Chair via the staff officer and the Hearing Panel will consider the request. The granting of a request for rehearing will be based only upon the requestor's obtaining new evidence which could not have been discovered with reasonable diligence and produced at the original hearing. If no petition for rehearing is filed, either party may appeal to the Board of Directors within thirty (30) days after a Hearing Panel decision has been transmitted to the parties. If rehearing has been sought, appeals must be requested within ten (10) days after denial of a petition for rehearing.

FINAL ACTION BY BOARD: Both the Complainant(s) and the Respondent(s) will be notified upon final action of the Board of Directors on SIOR-PS Form 12.

typed/printed

Case # _____ SIOR-PS Form 12

Society of Industrial and Office Realtors[®] Professional Standards Process 1201 New York Avenue, NW, Suite 350 Washington, DC 20005 USA 202.449.8200 phone | 202.216.9325 fax

ETHICS HEARING ACTION BY THE BOARD OF DIRECTORS

Ethics Hearing Date:
Complainant(s):
Respondent(s):
IF NO APPEAL HAS BEEN FILED AS TO THE HEARING PANEL'S DECISION:
The decision of the Hearing Panel in the above referenced case (attached) came before SIOR's Board of Directors (or its Governance Committee) and was considered
The decision of the Hearing Panel and its recommendation for disciplinary action, if any, in the above referenced case was:
adopted verbatim.
adopted, but the recommendation for discipline is modified as follows:
remanded to the Hearing Panel for further consideration of the recommended discipline.
remanded to the Professional Standards Committee for a new hearing by a differe Hearing Panel based on perceived procedural deficiency(ies) or errors.
Reason(s) for rejection/alteration of the Hearing Panel's decision and/or recommendation and Boa concerns with the Hearing Panel's decision and/or recommendation:
This action of the Board of Directors (or the Governance Committee) was adopted o
SIOR President

Case #	_ SIOR-PS Form 12 continued
<u>IF AN APPEAL</u>	HEARING IS HELD:
The decision of	the Hearing Panel in the above referenced case was appealed by:
	(Appellant)
and was consi	dered by SIOR's Board of Directors, the Governance Committee, or an appeal panel on
A copy of the A	ppellant's appeal request stating the basis for appeal is attached.
Basis f	or Appeal:
The decision of referenced cas	of the Hearing Panel and its recommendation for disciplinary action, if any, in the above e was:
	adopted verbatim.
	adopted, but the recommendation for discipline is modified as follows:
	remanded to the Hearing Panel for further consideration of the recommended discipline.
	remanded to the Professional Standards Committee Chair for a new hearing by a different Hearing Panel based on perceived procedural deficiency(ies) or error(s).
	rejected and the matter is dismissed.
	remand/alteration/rejection of the Hearing Panel's decision and/or disciplinary on and Board concerns with the Hearing Panel's decision and/or recommendation:
DISPOSITION	OF APPEAL DEPOSIT (if any):
	returned to Appellantother; specify:
	retained by SIOR.
This action of the	ne Board of Directors was adopted on

Appeal Hearing Panel Chair_____

SIOR President _____

Case # _____ SIOR-PS Form 13

Society of Industrial and Office Realtors® Professional Standards Process 1201 New York Avenue, NW, Suite 350 Washington, DC 20005 USA 202.449.8200 phone | 202.216.9325 fax

ARBITRATION DECISION AND AWARD

Complainant(s):	·
Respondent(s):	
Arbitration Hearing Date:	
equitable resolution of a business dispute in a Standards of Practice, do hereby certify that controversy in accordance with the applicable	el members, duly appointed to hear and determine an accordance with SIOR's Code of Ethical Principles and we proceeded to hear evidence of the parties to this provisions of SIOR's Ethics, Mediation and Arbitration idence, proof, and arguments of the respective parties to embers has reached the following decision:
There is an amount due in the su	um of \$to the
The fees are subject to the follow	ving disposition:
There is no amount due to either	party.
arbitration, except as follows: retained fee not specifically use	will be retained by SIOR to offset the expenses of the Any portion of any d to offset arbitration expenses will be retained by SIOR and procedures set forth in the Ethics, Mediation and
twenty (20) days after this decision has been ser	hearing must be filed in writing to SIOR President within rved on the parties. Requests for procedural review must s or errors the appealing party believes prevented a fair
Date of Decision:	
Hearing Panel Chair:	
typed/printed	signature

Case #	SIOR-PS Form 13 continued
Hearing Panel Members:	
typed/printed	signature
typed/printed	signature

signature

typed/printed

Case #	SIOR-PS Form 14

Professional Standards Process
1201 New York Avenue, NW, Suite 350
Washington, DC 20005 USA
202.449.8200 phone | 202.216.9325 fax

ARBITRATION GUIDELINES

Factors for Consideration by Arbitration Hearing Panels

The following factors are recommended for consideration by Hearing Panels convened to arbitrate disputes between brokers or between brokers and their clients, customers and/or other real estate professionals. This list is not all-inclusive nor is it applicable to every situation. It is intended to guide panels as to the facts, issues, and relevant questions that may aid them in reaching fair, equitable, and reasoned decisions and to guide parties to an arbitration in preparing their testimony and evidence.

Factor 1: No Predetermined Rule of Entitlement

Every arbitration hearing is considered in light of the relevant facts and circumstances as presented by the parties and their witnesses. Every situation is unique. Previous decisions made by other panels in other matters are to be disregarded.

"Procuring cause" is the primary determining factor in entitlement to compensation in many commission-related disputes. Agency relationships, in and of themselves, do not determine entitlement to compensation. An agency relationship with a client and entitlement to compensation are separate issues. A relationship with a client, or lack of one, should be considered in accordance with the guidelines established to assist panel members in determining procuring cause.

Factor 2: Arbitrability and Appropriate Parties

While primarily the responsibility of the Grievance review panel, arbitration Hearing Panels may consider questions of whether, in light of the evidence (as distinct from allegations) an arbitrable issue actually exists and whether the parties named are appropriate for arbitration. For additional information on arbitrable issues, refer to Principle 16 of SIOR's Code of Ethics and the associated Standards of Practice.

Factor 3: Relevance and Admissibility

Hearing panels often are asked to rule on questions of admissibility and relevancy. The general rule is that anything a Hearing Panel believes may assist it in reaching a fair, equitable, and knowledgeable decision is admissible.

Arbitration Hearing Panels are intended to resolve contractual questions, not to determine whether the Code of Ethics has been violated (which is the purpose of an Ethics Hearing Panel). An otherwise substantiated award cannot be withheld solely on the basis that the Arbitration Hearing Panel looks with disfavor on the potential recipient's manner of doing business or even that the panel believes that unethical conduct may have occurred.

At the same time, evidence or testimony is not inadmissible simply because it relates to potentially unethical conduct. While an award (or failure to make a deserved award) cannot be used to punish a perceived "wrongdoer," it is equally true that Hearing Panels are entitled to (and fairness requires that they) consider all relevant evidence and testimony so that they will have a clear understanding of what transpired before determining entitlement to any award.

Factor 4: Communication and Contact — Abandonment and Estrangement

Many arbitrable disputes turn on the relationship (or lack thereof) between a broker (often a cooperating broker) and a prospective purchaser or tenant. Hearing panels will consider whether, under the circumstances and in accord with local custom and practice, brokers made reasonable efforts to develop and maintain ongoing relationships with purchasers/tenants. Panels will want to determine, in cases where two cooperating brokers have competing claims against a listing broker, whether the first cooperating broker actively maintained ongoing contact with the purchaser or tenant or, alternatively, whether the broker's inactivity, or perceived inactivity, may have caused the purchaser/tenant to reasonably conclude that the broker had lost interest or disengaged from the transaction (abandonment).

In other instances, a purchaser/tenant, despite reasonable efforts by the first broker to maintain contact, may seek assistance from another broker. The panel will consider why the purchaser was estranged from the first broker. In still other instances, there may be no question that there was an ongoing relationship between the first broker and the buyer/tenant; the issue then becomes whether the first broker engaged in conduct which caused the purchaser/tenant to terminate the relationship (estrangement). This can be caused, among other things, by words or actions. The panel will consider whether such conduct caused a break in the series of events leading to the transaction and whether the successful transaction was actually brought about through the initiation of a separate, subsequent series of events by the second cooperating broker.

In assessing issues in this area, panels should be mindful of the rules imposed on Adherents by Code Principle 14 ("Do Not Interfere With the Relationship or Rights of Other Real Estate Professionals"), Standard of Practice 14.1 ("Abandonment"), Standard of Practice 14.2 (Alienation and Estrangement"), Standard of Practice 14.3 ("Termination"), Standard of Practice 14.4 ("No Other Exceptions"), and Standard of Practice 14.5 ("Compensation Rights").

Factor 5: Conformity with the Law

The procedures by which arbitration requests are received, hearings are conducted, and awards are made must be in conformity with governing law. The parties have the obligation to explain their perception of the requirements of applicable law. In such matters, the advice of SIOR's legal counsel may be sought in any case where the panel believes that the procedural requirements of law may be critical or where the parties insist that the procedures being followed are not adequate.

Factor 6: Consideration of the Entire Course of Events

The standard of proof in SIOR-conducted arbitration is a preponderance of the evidence and the burden of proof rests with the party initially requesting arbitration. However, this does not preclude arbitration Hearing Panelists from asking questions of the parties or witnesses to confirm their understanding of testimony presented or to ensure that panel members have a clear understanding of the events that led to the transaction and to the request for arbitration.

General Issues and Questions to Be Considered by Arbitration Hearing Panels

Since each transaction is unique, it is impossible to develop a comprehensive list of all issues or questions that panel members may consider in a particular hearing. Panel members are advised to consider the following, which are representative of the issues and questions frequently involved in arbitration hearings.

Nature and Status of the Transaction: What was the nature of the transaction? Was there an industrial or office sale or lease?

Nature, Status, and Terms of the Listing Agreement: What was the nature of the listing or other agreement? Exclusive right to sell, exclusive agency, open, or some other form of agreement? Was the listing agreement in writing? If not, is the listing agreement enforceable? Was the listing agreement in effect at the time the sale/lease was executed? Was the listed property subject to a management agreement? Were the broker's actions in accordance with the terms and conditions of the listing agreement? Were all conditions of the agreement met? Did the final terms of the transaction meet those

specified in the listing agreement? Did the transaction close? Did the listing broker receive a commission? If not, why?

Nature, Status, and Terms of the Offer to Compensate: Were an offer of cooperation and an offer of compensation made in writing? If not, how were they communicated? Is the claimant a party to whom the listing broker's offer of compensation was extended? Were the broker's actions in accordance with the terms and conditions of the offers of cooperation and compensation (if any)? Were all conditions of the agreement met?

Roles and Relationships of the Parties: Who was the listing broker? Who was the cooperating broker or brokers? Were any of the parties acting as subagents? As buyer/tenant representatives? In some other capacity? Did any of the cooperating brokers have an agreement, written or otherwise, to act as an agent or in some other capacity on behalf of any of the parties? Were any of the brokers (including the listing broker) acting as a principal in the transaction? What were the brokers' relationships with respect to the seller/landlord, buyer/tenant, listing broker, and any other cooperating brokers involved in the transaction? Was the party to whom the property was sold or leased represented by a party with whom the broker has previously dealt? Is the primary shareholder of the buyer/tenant corporation a party with whom the broker had previously dealt? Was a prior prospect a vital link to the buyer? Are all appropriate parties to the matter joined?

Initial Contact with the Buyer/Tenant: Who first introduced the buyer/tenant to the property? When was the first introduction made? Was the introduction made when the buyer/tenant had a specific need for that type of property? Was the introduction instrumental in creating the desire to purchase/lease? Did the buyer/tenant know about the property before the broker contacted him/her? Did he/she know it was for sale/lease? Were there previous dealings between the buyer/tenant and the seller/landlord? Did the buyer/tenant find the property on his/her own? How was the first introduction made? Was the property introduced at an "open house"? What subsequent efforts were made by the broker after the "open house"? Was the introduction made to a different representative of the buyer/tenant? Was the introduction merely a mention that the property was listed? Which property was first introduced?

Conduct of the Brokers: Were all required disclosures complied with? Was there a faithful exercise of the duties a broker owes to his/her client/principal? If more than one cooperating broker was involved, was either (or both) aware of the other's role in the transaction? Did the broker who made the initial introduction to the property engage in conduct (or fail to take action) which caused the buyer/tenant to utilize the services of another broker? Did the cooperating broker (or second cooperating broker) initiate a separate series of events, unrelated to and not dependent on any other broker's efforts, which led to the successful transaction — i.e., did the broker perform services which assisted the buyer/tenant in making the decision to purchase/lease? Did the broker make preparations to show the property to the buyer/tenant? Did the broker make continued efforts after showing the property? Did the broker remove an impediment to the sale/lease? Did the broker make a proposal upon which the final transaction was based? Did the broker motivate the buyer to purchase? How do the efforts of one broker compare to the efforts of another? What was the relative amount of effort by one broker compared to another? What was the relative success or failure of negotiations conducted by one broker compared to the other? If more than one cooperating broker was involved, how and when did the second cooperating broker enter the transaction?

Continuity and Breaks in Continuity (Abandonment and Estrangement): What was the length of time between the broker's efforts and the final sale/lease agreement? Did the original introduction of the buyer/tenant to the property start an uninterrupted series of events leading to the sale/lease, or was the series of events hindered or interrupted in any way? Did the buyer/tenant terminate the relationship with the broker? Why? Did negotiations break down? If there was an interruption or break in the original series of events, how was it caused and by whom? Did the seller/landlord change the listing agreement from an open listing to an exclusive listing agreement with another broker? Did the buyer/tenant's motive change? Was there interference in the series of events from any outside or intervening cause or party? Did the broker who made the initial introduction to the property maintain contact with the buyer/tenant or could the broker's inaction have reasonably been viewed by the buyer/tenant as a withdrawal from the transaction?

Was the entry of any cooperating broker into the transaction an intrusion into an existing relationship between the buyer/tenant and another broker or was it the result of abandonment or estrangement of the buyer/tenant or at the request of the buyer/tenant?

Conduct of the Buyer/Tenant: Did the buyer/tenant make the decision to buy independent of the broker's efforts? Did the buyer/tenant negotiate without any aid from the broker? Did the buyer/tenant seek to freeze out the broker? Did the buyer/tenant seek another broker in order to get a lower price/rate? Did the buyer/tenant express a desire not to deal with the broker and refuse to negotiate through him/her? Did the contract provide that no brokers or certain brokers had been involved?

Conduct of the Seller/Landlord: Did the seller/landlord act in bad faith to deprive the broker of his/her commission? Was there bad faith evident from the fact that the difference between the original bid submitted and the final sale price/lease rate equaled the commission? Was there bad faith evident from the fact that a sale/lease to a third party was a straw transaction (one in which an uninvolved party posed as the buyer/tenant) which was designed to avoid paying a commission? Did the seller/landlord freeze out the broker to avoid a commission dispute or to avoid paying a commission at all? Was there bad faith evident from the fact that the seller/landlord told the broker he/she would not sell/lease on certain terms, but did so via another broker or via the buyer/tenant directly?

Leasing Transactions: Did the cooperating broker have a tenant representation agreement? Was the cooperating broker working with the "authorized" staff member of the tenant company? Did the cooperating broker prepare a tenant needs analysis? Did the cooperating broker prepare a market analysis of available properties? Did the cooperating broker prepare a tour book showing alternative properties and conduct a tour? Did the cooperating broker show the tenant the property leased? Did the cooperating broker issue a request for proposal on behalf of the tenant for the property leased? Did the cooperating broker take an active part in the lease negotiations? Did the cooperating broker obtain the tenant's signature on the lease document? Did the tenant work with more than one broker? If so, why?

Other Issues: Is there any other information that would assist the arbitration Hearing Panel in having a full, clear understanding of the transaction giving rise to the arbitration request or in reaching a fair and equitable resolution of the matter?

Case # _____ SIOR-PS Form 16

Society of Industrial and Office Realtors[®] Professional Standards Process 1201 New York Avenue, NW, Suite 350 Washington, DC 20005 USA 202.449.8200 phone | 202.216.9325 fax

MEDIATION RESOLUTION AGREEMENT

Complainant(s):	_
Respondent(s):	_
The undersigned parties have participated in the mediat attached SIOR-PS Form 3, pursuant to the mediation pro and Arbitration Procedures Manual, and as a result of that their dispute:	cedures specified in SIOR's Ethics, Mediation
The undersigned parties agree to be bound by the above resubmit the controversy for arbitration or to litigate the mathematical harmless, and expressly waive any and all liability of SIOR SIOR arising out of the manner in which SIOR conducted reached as a result of SIOR's mediation procedures. Furtherforced, the non-complying party agrees to reimburse that torney's fees.	atter. Each party further agrees to hold SIOR, or any claims that the party may have against mediation, or the resolution of the controversy ther, in the event the award must be judicially
Complainant(s) Signature:	Date:
Respondent(s) Signature:	Date:
The undersigned, as mediation officer for SIOR, the mediation process and that the above resolut the parties to the dispute.	
The parties to this dispute were not able to res mediation conference has been terminated. The di	
Mediation Officer Signature:	Date:
Mediation Officer Name (printed):	

Professional Standards Process 1201 New York Avenue, NW, Suite 350 Washington, DC 20005 USA 202.449.8200 phone | 202.216.9325 fax

MEDIATION OFFICER SELECTION FORM

In the matter of	agreed to serve ediation, you hav ou believe would ation officers, in the are not accept	te the right to challenge all but two (2) of the dinot be acceptable to serve in this case dicate whether each is acceptable or no table.
·	,	
Mediation Officer Name	<u>Acceptable</u>	Not Acceptable — Reason
Mediation Party Name (type/print)		
Mediation Party Signature		
Date		
(Staff Use Only)		
Date Received by SIOR	Initia	/s

Case #	SIOR-PS Form 17A

Society of Industrial and Office Realtors[®] Professional Standards Process 1201 New York Avenue, NW, Suite 350 Washington, DC 20005 USA 202.449.8200 phone | 202.216.9325 fax

MEDIATION OFFICER SELECTION NOTICE

In the matter of	(Complainant) vs.
	(Respondent):
The parties are hereby notified that has agreed to serve as mediation officer for your	mediation conference.
The mediation officer will contact you directly mediate this issue.	to set the date for a conference call or live meeting to
will be referred for an arbitration hearing. Par process at any point prior to reaching an agreer	pitration system. If mediation is not successful, the matter rties to a mediation conference may withdraw from the ment. Should either party withdraw from the process prior roceed in accordance with SIOR's Ethics, Mediation and
reduced to writing and has been signed by all of will not be the subject of a subsequent arbitrationabide by the terms of the settlement, the man	ee to a settlement of the dispute and the settlement is of the parties, the matter will be considered resolved, and on hearing. In the event that either of the parties fails to atter may not be arbitrated; instead, the other party is adicially enforced by a court of competent jurisdiction.
is your intention to have legal counsel present	y and represented at the conference by legal counsel. If it at the mediation conference, the mediation officer or the ner parties should be advised of this fact at least ten (10)
conference. If the parties are unable to reso mediation officer may make a recommendation at the conclusion of the conference. Both partie proposed resolution and advise the media recommendation. Failure to respond by the dead	to a mutual resolution of the matter at any time during the alve the matter after discussing all pertinent facts, the for the resolution of the dispute, either orally or in writing as will be given a specified length of time to consider the tion officer of their acceptance or rejection of the dline will be deemed a rejection of the recommendation. If , the mediation conference will be deemed concluded and standards Committee for an arbitration hearing.
If you have questions about the mediation cor Staff Officer listed below.	nference, please contact SIOR's Professional Standards
Staff Officer	Date

Case # _____ SIOR-PS Form 18

Society of Industrial and Office Realtors[®] Professional Standards Process 1201 New York Avenue, NW, Suite 350 Washington, DC 20005 USA 202.449.8200 phone | 202.216.9325 fax

REQUEST FOR APPEAL OF AN ETHICS HEARING DECISION

TO:	President of the Society of Industrial and Office Realtors®
RE:	The Case of Complainant:
	versus
	Respondent:
FROM:	
	nereby request an appeal before the Board of Directors (or its designated appeal panel) in with SIOR's Ethics, Mediation and Arbitration Procedures Manual. My (our) appeal is based wing:
	one or more of the following and attach a narrative description of the facts and evidence of which support the bases cited.)
	procedural deficiency or procedural error resulting in an unfair hearing
	misapplication or misinterpretation of one or more provisions of the Code of Ethical Principles and Standards of Professional Practice
	inappropriateness or excessiveness of the discipline recommended by the Hearing Panel
	Other reasons as stated here:
•	est for Appeal form must be accompanied by a deposit in the amount of \$1,000, made payable erry of Industrial and Office Realtors [®] .
Only those the appeal	facts and issues raised in this written request will be considered by the Board of Directors at hearing.
Date Subm	itted:
Signature(s	s):
Typed/Print	ted Name(s):

ase #	SIOR-PS Forr
ase #	SIUR-PS FOR

Society of Industrial and Office Realtors[®] Professional Standards Process 1201 New York Avenue, NW, Suite 350 Washington, DC 20005 USA 202.449.8200 phone | 202.216.9325 fax

OFFICIAL NOTICE OF ETHICS APPEAL HEARING

case of, Compl	lainant, versus
ndent, the parties are hereby notified:	
Hearing Panel in the above cited case, based or	(Appellant) has appealed the decision of the A copy of the pecision (SIOR-PS Form 18), together with its
A Board of Directors appeal panel will meet or the appeal.	n(date) a (location) to hear
Appeal panel members include:	
	Hearing Panel in the above cited case, based or Request for Appeal of an Ethics Hearing D supporting documentation, is attached. A Board of Directors appeal panel will meet or time) at the appeal.

- 4. Any party may be represented by legal counsel and each party is requested to provide written notice of an intention to have legal counsel present at least ten (10) days before the scheduled hearing date. Failure to provide this notice will not invalidate a party's right to legal representation but may result in a continuance of the hearing if the appeal panel determines that the rights of any other party require representation by legal counsel.
- 5. Any party may file with the Professional Standards Staff Officer within ten (10) days a written request for disqualification from the appeal panel of any member listed above for the following reasons:
 - a. Anyone who is related by blood or other familial relationship to either the Complainant or the Respondent.
 - b. Anyone who is an employer, partner, or employee or directly associated in business with, or affiliated with any person or entity who is directly associated in business with, either the Complainant or the Respondent.
 - c. Anyone who was a party to the hearing or is a party to or testified as a witness in any other pending case involving the Complainant or the Respondent; or
 - d. Anyone who is in any other similar relationship or has knowledge which may prevent him/her from rendering an impartial decision.
 - e. Anyone who was present as a member of the Grievance review panel initially reviewing the ethics complaint, was a member of the Professional Standards Hearing Panel which

Case #	SIOR-PS Form	

reached the decision currently on appeal, or who was otherwise directly involved in the matter prior to the appeal.

f. Anyone who for any other reason may be prevented from rendering an impartial decision.

Challenges should be made using the accompanying SIOR-PS form 20, "Notice Of Right To Challenge Hearing Panel Members In An Ethics Or Arbitration Hearing." The challenge due date is shown on the top of that form.

6. The parties will not discuss the case with any member of the appeal panel, Governance Committee, or Board of Directors prior to the appeal hearing or after the appeal hearing and prior to announcement of the decision.

Professional Standards Staff Officer

Signature:

Printed Name:

Date of Issuance:

SIOR-PS Form 20 Case # _____

SOCIETY OF INDUSTRIAL AND OFFICE REALTORS® **Professional Standards Process** 1201 New York Avenue, NW, Suite 350 Washington, DC 20005 USA 202.449.8200 phone | 202.216.9325 fax

NOTICE OF RIGHT TO CHALLENGE HEARING PANEL MEMBERS IN AN ETHICS OR ARBITRATION APPEAL HEARING

Notice is given herewith to:			
VS.			
an _ethics / _arbitration appeal hearing, that a individual who may be appointed to serve on provided below. If you wish to challenge the cappeal Hearing Panel, please indicate by circlin to the Professional Standards Staff Officer with your reason(s) is deemed sufficient to suppose appointed to the appeal Hearing Panel for your SIOR's Ethics, Mediation and Arbitration Proceed ten (10) days from the date the list of names is set.	an appeal Hearing Panel. A qualifications of any of the inc ag the appropriate response ar h a statement explaining your our challenge, the indivic our case. In accordance with adures Manual, challenges mu	A list of such indi- dividuals listed to not return a copy of reason(s) for cha- lual challenged wanthe guidelines spans to st be filed with SI	viduals is sit on the f this form allenge. If vill not be pecified in
Appeal Panel Members		Challe	nge?
		YES	NO
	_	YES	NO
Please attach a written explanation of why any o	of the above are being challeng	red.	
Name (typed/printed):	Signature:		
Challenge Date:	Date Challenge Rec	eived:	

Case #	SIOR-PS Form 2 ^r

Professional Standards Process 1201 New York Avenue, NW, Suite 350 Washington, DC 20005 USA 202.449.8200 phone | 202.216.9325 fax

REQUEST FOR ARBITRATION HEARING PROCEDURAL REVIEW

TO:	President of the Society of Industrial and Office Realtors®
RE:	The Case of Complainant: versus Respondent:
FROM:	
arbitration hear	by request a procedural review by the Board of Directors (or its designated panel) of the ring held in the above-named case, as provided in the applicable provisions of SIOR's on and Arbitration Procedures Manual.
be based o	the merits of the arbitration award will not be made. A request for procedural review must on allegation(s) of procedural deficiency or other irregularities which may have deprived a air hearing. Only those facts and issues raised in this written request will be considered.
	based on the following facts: (Attach documents, including portions of hearing transcripts substantiate the occurrence of the procedural events or decisions that the appeal
Submitted by: _	
Date:	

Professional Standards Process 1201 New York Avenue, NW, Suite 350 Washington, DC 20005 USA 202.449.8200 phone | 202.216.9325 fax

OFFICIAL NOTICE OF ARBITRATION PROCEDURAL REVIEW

	case of ndent, the parties are hereby notified:	, Complainant, versus
1.		has requested a procedural review of the case. A copy of the Request for Arbitration Hearing ogether with its supporting documentation, is attached.
2.	A procedural review panel will meet on	(date) at(location) to hear rs be overturned.
3.	Procedural review panel members include	

- 4. Any party may be represented by legal counsel and each party is requested to provide written notice of an intention to have legal counsel present at least ten (10) days before the review hearing date. Failure to provide this notice will not invalidate a party's right to legal representation but may result in a continuance of the review if the procedural review panel determines that the rights of any other party require representation by legal counsel.
- 5. Either party may file with the Professional Standards Staff Officer within ten (10) days a written request for disqualification from the procedural review panel of any member listed above for the following reasons:
 - a. Anyone who is related by blood or other familial relationship to either the Complainant or the Respondent.
 - b. Anyone who is an employer, partner, or employee or directly associated in business with either the Complainant or the Respondent, or affiliated with any person or entity who is directly associated in business with.
 - c. Anyone who was a party to the hearing or is a party to or testified as a witness in any other pending case involving the Complainant or the Respondent.
 - d. Anyone who is in any other similar relationship or has knowledge which may prevent him/her from rendering an impartial decision.
 - e. Anyone who was present as a member of the Grievance review panel initially reviewing the arbitration request, was a member of the professional standards Hearing Panel which

Case #	SIOR-PS Form 22
Case #	310K-F3 F01111 22

reached the decision currently on appeal, or who was otherwise directly involved in the matter prior to the appeal.

f. Anyone who for any other reason may be prevented from rendering an impartial decision.

Challenges should be made using the accompanying SIOR-PS Form 20, "Notice Of Right To Challenge Hearing Panel Members In An Ethics Or Arbitration Appeal Hearing." The challenge due date is shown on the top of that form.

6. The parties will not discuss the case with any member of the procedural review panel, Governance Committee, or Board of Directors prior to the procedural review hearing or after the procedural review hearing and prior to announcement of the decision.

Signature:
Printed Name:
Date of Issuance:

Professional Standards Staff Officer

Society of Industrial and Office Realtors®
Professional Standards Process
1201 New York Avenue, NW, Suite 350
Washington, DC 20005 USA
202.449.8200 phone | 202.216.9325 fax

SIOR CONFIDENTIALITY STATEMENT

Please sign and return this form to SIOR Headquarters, 1201 New York Ave., NW, Ste. 350, Washington, DC 20005, ATT: Professional Standards Officer.

I understand that all SIOR professional standards proceedings are conducted in strict confidentiality, and I agree to limit any communication about any aspects of a case before, during or after any procedure or hearing to the chair of the Professional Standards Committee, the staff liaison, members of a Grievance Review Panel or a Hearing Panel, and to my SIOR advisor or legal counsel, if applicable.

Signature	() Complainant
9	() Respondent
Printed Name	() Professional Standards Committee Member
	() SIOR Advisor
Date	() Review Panel Member
	() Legal Counsel

SOCIETY OF INDUSTRIAL AND OFFICE REALTORS® (SIOR) Ethics, Mediation & Arbitration Procedures Manual

APPENDICES

APPENDIX A: CODE OF ETHICAL PRINCIPLES AND STANDARDS OF PROFESSION. PRACTICE OF THE SOCIETY OF INDUSTRIAL AND OFFICE REALTORS®	
TACTICE OF THE SOCIETY OF INDUSTRIAL AND OFFICE REALTORS	90
PREAMBLE TO THE CODE	96
Code Applicability and Intended Effects	
Administration, Interpretation and Enforcement	
Timely Filing of Ethics Complaints and Dispute Resolution Requests	
Other Professional Codes	
Understanding the Code: Recurrent Terms and Concepts	
Relationship Parties	
Party to an Act	
Pertinent Facts	
CODE OF FTUICAL PRINCIPLES AND STANDARDS OF PROFESSIONAL PRACTICE	400
CODE OF ETHICAL PRINCIPLES AND STANDARDS OF PROFESSIONAL PRACTICE	
Principle 1 – Act in the Best Interest of Those Who Should Benefit from Your Conduct Principle 2 – Be Honest at All Times and with All Parties	
Principle 2 – Be Honest at Ali Times and with Ali Parties Principle 3 – Honor Your Fiduciary Duties	
Principle 4 – Be Sensitive to Potential Conflicts of Interest	
Principle 5 – Disclose and Obtain Approval for Self-Dealing	
Principle 6 – Document All Agreements and Keep Accurate Records	
Principle 7 – Do Not Discriminate	
Principle 8 – Obey the Law	
Principle 9 – Other Codes Govern Certain Activities	
Principle 10 – Be Truthful in All Representations and Advertising	
Principle 11 – Do Not Make False or Misleading Statements about Others	107
Principle 12 – Disclose the Nature of Your Relationship	
Principle 13 – Offer to Cooperate with and Compensate Other Real Estate Professionals Principle 14 – Do Not Interfere with the Relationships or Rights of Other Real Estate	108
Professionals	
Principle 15 – Cooperate in All Professional Standards Proceedings	
Principle 16 – Resolve Financial Disputes through Mediation and/or Arbitration	110
APPENDIX B: SIOR CODE OF CONDUCT	112
SIOR Code of Conduct	
APPENDIX C: APPLICATION ETHICS COMPLAINT PROCEDURES	
Filing an Application Ethics Complaint	
What are the grounds for denying admission to SIOR?	
How do I file an application ethics complaint?	
Where do I send a completed complaint form?	
What happens to the membership application after a complaint is submitted?	
What are my obligations if I file a complaint?	113

APPENDIX A: CODE OF ETHICAL PRINCIPLES AND STANDARDS OF PROFESSIONAL PRACTICE OF THE SOCIETY OF INDUSTRIAL AND OFFICE REALTORS®

Approved by the Board of Directors on April 26, 1998 | Effective May 3, 1999

PREAMBLE TO THE CODE

The Society of Industrial and Office Realtors (§) is an association of commercial real estate specialists who are dedicated to the highest principles of professional practice. SIOR's Mission Statement declares that the "mission of SIOR is to establish, maintain, and sustain a designation of the highest level for commercial real estate providers," and its Vision Statement stipulates that "the designation will define excellence in performance . . . and that its designees are considered the pre-eminent professionals in their fields".

This Code embodies the principles that are the foundation of SIOR's mission and vision. The following paragraphs briefly describe SIOR context in which the Code operates and SIOR policies on interpretation and enforcement of the Code. That discussion is followed by a statement of the specific Principles and Standards of Practice that together form the Code.

Code Applicability and Intended Effects

The obligations imposed in this Code apply to all Active designees and Member Associates of SIOR ("Adherents"). These obligations may also be applied to conduct arising from the actions of others who are subject to the Adherents' administration, management, and/or control - such as conduct in which Adherents are involved in conjunction with an employee, associate, or partner.

The obligations imposed by the Code should be viewed as the terms of a contract to which all Adherents are bound. The terms of this contract - hence the obligations of the Code - apply to all real estate-related professional and business dealings, except to the extent specifically exempted by law. Some of the obligations imposed by this Code may exceed the minimum contents of the law – and are intended to do so.

The beneficiaries of the contract created by this Code are the clients, customers, principals and other parties who are served by or who conduct real estate-related business with SIOR's Active designees and Member Associates. These other parties may include other members of SIOR and other real estate professionals. The Code is also intended to help raise the level of the professional practice of Adherents to the benefit of SIOR, other professional associations, and the general public.

The concepts of "real estate related professional dealings" and "professional practice" include an Adherent's participation in the activities of professional associations and organizations. Consequently, an Adherent's fulfillment of responsibilities assumed as an officer, director, committee chairman, or committee member, or those that are otherwise assumed on behalf of SIOR, at either the national, regional, or chapter level, are subject to the requirements of this Code.

Administration, Interpretation and Enforcement

This Code is intended to be administered, interpreted, and enforced solely by SIOR. Anyone having reason to believe that an Adherent may have violated the Code may notify SIOR. SIOR will process any notice or complaint that it receives in accordance with its policies and procedures. Any Adherent who is found to have violated the Code may be sanctioned by SIOR.

Under this Code, Adherents are specifically obligated to participate in and cooperate with any investigation or proceeding arising from an inquiry or complaint concerning their professional conduct.

Adherents are also obligated to resolve certain types of financial disputes through mediation and/or binding arbitration.

The Code itself is only part of SIOR's Professional Standards regimen. The tasks required to enforce the Code and conduct ethics and arbitration hearings are performed by SIOR's Professional Standards Committee, whose makeup and jurisdiction are defined in SIOR's Policy Manual. Staff support for their activities is provided by a Professional Standards Staff Officer, who may be reached at SIOR's office in Washington, D.C. The procedures to be followed in initiating proceedings under this Code, for processing complaints, and the conduct of ethics and arbitration hearings are detailed in SIOR's Ethics, Mediation and Arbitration Procedures Manual, copies of which may be obtained from SIOR's office (phone: 202-449-8200).

Timely Filing of Ethics Complaints and Dispute Resolution Requests

Notices or complaints alleging ethical misconduct on the part of an Adherent must be filed within one (1) year of the latter of (a) the date of the closing of a sale from which the matter arises or to which it is related; (b) the commencement date of a lease (or its renewal or extension) from which the matter arises or to which it is related; (c) the rental commencement date of a lease (or its renewal or extension) from which the matter arises or to which it is related; (d) the date on which any litigation arising from or related to the matter is concluded; (e) the date on which any proceeding (arising from or related to the matter) before a licensing or regulatory authority is concluded; or (f) the date on which the matter is known, or could be known, in the exercise of reasonable diligence.

Notices or requests for the mediation and/or arbitration of financial disputes in which an Adherent is involved must be filed within one (1) year of the latter of: (a) the date of the closing of a sale from which the matter arises or to which it is related; (b) the commencement date of a lease (or its renewal or extension) from which the matter arises or to which it is related; (c) the rental commencement date of a lease (or its renewal or extension) from which the matter arises or to which it is related; or (d) the date on which the basis of the dispute is known, or could be known, in the exercise of reasonable diligence.

SIOR may consider conduct and other facts that predate the one (1) year filing period, such as those involved in a pattern of repeated ethical misconduct, provided there is a logical nexus between the conduct which falls within the filing period and that which precedes it.

Other Professional Codes

Adherents may also be subject to obligations arising from their membership in other associations or organizations, and to the obligations imposed by regulatory authorities for professional licensing purposes. This Code defines only an Adherent's obligations as an Active designee or Member Associate of SIOR. Conduct that is consistent with the obligations of certain other associations, organizations, or regulatory authorities may, nonetheless, be inconsistent with the requirements of this Code, but it is this Code to which an Adherent's conduct is measured by SIOR.

In the case of certain real estate-related activities of commercial specialists, such as appraisals or property management, the standards of other codes of specified professional organizations are adopted by reference as the requirements of this Code.

SIOR Adherents are also subject to the <u>Code of Ethics and Standards of Practice</u> of the NATIONAL ASSOCIATION OF REALTORS[®]. SIOR Code is substantially consistent with the REALTORS[®] Code, but is intended to apply specifically to commercial real estate scenarios. SIOR intends to continually attempt to harmonize its Code with the REALTORS[®] Code.

Understanding the Code: Recurrent Terms and Concepts

The Code consists of sixteen (16) Principles, each of which has several Standards of Practice associated with it. The Standards of Practice discuss specific applications, limits or exceptions to the Principle. The descriptive titles are for convenience only and do not themselves constitute interpretations or impose obligations.

The Code Principles and Standards are drafted using certain phrases or terms that are sometimes repeated and involve important concepts. Several of these are discussed here to assist Adherents in understanding the requirements of provisions in which these phrases are used, while avoiding the burdensome repetition that would result from describing them each time they appear. Reference may be made to the following paragraphs discussing such terms (thereby partially defining them) during the process of Code interpretation or enforcement.

Relationship Parties

The Principles and Standards of Practice establish a group of duties and related requirements, such as to undertake or to refrain from specified conduct or activities, to make disclosures, to withhold disclosures, to obtain acknowledgments or to obtain consents.

The words "client, customer, principal, or other party with whom the Adherent has a legally recognized professional or business relationship," or somewhat shortened forms of that language having the same intended meaning, are used frequently to describe the individuals or business entities to whom such duties are owed.

The breadth of this language in part reflects the changing and diverse activities of today's commercial practitioners, which place them in a wide variety of real estate-related professional or business roles and relationships, all of which the Code is intended to encompass.

The breadth of the clause also is intended to reflect the affirmative objective of the Code that Adherents will behave in a principled manner toward numerous parties, including but not limited to those who will be the source of their compensation. Some duties are owed to fewer than all participants in a transaction or activity, but others, chiefly the duty of honesty, are owed to virtually every party.

Party to an Act

Several provisions require that Adherents not "be a party to" an act of a specified type or having specified consequences. The obligation not to "be a party to" prohibited conduct is intended to be a broad obligation. First, it requires that Adherents not perform such an act or engage in such conduct themselves. Second, it requires that Adherents neither request others to perform prohibited acts nor assist others in their performing prohibited acts. Third, an Adherent has a duty not to ignore prohibited conduct by other Adherents, even where the observing Adherent has no control over the commission of the act. The third dimension of "party to" meshes directly with the Adherents' obligation established under Principle 15 and Standard of Practice 15.4 to report to SIOR potential violations of which they have first-hand knowledge.

Pertinent Facts

Several provisions impose obligations with respect to "pertinent facts," or "pertinent facts and information," often in the context of requiring their communication or disclosure to one or more parties. The precise meaning of these terms cannot be set forth in the abstract, and may vary significantly depending on the nature of a particular professional or business scenario. Use of the word "pertinent" does reflect use of that term in the REALTORS[®] <u>Code of Ethics</u>, and SIOR anticipates that the word will be given a consistent interpretation in the application of its Code.

Certain legal requirements are framed in terms of "material" facts. Where a fact is legally "material," it typically will also be "pertinent" within the meaning of SIOR's Code. References to pertinent facts sometimes are accompanied by references to "any facts, conditions, or information that might affect or assist" a relationship party. The latter is intended to be an even broader category, including matters that might be viewed as pertinent or which only have some probability of becoming pertinent.

CODE OF ETHICAL PRINCIPLES AND STANDARDS OF PROFESSIONAL PRACTICE

Principle 1 – Act in the Best Interest of Those Who Should Benefit from Your Conduct

Adherents will act in the best interest of their client, customer, principal, or other party with whom the Adherent has a legally recognized professional or business relationship. Adherents will subordinate pursuit of their own self-interest to the best interest of such other parties in circumstances where those two interests may conflict.

Standard of Practice 1.1 – Understand Your Legal and Ethical Duties

When acting in a real estate-related professional or business matter, Adherents will understand their legal and ethical duties, and will conform their actions to these duties.

Standard of Practice 1.2 - Honor the Commitments You Make and Faithfully Fulfill Your Obligations

When acting in a real estate-related professional or business matter, Adherents will honor the legal, contractual, and ethical commitments they have made, and will not unreasonably withhold, delay, or condition the faithful fulfillment of their obligations.

Standard of Practice 1.3 – Code Applies to All Real Estate-Related Professional and Business Dealings

Regardless of the capacity in which an Adherent is acting (i.e., as an agent, subagent, broker, finder, facilitator, counselor, advisor, manager, or principal), Adherents are obligated to conduct themselves in accordance with the requirements of this Code. If an Adherent is engaged in a real estate-related professional or business activity that is conducted by, through, or in conjunction with an employee, subordinate, colleague, or associate, the conduct of such other persons who are under the direction or are otherwise acting at the request of or in conjunction with the Adherent must conform to the requirements of this Code. Additionally, the Adherent will promote understanding of the Code and advocate behavior that conforms to the Code by persons who are within the Adherent's sphere of influence. As to certain types of activities, this Code incorporates by reference Codes adopted by other real estate-related professional organizations. In those circumstances, the specific standards of conduct specified by the referenced Code govern the Adherent's conduct.

Standard of Practice 1.4 – Avoid Dishonest Conduct

Adherents will not engage in any conduct that: (a) dishonestly advantages any party with whom the Adherent has a legally recognized professional or business relationship; (b) dishonestly disadvantages any party with whom the Adherent has a legally recognized professional or business relationship; (c) dishonestly interferes with the rights or interests of any party with whom the Adherent has a legally recognized professional or business relationship; or (d) dishonestly enriches the Adherent or others at the expense of any party with whom the Adherent has a legally recognized professional or business relationship.

Standard of Practice 1.5 – Avoid Misuse of Authority

Adherents will not dishonestly use any position as a corporate officer or director, public official, or other figure of authority: (a) to their own personal advantage; (b) to the advantage of their business colleagues or associates; (c) to the advantage or to the disadvantage of any client, customer, or other party with whom they have a legally recognized professional or business relationship; (d) to the disadvantage of a competitor; or (e) in any way that could potentially cause harm or liability to the company, public body, or any entity to which they owe legal, fiduciary, and/or ethical duties.

Standard of Practice 1.6 – Secure Informed Consent First

Adherents will not take any substantial action on behalf of a client, customer, principal, or other party with whom they have a legally recognized professional or business relationship without first obtaining the informed consent of such party as to the nature, objectives and expected consequences of the action. While this requirement is intended to foster a regime of knowledgeable supervision, it is not intended to require that Adherents be "micro-managed."

Standard of Practice 1.7 - Function Competitively

The commercial real estate profession is a competitive business, and numerous laws seek to assure the benefits of competitive markets to all participants. Adherents are expected to compete among themselves and with other real estate professionals. This Code imposes no ethical injunction intended to unlawfully: (a) impair or impede such competition; (b) limit the lawful types of services or activities in which Adherents may engage; (c) set the compensation rates or terms that Adherents may charge or receive; or (d) prohibit current or potential clients, customers, principals, or other parties from selecting the real estate professional(s) from whom to procure the services they desire.

Principle 2 – Be Honest at All Times and with All Parties

Adherents will be honest at all times with all parties including but not limited to parties with whom they have a legally recognized professional or business relationship.

Standard of Practice 2.1 – Avoid Exaggeration, Misrepresentation, Withholding, or Concealment of Pertinent Facts

Adherents will avoid exaggeration, misrepresentation, withholding, or concealment of pertinent facts relating to a property, transaction, or matter in which they are involved in a professional or business capacity.

Standard of Practice 2.2 – Provide Requested Information

Adherents will timely provide information that is reasonably requested by a client, customer, principal, or other party having an interest in a property, transaction, or matter, unless the provision of such information would violate the Adherent's duty of confidentiality, or unless the provision of such information is excepted under Standard of Practice 2.3.

Standard of Practice 2.3 – Excepted Undertakings and Disclosures

Adherents are not expected (a) to discover, disclose, or advise on matters or adverse conditions that require special expertise or a technical discipline outside of the legal requirements of their professional license, (b) to disclose factors that are defined as "nonmaterial" by law or regulation, or (c) to provide information where doing so would violate a specifically applicable legal obligation.

Standard of Practice 2.4 - Certain Dishonest Acts to Be Avoided

Adherents will not be a party to any act that: (a) involves the knowing falsification of information; (b) attempts to deceive; (c) attempts to defraud; (d) attempts to mislead; or (e) involves a breach of faith, trust, or fiduciary duty.

Principle 3 – Honor Your Fiduciary Duties

Adherents will fulfill, without breach, the fiduciary duties of loyalty, obedience, disclosure, confidentiality, good faith, due care, and accounting that they owe to their client, customer, principal, or other party with whom they have a legally recognized professional or business relationship.

Standard of Practice 3.1 – Duty of Loyalty

Adherents will be loyal to their client, customer, principal, or other party with whom they have a legally recognized professional or business relationship, and will faithfully conform their actions to the interests, needs, and objectives of such party or parties.

Standard of Practice 3.2 – Duty of Obedience

Adherents will be obedient to their client, customer, principal, or other party with whom they have a legally recognized professional or business relationship, and will conform their actions to the instructions, directions, and objectives of such party or parties. However, Adherents will not conduct themselves in a way that violates any legal or ethical duty or obligation that they owe to any other party.

Standard of Practice 3.3 - Duty of Disclosure

Adherents will disclose all pertinent facts and information to their client, customer, principal, or other party with whom they have a legally recognized professional or business relationship. Such required disclosures include, but are not limited to, those mandated by law. In addition, Adherents will keep each relationship party reasonably informed of any facts, conditions, or information that might affect or assist such party.

Standard of Practice 3.4 - Duty of Confidentiality

Adherents will maintain the confidentiality of information that concerns their client, customer, principal, or other party with whom they have a legally recognized professional or business relationship, and will not knowingly, during or following the termination of a real estate-related professional or business relationship: (a) reveal confidential information of or about such party; (b) use confidential information of or about such party to the disadvantage of the party; (c) use confidential information of or about such party to the Adherent's advantage; or (d) use confidential information of or about such party to the advantage of any other party, including, without limitation, to: (i) other clients, customers, or principals with whom the Adherent has a legally recognized professional or business relationship, (ii) the Adherent's colleagues and associates, and (iii) the Adherent's friends and family relations. The only exceptions to an Adherent's absolute duty of confidentiality are: (a) the transmission of confidential information after disclosure to and approval from all affected parties; (b) the transmission of confidential information is believed by the Adherent to be required to prevent the commission of a crime; or (d) where the use or transmission of confidential information is required to defend the Adherent or the Adherent's employees or associates against accusations of wrongful conduct.

Standard of Practice 3.5 - Duty of Good Faith and Due Care

Adherents will exercise good faith and due care in actions affecting their client, customer, principal, or other party with whom they have a legally recognized professional or business relationship, and will conduct themselves in a manner consistent with the confidence, trust, and reasonable expectations of such party.

Standard of Practice 3.6 - Duty of Accounting

Adherents will provide their client, customer, principal, or other party with whom they have a legally recognized professional or business relationship with a true, complete, timely, and accurate accounting of all financial facts and information. Adherents will not seek to deny, hinder, or obstruct such party's access to such financial facts and information. Adherents have an affirmative duty to provide such financial facts and information to such party.

Principle 4 – Be Sensitive to Potential Conflicts of Interest

In situations involving a potential for conflict of interest, Adherents will subordinate their interests to the interests of their client, customer, principal, or other party with whom they have a legally recognized professional or business relationship.

Standard of Practice 4.1 – Client's Interests Come First

Adherents will not place their rights in, or entitlement to, a fee or commission before the interests of their client, customer, principal, or other party with whom they have a legally recognized professional or business relationship. This does not mean that Adherents must forgo their claim to compensation to which they are legally entitled. However, Adherents must not allow their entitlement to compensation to take precedence over the interests of such parties in their conduct.

Standard of Practice 4.2 - Non-Interference with Client's Rights and Interests

Adherents will not take any action to obstruct or impede the rights or interests of their client, customer, principal, or other party with whom they have a legally recognized professional or business relationship.

Standard of Practice 4.3 – Permitted Exceptions

Adherents may represent owners, buyers, tenants, or other parties that may have competing interests. Adherents may buy, own, and develop real estate for their own account. Adherents may also charge and earn fees for the services they provide on behalf of clients, customers, principals, or other parties. However, to the extent that any activity in which an Adherent may engage has the appearance of a potential conflict, the Adherent will make proper disclosures to and seek to obtain the acknowledgment of those parties with whom a conflict may arise. In the event any of those parties acknowledges the disclosure but objects to an Adherent's continuing with the activity that involves the appearance of conflict, then the Adherent will refrain from doing so.

Principle 5 - Disclose and Obtain Approval for Self-Dealing

Adherents will not engage in any activity that may be deemed to constitute self-dealing without disclosure to and approval from all affected parties.

Standard of Practice 5.1 - Disclosure of Interest

Adherents will not provide real estate-related professional services to any affected party when they have a present or contemplated interest in a property that is the subject of an assignment or transaction, without making prior disclosure of such interest to all affected parties and receiving prior approval from all affected parties.

Standard of Practice 5.2 – Acquisition of Ownership Interest

Adherents will not acquire an interest in a property or present offers from themselves, any person with whom they have a familial relationship, their firm or any members of their firm, any person with whom they are involved in a business venture, or any entity in which they have a present or contemplated interest, without disclosing their true position to all affected parties prior to the commencement of negotiations.

Standard of Practice 5.3 – Professional Services for Owned Properties

Adherents will not provide real estate-related professional services for a property in which they have a beneficial interest without first obtaining approval for such activities, including, without limitation, an agreement regarding compensation and costs with the other parties who also have a beneficial interest in such property.

Standard of Practice 5.4 – Rebates from Expenditures Made for Client

Adherents will not accept any commission, rebate, or profit on expenditures they make for goods or services that are obtained for or provided to their client, customer, principal, or other party with whom they have a legally recognized professional or business relationship without making prior disclosure to and receiving prior approval of such party.

Standard of Practice 5.5 – Unearned Compensation

Adherents will not accept commissions, fees, or other non-nominal benefits from the vendors or providers of goods or services purchased by a client, customer, principal, or other party with whom they have a legally recognized professional or business relationship, without making prior disclosure to and receiving prior approval of the party making such purchase.

Standard of Practice 5.6 – Undisclosed Participation

Adherents will not provide ancillary real estate-related services (such as property maintenance, construction, or environmental remediation) to a client, customer, principal, or other party with whom they have a legally recognized professional or business relationship, through a business or enterprise in which they have a beneficial interest or from which they receive compensation or other benefits, without making prior disclosure to and receiving prior approval of all affected parties.

Standard of Practice 5.7 – Undisclosed Compensation

Adherents will not accept compensation from more than one party in a transaction, even if permitted by law, without making prior disclosure to and receiving prior approval from all affected parties.

Principle 6 – Document All Agreements and Keep Accurate Records

Adherents will use reasonable diligence to assure that all offers, contracts, agreements, understandings, disclosures, acknowledgments, approvals and consents are properly documented, transmitted to and acknowledged by all affected parties, and that the specific terms, conditions, obligations, commitments, and understandings of the parties are clearly and fully articulated. Adherents will use reasonable diligence to assure that complete and accurate records of all real estate-related professional assignments and business dealings in which they are engaged are properly assembled and maintained for an adequate period of time.

Standard of Practice 6.1 – Types of Agreements

The agreements and related communications intended to be covered by this obligation to document include, but are not limited to: listing agreements, agreements of sale, lease agreements, offers to buy or lease, letters of intent, notices, financing agreements, extensions of contracts, compensation agreements, offers to and acknowledgments of terms of cooperation with other real estate professionals, offers to and acknowledgments of terms of compensation with other real estate professionals, partnership agreements, management agreements, referral agreements, and other contracts and substantial communications that are reasonably expected or required in any transaction, assignment, or real estate-related business or professional matter.

Standard of Practice 6.2 - Copies to All Parties

Adherents will use reasonable diligence to assure that a copy of each document is provided to each party to an agreement at the time of its execution, acceptance, or acknowledgment, and thereafter upon the reasonable request of any party having a direct interest in the terms arising from such document.

Standard of Practice 6.3 – False Consideration

Adherents will not be party to the naming of a false consideration in any document, unless it is the naming of an obviously nominal consideration.

Standard of Practice 6.4 – Scope and Duration of Record Retention

The types of records intended to be covered by the obligation to assemble records include, but are not limited to: correspondence, notices, financial records, bank statements and records, tax-related records, legal records, and corporate and partnership records. The period of time for which these records must be maintained is intended to be no less than the period required by law. Adherents are obligated to determine (a) the specific types of records that they are required to maintain and (b) the specific period of time that they are legally required to maintain such records. The time period of retention should, at minimum, be long enough to ensure that the relevant records are available for use in SIOR ethics or arbitration proceedings.

Standard of Practice 6.5 – Availability and Copies of Records

Adherents will make the records that they are obligated to maintain available for the reasonable inspection of any party having a direct interest in such records, and, upon reasonable request, will provide copies of such records to such parties.

Standard of Practice 6.6 – Avoid Falsification of Information

Adherents will not be a party to the knowing falsification of information in any document or business record.

Principle 7 - Do Not Discriminate

Adherents will not discriminate, or participate in any plan or agreement to discriminate, on the basis of race, creed, color, religion, gender, handicap, national origin, familial status, age, sexual orientation or any other category or characteristic defined as discriminatory by law.

Standard of Practice 7.1 - Demographic Reports

Adherents may provide demographic information (related to a property, transaction or assignment) to a client, customer, principal, or other party with whom they have a legally recognized professional or business relationship, if such demographic information is (a) needed to assist with or complete in a non-discriminatory manner a real estate transaction or professional assignment; (b) is obtained or derived from a recognized, reliable, independent, and impartial source; and (c) the source of such information and any additions, deletions, modifications, interpretations, or other changes are disclosed in reasonable detail.

Standard of Practice 7.2 - Credit and Financial Reports

Adherents may provide credit or financial information (concerning, as may be applicable, a prospective buyer, tenant, landlord, etc.) to a client, customer, principal, or other party with whom they have a legally recognized professional or business relationship, if such information is (a) needed to assist with or complete in a non-discriminatory manner a real estate transaction or professional assignment; (b) is obtained or derived from a recognized, reliable, independent, and impartial source; and (c) the source of such information and any additions, deletions, modifications, interpretations, or other changes are disclosed in reasonable detail.

Principle 8 – Obey the Law

Adherents will conform their real estate-related professional and business conduct to all applicable principles and requirements of law.

Standard of Practice 8.1 - Licensing Requirements

Adherents will conduct themselves in accordance with the requirements of the licensing authority in any jurisdiction in which they conduct business.

Standard of Practice 8.2 - Commingling of Funds

Adherents will maintain funds belonging to clients, customers, or other parties in escrow, fiduciary, or trust accounts, and will not commingle the funds of others with their own personal or business funds. Adherents will administer such accounts either: (1) in the manner agreed to, or as otherwise directed, by the appropriate parties; (2) as provided by law or regulation; or (3) in the absence of such guidance, in a manner that will protect such funds from losses and yield a reasonably favorable interest rate of return for the benefit of the appropriate parties.

Standard of Practice 8.3 – Use or Conversion of Funds or Assets

Adherents will not access, use, apply, or convert the funds, property, or assets belonging to a client, customer, principal, or other party with whom they have a legally recognized professional or business relationship in any way that is inconsistent with the rights or interests of such party.

Standard of Practice 8.4 – Unauthorized Practice of Law

Adherents will not engage in activities that constitute the unauthorized practice of law, and will recommend that legal counsel be engaged whenever Adherents believe that the interest of any party requires it.

Standard of Practice 8.5 – Activities Requiring Professional Licenses

Adherents will not engage in activities for which a professional license is required unless they have such a license, and will, whenever Adherents believe that the interest of any party requires it, recommend the engagement of a properly licensed professional when they do not have such a license.

Principle 9 – Other Codes Govern Certain Activities

In addition to all applicable legal requirements, Adherents will conform their real estate-related professional and business conduct to the ethical and Professional Standards that are reasonably expected in the specific disciplines in which they engage.

Standard of Practice 9.1 - Commercial Real Estate

The practices of Adherents of SIOR who engage in industrial, office, retail, investment, and other commercial real estate activities will, with respect to such activities, conform to this Code.

Standard of Practice 9.2 – Residential Real Estate

The practices of Adherents who engage in residential real estate activities will, with respect to such activities, conform to the <u>Code of Ethics and Standards of Practice</u> of the NATIONAL ASSOCIATION OF REALTORS[®].

Standard of Practice 9.3 – Appraisal and Valuation

The practices of Adherents who engage in appraisal activities or the valuation of real property will, with respect to such activities, conform to the <u>Uniform Standards of Professional Appraisal Practice</u> (USPAP) promulgated by the Appraisal Standards Board of the Appraisal Foundation. This requirement will not apply to a broker's pricing opinion, a broker's opinion of value, or to pricing opinions that are incidental to the listing, sale, or lease of a property.

Standard of Practice 9.4 – Property and Asset Management

The practices of Adherents who engage in the management of real estate will, with respect to such activities, conform to the <u>Code of Professional Ethics</u> of the Institute of Real Estate Management (IREM), and will be conducted: (a) with due regard for the rights, responsibilities, benefits, safety, and health of tenants and others lawfully on the premises; and (b) with due diligence and reasonable efforts to protect the property and all affected parties from reasonably foreseeable contingencies and losses.

Standard of Practice 9.5 - Counseling

The practices of Adherents who engage in real estate counseling and advisory services which do not include real estate brokerage, with respect to such activities, shall conform to the Standards of Professional Practice of the Counselors of Real Estate (CRE), the counseling affiliate of the NATIONAL ASSOCIATION OF REALTORS[®]. Any fee or compensation for such counseling or advisory services will not be conditioned or contingent on the substance of the advice or counsel given. If counseling or advisory services are provided in addition to other professional services, the compensation for such other services must be agreed upon separately.

Principle 10 – Be Truthful in All Representations and Advertising

Adherents will present a true picture in all real estate-related representations and advertising, and will clearly identify their professional role and/or business interest in the matter being communicated or promoted.

Standard of Practice 10.1 - Authority to Advertise or Make Representations

Adherents will not advertise or promote a property, assignment, or requirement, or make any other representations about a property, assignment, or requirement, without the authority of their client, customer, principal, or other party with whom the Adherent has a legally recognized professional or business relationship.

Standard of Practice 10.2 – Proper Identification of Role and Professional Status in Advertising and Promotional Material

Adherents will identify, with reasonable clarity, in their advertising and promotional material: (a) the name of the firm through which they are licensed to practice; (b) the capacity in which they are working (e.g., as the owner's exclusive agent, as the tenant's exclusive agent, as the manager of the property, as an owner of the property, etc.); and (c) whether they, or any member of their firm, or any person with whom they have a familial relationship, have an interest in the property or other relationship as a principal with the matter.

Standard of Practice 10.3 - Honest, Accurate and Complete Representations

Unless such representations conflict with the legal or ethical duties that Adherents owe to a client, customer, principal, or other party with whom they have a legally recognized professional or business relationship (such as their duty of confidentiality to a client), the representations that Adherents make in the context of real estate-related professional and business activities will be honest, accurate, and complete. Adherents will identify, with reasonable clarity, information that is subjective, speculative, or otherwise conditioned on facts or circumstances that are not known. Adherents will clearly identify the source or basis of all empirical information. Adherents will not withhold pertinent facts and information of which they are aware, and will keep each relationship party reasonably informed of any facts, conditions, or information that might affect or assist such party.

Standard of Practice 10.4 - Professional Credentials

Adherents (a) who are Active designees will identify themselves as SIORs in accordance with SIOR's policies in their advertising and representations; (b) if they choose or are required to, will identify accurately any other professional affiliation or designation which they possess; (c) will not make inaccurate, exaggerated, or misleading representations about their professional experience, capabilities, credentials, or expertise; and (d) will not make inaccurate, exaggerated, or misleading representations about the professional experience, capabilities, credentials, or expertise of their firm, or of others with whom they are affiliated.

Standard of Practice 10.5 - "Free of Charge" Services

Adherents will not promote or offer a service described as "free of charge" (or words to that effect) unless they simultaneously disclose: (I) their expectation, if any, of a commission, fee or other benefit (such as a listing or other assignment); and (2) the source of such expected commission, fee or other benefit.

Standard of Practice 10.6 - Premiums and Prizes

Adherents will exercise reasonable care and candor when offering premiums, prizes, or other inducements or benefits in any promotion, advertising, or representation. Any such offer should not mislead any party interested in receiving or otherwise benefiting from the offer. The terms, conditions, and limitations of such offer must be clearly articulated in advance.

Principle 11 – Do Not Make False or Misleading Statements about Others

Adherents will not knowingly or recklessly make any false or misleading statement about their competitors, their competitors' firms, or their clients, customers, principals, or other parties with whom they have a legally recognized professional or business relationship.

Standard of Practice 11.1 – Unsolicited Comments

Adherents will refrain from making unsolicited comments about their competitors, their competitors' firms, or their clients, customers, principals, or other parties with whom they have a legally recognized professional or business relationship.

Standard of Practice 11.2 - Objective Opinion

In instances where their opinion about a competitor, a competitor's firm, or a client, customer, principal, or other party with whom they have a legally recognized professional or business relationship is sought, or where they believe that such opinion is required, Adherents will render such opinion in an impartial and objective manner. Where needed to avoid misinterpretation of the opinion or its significance, the presentation should include a description of the basis for the opinion.

Principle 12 – Disclose the Nature of Your Relationship

Adherents may represent a client, principal, customer, or other party in any capacity permitted by law, but must reasonably advise the affected parties about the professional implications of serving in that capacity.

Standard of Practice 12.1 - Pre-engagement Disclosures

As early as commercially practical and before performing real estate-related services for a client, customer, principal, or other party with whom they have a legally recognized professional or business relationship, an Adherent will, to the extent that such disclosures apply to the nature of the assignment, advise such party of (a) the nature of the agency or non-agency relationship in which they will be involved; (b) the pertinent facts and information regarding the Adherent's compensation; (c) the policy of the Adherent's firm regarding cooperation with and compensation of other real estate professionals; (d) the nature and professional implications of possible relationships with other real estate professionals (such as subagency), including reasonably foreseeable compensation issues that may arise; and (e) if there is any potential for the Adherent or the Adherent's firm to become involved in representational conflicts, such as dual agency, and the manner in which such conflicts may be resolved.

Standard of Practice 12.2 – Preexisting Obligations

As early as commercially practical in the context of providing real estate-related services for a client, customer, principal, or other party with whom they have a legally recognized professional or business relationship, and to the extent applicable to the nature of a specific assignment, Adherents will use reasonable efforts to determine if any other real estate professional has a prior right or interest arising from the same property or a relevant prior agreement or relationship. Adherents will not subject a client, customer, principal, or other party with whom they have a legally recognized professional or business relationship to any obligation to pay additional compensation related to such prior existing right without first obtaining such party's informed consent.

Principle 13 - Offer to Cooperate with and Compensate Other Real Estate Professionals

Adherents will cooperate with other real estate professionals except when cooperation is not in the best interests of their client, customer, principal, or other party with whom they have a legally recognized professional or business relationship. The obligation to cooperate with other real estate professionals is not an obligation to share commissions or fees, or to otherwise compensate other real estate professionals. Where consistent with the best interests of and authority granted by their client, customer, principal, or other party with whom they have a legally recognized professional or business relationship, Adherents will, when appropriate, offer compensation to other real estate professionals and will timely disclose the terms and conditions related to the compensation offered.

Standard of Practice 13.1 – Terms of Cooperation

Adherents have the right to establish the terms and conditions by which they will cooperate with other real estate professionals, and will (a) disclose these terms and conditions to their client, customer, principal, or other party with whom they have a legally recognized professional or business relationship as soon as commercially practical when seeking authority to perform any real estate-related services on behalf of such party (which shall always be prior to consummation of a binding agreement); and (b) articulate these terms and conditions to other real estate professionals when soliciting their cooperation or upon inquiry from any real estate professional seeking cooperation.

Standard of Practice 13.2 – Terms of Compensation

Adherents have the right to establish the terms and conditions by which they will compensate other real estate professionals, and will (a) disclose these terms and conditions to their client, customer, principal, or other party with whom they have a legally recognized professional or business relationship as soon as commercially practical when seeking authority to perform any real estate-related services on behalf of such party; and (b) articulate these terms and conditions to other real estate professionals when soliciting their cooperation or upon inquiry from any real estate professional seeking cooperation and/or compensation.

Standard of Practice 13.3 – Seeking Cooperation or Compensation

Adherents who are seeking to cooperate with or to be compensated by another real estate professional will determine the terms of cooperation and/or compensation as soon as commercially practical in the context of providing real estate-related services for a client, customer, principal, or other party with whom

they have a legally recognized professional or business relationship. Adherents will not assume that another real estate professional's offer of cooperation is an offer of compensation.

Standard of Practice 13.4 – Changes in Terms of Cooperation or Compensation

Adherents who have made an offer to cooperate with and/or an offer to compensate other real estate professionals have an affirmative obligation to promptly communicate any change in the terms of cooperation or compensation to any parties to whom the previous terms were previously offered.

Standard of Practice 13.5 – Negotiations to Change Terms of Cooperation or Compensation

Offers to cooperate and/or offers to compensate other real estate professionals, unless accepted, will not preclude Adherents from negotiating, in good faith, agreements that deviate from such terms. Further, even after the terms of cooperation and/or compensation have been offered and accepted, Adherents may negotiate, in good faith, for a change in such terms.

Principle 14 – Do Not Interfere with the Relationships or Rights of Other Real Estate Professionals

Adherents will respect the agency or other exclusive relationships, recognized by law, of other real estate professionals, and will not engage in any practice or take any action that is inconsistent with such relationships or rights.

Standard of Practice 14.1 - Abandonment

Adherents will not attempt to represent a party in an ongoing transaction or assignment in which another real estate professional has been representing such party, unless it is first determined through direct inquiry to all affected parties that the transaction or assignment has been abandoned by such other real estate professional.

Standard of Practice 14.2 – Alienation and Estrangement

Adherents will not attempt to represent a party in an ongoing transaction or assignment in which another real estate professional has been representing such party, unless it is first determined through direct inquiry to all affected parties that such other real estate professional, although retaining related rights to compensation, has become so alienated or estranged from such transaction or assignment as to no longer be operative in it. However, Adherents may not be a cause of, or a party to, the alienation or estrangement of another real estate professional involved in an ongoing transaction or assignment.

Standard of Practice 14.3 – Termination

Adherents may represent a party in an ongoing transaction or assignment in which another real estate professional has been involved, if it is first determined that such other real estate professional's representational role in the transaction or assignment has been terminated. However, Adherents may not be a cause of, or a party to, the termination of such other real estate professional's representational role in such ongoing transaction or assignment.

Standard of Practice 14.4 – No Other Exceptions

Adherents will not represent a party in an ongoing transaction or assignment in which another real estate professional has been representing such party, if the other real estate professional has not abandoned such transaction or assignment, has not been alienated or estranged from such transaction or assignment, or has not been terminated from such transaction or assignment.

Standard of Practice 14.5 – Compensation Rights

If Adherents are engaged to represent a party in an ongoing transaction or assignment, they will not take any action inconsistent with the compensation rights of other real estate professionals who may have been or may be representing any party to such transaction or assignment.

Principle 15 - Cooperate in All Professional Standards Proceedings

Adherents will cooperate with any professional standards investigation or proceeding, and will take no action to impede or disrupt such investigation or proceeding.

Standard of Practice 15.1 - Pertinent Facts

Adherents who are involved in a professional standards proceeding as a party or as a witness will: (a) produce all requested documents, information, and other potential evidence; (b) place all pertinent facts before the proper committees or tribunals; and (c) testify fully and truthfully about the matter.

Standard of Practice 15.2 – Proceedings before Other Boards

In addition to proceedings before SIOR, Adherents may be subject to Professional Standards investigations or proceedings in which issues of professional conduct have been finally adjudicated before other professional Boards or Associations on matters related to the same transaction or event. Adherents are required to advise SIOR if they are named as a respondent party in any other Professional Standards proceeding in which issues of professional conduct have been finally adjudicated. The advisory to SIOR must be made promptly after they receive notice of the final disposition of such Professional Standards matter from another Board or Association. SIOR may then investigate such matter, and may initiate professional standards proceedings in accordance with its policies and procedures.

Standard of Practice 15.3 – Proceedings before Licensing Authorities, Regulatory Authorities, or Courts

In addition to proceedings before SIOR, Adherents may be subject to investigations or proceedings before real estate licensing authorities or regulatory authorities, or may be subject to civil or criminal judicial proceedings on matters related to the same transaction or event. Adherents are required to advise SIOR if they are named as a respondent party in any licensing or regulatory proceeding, or as a defendant in any judicial proceeding, in which issues of professional conduct have been finally adjudicated. The advisory to SIOR must be made promptly after they receive notice of the final disposition of such proceeding from a real estate licensing authority, regulatory authority, or court. SIOR may then investigate such matter, and may initiate professional standards proceedings in accordance with its policies and procedures.

Standard of Practice 15.4 – Duty to Report Potential Violations

Adherents having first hand knowledge of the potential violation of this Code by another Adherent are required to promptly report any such potential violation to SIOR. SIOR may investigate such matter, and may initiate professional standards proceedings in accordance with its policies and procedures.

Standard of Practice 15.5 - Obstruction of Proceedings

Adherents who are involved in a professional standards matter as a party or as a witness will not: (a) hinder or obstruct any professional standards process or proceeding; or (b) institute or threaten to institute any action for libel, slander, or defamation against any party involved in such professional standards matter. Adherents expressly waive any cause of action for liable, slander, or defamation that might arise from the filing or consideration of any professional standards matter before SIOR in which they may be a participant.

Standard of Practice 15.6 – Confidentiality of Proceedings

Adherents will not discuss, disclose, or disseminate any information regarding the allegations, findings, or decisions of any professional standards matter, except as authorized by SIOR.

Standard of Practice 15.7 – Multiple Filings

Adherents will not file multiple or repetitive complaints against the same party that are based upon the same transaction or event. This does not preclude Adherents from amending a complaint after it has been filed to expand the allegations or scope of a complaint, or from filing a new complaint if new facts or information become known or available.

Principle 16 – Resolve Financial Disputes through Mediation and/or Arbitration

Unless all affected parties agree otherwise, Adherents are required to resolve compensation and related financial disputes that arise from their real estate-related professional and business activities through mediation and/or binding arbitration. Further, unless the disputes involve claims that extend substantially

beyond a disagreement over compensation or the fulfillment of the terms of a contract (such as allegations of unlawful conduct), Adherents will not litigate such disputes.

Standard of Practice 16.1 - Disputes between Adherents

In the event two or more Adherents, who are employed by different firms, claim a right, entitlement, or interest in a fee or commission arising from or related to the same real estate transaction, the Adherents will submit such dispute to mediation and/or binding arbitration in accordance with the policies and procedures of SIOR.

Standard of Practice 16.2 – Financial Disputes with Clients, Customers, or Principals

If a client, customer, principal, or other party with whom an Adherent has a legally recognized professional or business relationship wishes to resolve a real estate-related financial dispute that arises from such professional or business relationship through mediation or arbitration, and such party agrees to be bound by the resulting decision, such dispute will be submitted to mediation and/or binding arbitration in accordance with the policies and procedures of SIOR.

Standard of Practice 16.3 - Financial Disputes with Non-Adherent Real Estate Professionals

If Adherents become involved in financial disputes with non-Adherent real estate professionals of other firms arising from real estate-related activities, Adherents are obligated to submit such disputes to mediation and/or binding arbitration. Such mediation and/or binding arbitration may be conducted, in accordance with the request of the non-Adherent: (a) through SIOR (provided the non-Adherent agrees to be bound by the arbitration decision); (b) through a local Board or State Association (provided (i) the non-Adherent is obligated to submit the dispute to mediation and/or arbitration and (ii) alternative dispute resolution facilities are available through the local Board or State Association); or (c) through other alternative dispute resolution sources (provided the non-Adherent is obligated or agrees to be bound by the arbitration decision).

Standard of Practice 16.4 – SIOR Prefers Alternative Dispute Resolution to Litigation

If Adherents become involved in financial disputes that do not otherwise conform to the Standards of Practice under this Principle, Adherents are obligated to recommend, seek, and agree to be bound by alternative dispute resolution procedures that are available through the American Arbitration Association and other recognized providers. To the greatest extent reasonably possible, Adherents will specify alternative dispute resolution preferences in the contracts and agreements that they employ in their real estate-related professional and business activities.



APPENDIX B: SIOR CODE OF CONDUCT

In 2019, the SIOR Board of Directors adopted the following *Code of Conduct*. As distinguished from the *Code of Ethical Principles and Standards of Professional Practice*, which relates to professional practice, the Code of Conduct relates to personal conduct at SIOR events and in member communications related to SIOR business. The *Code of Ethical Principles and Standards of Professional Practice* is enforced through the professional standards grievance review and hearing process, while the *Code of Conduct* is enforced through investigation and disciplinary action by SIOR executive leadership.

SIOR Code of Conduct

SIOR is committed to maintaining an inclusive and supportive professional environment that is free of discrimination, intimidation, or harassment. SIOR's Code of Conduct applies during all SIOR functions, and any member communications related to SIOR business or with SIOR staff. Members are expected to conduct themselves in a professional manner, and to act with courtesy and respect toward each other, SIOR staff, service providers, speakers, and meeting participants. This policy is reflective of SIOR's mission and values.

Any form of harassment, including sexual harassment, or discriminatory behavior based on race, color, national origin, sex, age, religion, disability, or any other protected characteristic is strictly prohibited. Harassment includes, but is not limited to verbal, non-verbal or physical conduct that denigrates or shows hostility toward an individual on the basis of a protected characteristic that (1) has the purpose or effect of creating an intimidating, hostile or offensive environment; or (2) has the purpose or effect of interfering with an individual's performance or ability to participate in SIOR events.

Any SIOR member who believes they have experienced or witnessed harassment or other inappropriate behavior in violation of this policy should promptly report the incident to the SIOR Chief Executive Officer, Chief Operating Officer, or SIOR President. Retaliation against any person for either reporting an incident or cooperating in an investigation is prohibited.

Upon receipt of a report of possible harassment or inappropriate behavior in violation of this policy, SIOR will promptly conduct an investigation, at the direction of SIOR appointed counsel. SIOR will maintain confidence to the extent possible, while still allowing for an appropriate investigation. If it is determined that a violation of this policy has occurred, SIOR's President, President-Elect, and Vice President, in consultation with SIOR counsel, will determine any disciplinary action. If one or more of the foregoing officers are named in the complaint of harassment or inappropriate behavior, SIOR's counsel will designate a substitute to be selected from the SIOR Governance Committee.

SIOR reserves the right to take any necessary and appropriate action against any member who is found to be in violation of this policy. Disciplinary actions may include, but are not limited to, prohibition from attending future SIOR meetings and events, removal from a committee appointment, expulsion from membership, or any other action deemed appropriate by SIOR.

Any questions about this policy may be directed to SIOR's Chief Executive Officer or Chief Operating Officer.

APPENDIX C: APPLICATION ETHICS COMPLAINT PROCEDURES

Filing an Application Ethics Complaint

Any SIOR member who believes that an applicant should be refused admission to SIOR because of ethical misconduct may file a complaint with the Professional Standards Committee.

What are the grounds for denying admission to SIOR?

The SIOR Bylaws state that "Applicants for Active membership must have a satisfactory real estaterelated business reputation," and specify that an application will be denied under this standard only for serious offenses, such as falsifying information on the application, conviction of a felony, revocation of professional license, or a pattern of professional misconduct that would result in suspension or expulsion of an SIOR member. See the Application Ethics Complaint Form for specific reasons for denying an application.

How do I file an application ethics complaint?

- A completed Application Ethics Complaint Form (see attached), along with a statement outlining specific allegations of misconduct by the Applicant, should be submitted to the Chair of SIOR's Professional Standards Committee.
- The Professional Standards Committee Chair forms an inquiry panel to determine if the alleged conduct would, if proven, warrant suspension or expulsion if engaged in by an Active member.
- If the inquiry panel concludes that a hearing is warranted, the applicant is notified that he/she has the option of either participating a hearing to determine the facts or withdrawing his/her application.
- Hearings are normally held at the time of SIOR's semi-annual conferences (at a private location on-site).

Where do I send a completed complaint form?

The completed form, and accompanying documentation, should be sent to SIOR Headquarters, 1201 New York Avenue NW, Ste. 350, Washington, DC, 20005, Att: Professional Standards Staff Liaison, who will transmit it to the Chair of the SIOR Professional Standards Committee.

What happens to the membership application after a complaint is submitted?

The Applicant is notified of the complaint and given an opportunity to respond. After reviewing the complaint and the information provided by the Applicant, if a Professional Standards inquiry panel determines a hearing is warranted, no further action is taken on the application until either 1) the applicant opts to withdraw his/her application, or 2) the applicant opts for a hearing; in that case, no action is taken on the application until the outcome of the hearing.

If the inquiry panel concludes that a hearing is not warranted, the application is returned to Admissions for continued processing in accordance with usual procedures.

What are my obligations if I file a complaint?

Any member(s) who file(s) an ethics complaint against an Applicant should be prepared to attend and participate in a hearing in the presence of the Applicant if one is scheduled.

Case #	SIOR-PS Form 24

Society of Industrial and Office Realtors®
Professional Standards Process
1201 New York Avenue, NW, Suite 350
Washington, DC 20005 USA
202.449.8200 phone | 202.216.9325 fax

APPLICATION ETHICS COMPLAINT

<u>Please review the attached excerpts from the SIOR Bylaws and Policy Manual before completing this form. Statements as outlined below should be on company letterhead and submitted as attachments to this form.</u>

To the Professional Standards Committee	ee of the Society of Industrial and Office REALTORS [®] :
Date Submitted:	Date Received by SIOR:
Complainant:	Chapter
Member's Firm	
Applicant:Subject(s) of the complaint.	
Applicant's Firm:	Designation Applied for:
Basis of Complaint:	
professional misconduct on the part Complainant believes should be co admission to SIOR. Also, please sp Practice pertain to the alleged misc www.sior.com/Resources/Ethics an	terhead outlining the date(s) and details of specific allegations of of the Applicant, along with any available documentation that the insidered in determining whether the individual should be denied becify which SIOR Ethics Code Principle(s) and/or Standard(s) of conduct. The Code may be accessed on the SIOR Web site: See and Professional Standards. Please call SIOR Headquarters at tith SIOR's Professional Standards staff liaison if you have any
	the SIOR Bylaws and Policy Manual, and understand that if there and and to respond to questions from the Hearing Panel in the
Signature:	
applicant unless and until a Profession	ed will be treated as confidential and will not be shared with the hal Standards inquiry panel has convened and concluded that a acts. If the applicant opts for a hearing, the complainant(s) should the hearing process.

Please submit this form and documentation directly and only to SIOR Headquarters, 1201 New York Avenue NW, Ste. 350, Washington, DC, 20005, Att: Professional Standards Staff Liaison, who will direct it to the Chair of the SIOR Professional Standards Committee.

SIOR Bylaws (Excerpts from Article III, IV, and V)

<u>Article III, Section 2.</u> Applicants for Active membership must have a satisfactory real estate-related business reputation. An application will be denied under this standard only where one or more of the following is found:

- A. The application contains materially false or misleading information or representations;
- B. The applicant has been convicted of or pleaded guilty or nolo contendere to a felony within the preceding ten (10) years;
- C. The applicant has had a professional license suspended or revoked for professional misconduct within the preceding ten (10) years;
- D. The applicant has been suspended or expelled from a Board of REALTORS or other professional association for professional misconduct within the preceding ten (10) years.
- E. The applicant has been involved in arbitration, litigation or other judicial proceedings in which the findings of fact and/or conclusions of law demonstrate conduct by the applicant within the ten (10) years preceding application, that if the applicant was a member of SIOR, could result in suspension or expulsion from SIOR;
- F. The applicant has engaged in conduct during the preceding ten (10) years for which a duly constituted government administrative body, such as a licensing authority, or an independent non-governmental authority, such as another professional association, has issued findings which, if applied to an Active member of SIOR, could result in suspension or expulsion under SIOR's Code of Ethical Principles and Standards of Professional Practice; or
- G. The applicant has engaged in a pattern of professional misconduct within the preceding ten (10) years which, if applied to an Active member of SIOR, could result in suspension or expulsion under SIOR's Code of Ethical Principles and Standards of Professional Practice.

<u>Article IV, Section 4.</u> Any application file containing one or more allegations that the applicant is disqualified from membership under Article III, Section 2. will be transmitted to the Chair of the Professional Standards Committee. The Chair will form a panel of inquiry to review the application file, pointing toward a determination of whether the conduct asserted (or some part of it) would be sufficient, if proven, to warrant suspension or expulsion if engaged in by a member, which determination will be made pursuant to the procedures specified in the SIOR Policy Manual.

<u>Article V, Section 1.</u> Any Active or Associate member of SIOR may be reprimanded, suspended or have membership status terminated for sufficient cause by the Board of Directors.

Article V, Section 2. Sufficient cause is hereby declared to be any of the following:

- A. Any conviction by a court of competent jurisdiction of insanity, incompetence, felony or fraud;
- B. Revocation of a member's real estate brokerage license;
- C. Violation of the Bylaws or policies of SIOR;
- Violations of the Code of Ethical Principles and Standards of Professional Practice of SIOR, or any action or conduct considered by the Board of Directors to be discreditable to SIOR or the profession;
- E. Refusal or willful neglect to comply with any order or decision of the Board of Directors;

- F. No longer meets the criteria for Active or Associate membership;
- G. Nonpayment of national or Chapter dues.

SIOR Policy Manual (Section 2.2.6 B)

B. Complaint Procedures

Any application file containing one or more allegations that the applicant is disqualified from membership under Bylaws Article III, Part I, Section 2 will be transmitted to the Chair of the Professional Standards Committee. The Chair will form a panel of inquiry to review the application file, pointing toward a determination of whether the conduct asserted (or some part of it) would be sufficient, if proven, to warrant suspension or expulsion if engaged in by a member.

This panel of inquiry review will be conducted in a manner consistent with the rules and procedures for ethics complaint review specified in SIOR's *Ethics, Mediation and Arbitration Procedures Manual.* Where a panel of inquiry concludes that the conduct described would not be sufficient to warrant suspension or expulsion, the chair will so indicate in the admissions file. The file will then be presented to the Admissions Committee for its assessment of other aspects of the application, in accordance with the Committee's normal procedures.

Where a panel of inquiry concludes that some of the alleged conduct, or all of the alleged conduct taken together, would warrant suspension or expulsion under SIOR's guidelines if the conduct were that of a member, then further processing of the application will be deferred pending a hearing to determine the facts. The applicant will be notified and given the option of participation in the hearing process. If the applicant declines, the application will be deemed withdrawn.

The hearing to determine the validity of the facts alleged will be conducted before an impartial panel having appropriate professional standards credentials that will be designated by the Chair of the Professional Standards Committee. The procedures governing the hearing will be those described in SIOR's *Ethics, Mediation and Arbitration Procedures Manual*. The applicant and others will be entitled to the rights and subject to the responsibilities described in those procedures.

The hearing panel will make findings as to what conduct had been proven to have occurred and whether that conduct (as distinct from the alleged conduct assessed by the panel of inquiry) would warrant suspension or expulsion of a member under the applicable suspension/expulsion guidelines. The rights of appeal provided by the Manual will be available.

When the hearing process has produced a conclusion, a written memorandum stating that conclusion will be transmitted to the application file, which then will be evaluated by the Admissions Committee following customary procedures, except that the results of the hearing process will be conclusive as to the issues addressed and the matters found.

In the event of a complaint that an application has not been acted upon in a timely fashion, a procedural review board will be constituted to include the Membership Standards Committee Chairman, the appropriate Regional Director and two Members from the Chapter having jurisdiction over the application. The board will promptly investigate the complaint and attempt to resolve informally any discrepancy in compliance with the established admissions procedures. In the event the board identifies a discrepancy that cannot be resolved informally, the Chairman of the Committee will submit a written report to the President and Chairman of the Board and Speaker of the Council of Chapter Presidents, who will initiate such formal remedial action as may prove necessary.